

## Job-housing co-dependent mobility decisions in life trajectories

*Citation for published version (APA):* Guo, J. (2020). *Job-housing co-dependent mobility decisions in life trajectories*. [Phd Thesis 1 (Research TU/e / Graduation TÚ/e), Built Environment]. Technische Universiteit Eindhoven.

Document status and date: Published: 28/10/2020

#### Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

#### Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

 The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

#### General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
   You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

#### Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.

## Job-Housing Co-Dependent Mobility Decisions in Life Trajectories

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Technische Universiteit Eindhoven, op gezag van de rector magnificus prof.dr.ir. F.P.T. Baaijens,

voor een commissie aangewezen door het College voor Promoties, in het openbaar te verdedigen op woensdag 28 oktober 2020 om 13:30 uur

door

Jia Guo

geboren te Shenyang, China

Dit proefschrift is goedgekeurd door de promotoren en de samenstelling van de promotiecommissie is als volgt:

voorzitter:	prof. dr. ir. T.A.M. Salet
1 <sup>e</sup> promotor:	prof. dr. H.J.P. Timmermans
2 <sup>e</sup> promotor::	dr. T. Feng
leden:	prof. dr. ir. D.F. Ettema (Universiteit Utrecht)
	dr. E.J.E. Molin (Technische Universiteit Delft)
	prof. dr. S. Rasouli
adviseur:	dr. D. Yang

Het onderzoek of ontwerp dat in dit proefschrift wordt beschreven is uitgevoerd in overeenstemming met de TU/e Gedragscode Wetenschapsbeoefening.

## Job-Housing Co-Dependent Mobility Decisions in Life Trajectories

A catalogue record is available from the Eindhoven University of Technology Library

ISBN 978-90-386-5119-4 NUR 955

Cover design by Jia Guo

Printed by the Eindhoven University Press, Eindhoven, The Netherlands Published as issue 274 in de Bouwstenen series of the Department of the Built Environment of the Eindhoven University of Technology

Copyright © Jia Guo, 2020

All rights reserved. No part of this document may be photocopied, reproduced, stored, in a retrieval system, or transmitted, in any from or by any means whether, electronic, mechanical, or otherwise without the prior written permission of the author.

## Contents

List of Tables	sxi
List of Figure	sxiii
1 Introductio	n1
1.1	Background and motivation1
1.2	Aim and objectives2
1.3	Organization of the thesis
2 Literature F	Seview5
2.1	Multidimensional choice of residence, job and travel mobility decision5
2.2	Life course approach7
2.3	Household decision process10
2.4	Conclusions
3 Survey Des	ign and Data Collection13
3.1	Introduction13
3.2	Survey design
	3.2.1 Stated choice experiment design and implementation15
	3.2.2 Retrospective survey
3.3	Data collection
3.4	Conclusions
4 Co-Depend	ent Choice of Residence, Job and Travel Mobility Decision23
4.1	Introduction
4.2	Data analysis24
4.3	Methods and results
	4.3.1 Capturing taste variations through mixed logit model26
	4.3.2 Capturing taste variation through a latent class model
	4.3.3 Capturing taste variations through an error component mixed
	logit model
4.4	Conclusions and discussion46
5 Simultaneo	us Mobility Decisions
5.1	Introduction49

5.2	Literature review	50
5.3	Data Analysis	52
5.4	Conceptual considerations and model estimation	53
	5.4.1 Conceptual considerations	53
	5.4.2 Model specification and estimation	54
5.5	Results and discussion	56
	5.5.1 Personal characteristics	56
	5.5.2 Household structure change	57
	5.5.3 Mobility decisions in different life domains	57
	5.5.4 Common error components among multiple long-term m	obility
	decisions	60
5.6	Conclusions and discussion	60
6 Temporal D	ependencies in Mobility Decisions over the Life Course	63
6.1	Introduction	63
6.2	Dynamic Bayesian networks	65
	6.2.1 Conceptual considerations	65
	6.2.2 Model specification	65
	6.2.3 Learning	66
6.3	Data	68
6.4	Results and analysis	68
	6.4.1 Results of structure learning	68
	6.4.2 Results of parameter learning	70
6.5	Conclusions and discussion	74
7 Temporal D	ependencies in Mobility Decisions: A Household-level Analysis	77
7.1	Introduction	77
7.2	Methodology	78
	7.2.1 Background	78
	7.2.2 Network structure	79
	7.2.3 Learning	79
7.3	Data analysis	79
	7.3.1 Data collection	79
	7.3.2 Sample description	80
7.4	Results and analysis	82
	7.4.1 Results of structure learning	82
	7.4.2 Results of parameter learning	83
7.5	Conclusions and discussion	90
8 Conclusions	and Discussions	91
References		95
Author Index		105
Subject Index		107

Summary	
Curriculum Vitae	
Acknowledgements	
Publication List	

## **List of Tables**

Table 3.1 Attributes and attribute levels of residence, job and transportation mode14
Table 3.2 Conditions of travel time of different transportation modes
Table 3.3 Population density of nine districts  21
Table 4.1 Descriptive analysis of the survey attributes  25
Table 4.2 Estimation results of MNL model and mixed logit model
Table 4.3 Value of time of different transportation modes
Table 4.4 AIC and BIC value of the base models with different number of classes33
Table 4.5 Estimation results of the MNL model and the latent class model35
Table 5.1 Descriptive statistics of life events and different mobility decisions
Table 5.2 Estimation results
Table 6.1 Conditional probability distribution of various mobility decisions for different
age categories72
Table 6.2 Transition probability of various mobility change given evidence of life events
Table 6.3 Transition probability of car ownership change given evidence of residential and work change       73
Table 6.4 Transition probability of residential/work location at time t given location at
time t-1 and mobility decisions73
Table 6.5 Transition probability of commuting time at time $t$ given commuting time at
time <i>t</i> -1 and various mobility decisions74
Table 7.1 Sample description 80
Table 7.2 Descriptive statistics of the life course events

Table7.3Conditional probability and probability difference (in parentheses) of residential mobility and car ownership change given different levels of child birth 83
Table 7.4 Conditional probability and probability difference (in parentheses) of job       mobility given different levels of child birth
Table 7.5 Conditional probability and probability difference (in parentheses) of car       ownership change given different levels of residential mobility
Table 7.6 Conditional probability and probability difference (in parentheses) of car       ownership change given different levels of job mobility
Table 7.7 Conditional probability and relative probability differences (in parentheses) of residential mobility at time $t$ given commuting time at time $t$ -1 of the spouses86
Table 7.8 Conditional probability and relative probability differences (in parentheses) of car ownership change at time $t$ given different levels commuting time at time $t$ -1 of the spouses
Table 7.9 Conditional probability and relative probability differences (in parentheses) ofjob mobility at time $t$ given different levels commuting time at time $t$ -1 of thespouses
Table 7.10 Conditional probability and relative probability differences (in parentheses) ofresidential mobility at time $t$ given different levels of annual income at time $t$ -1 ofthe spouses
Table 7.11 Conditional probability and relative probability differences (in parentheses) ofcar ownership change at time $t$ given different levels of annual income at time $t$ -1of the spouses
Table 7.12 Conditional probability and relative probability differences (in parentheses) of       job mobility at time t given different levels of annual income at time t -1 of the       spouses

## **List of Figures**

igure 3.1 The pivot choice design	15
igure 3.2 Historical mobility decisions	19
igure 3.3 Partner's historical job mobility decisions	19
igure 3.4 Survey area in the Shenyang city	22
igure 5.1 Interdependencies between various long-term mobility decisions	54
igure 6.1 Consideration dependencies	64
igure 7.1 Network structure	78
igure 7.2 Learned network structure of concurrent effects (left) and lagged e	effects
(right)	82

## **I** Introduction

#### 1.1 Background and motivation

The transportation literature is well endowed with models of transportation mode choice (e.g. Quarmby, 1967, Meyer, et al., 1978; Grava, 2003; Rodriguez and Joo, 2004; Meixell and Norbis, 2008; Heinen, et al., 2013). The most common way of modeling transportation mode choice is to assume that a set of vehicle and trip attributes generates a certain amount of utility for travelers of a particular socio-economic profile and that travelers choose the transportation mode that maximizes their utility. Sometimes, taste variation is incorporated by assuming the taste parameters exhibit a particular distribution (e.g., Bhat, 2000; Paulssen, et al., 2014).

This common restriction to vehicle and trip attributes has some important limitations. When some time ago, major transit-oriented development projects in the Netherlands decreased travel times by train, prevailing models predicted a shift in transportation mode choice towards the train. In reality, however, a substantial share of the workers decided to move further away from their workplace to enjoy large, relatively cheap housing, located in a more rural area without increasing their habitual commuting times. The transportation mode choice models did and could not predict this response to the development scheme, simply because the transportation mode choice model lacked the relevant variables and the necessary larger perspective. Ignoring the larger residential and job context, these transportation mode choice models may lead to an inflation of the importance of transportation-related variables. Consequently, results may be misleading and suggest ill-founded policy recommendations.

To complement this single-domain oriented research, recently the view has been advocated that individuals and households consider different choice dimensions jointly, implying that various choice dimensions/domains are strongly co-dependent (e.g., Bhat and Guo, 2007; Eluru, et al., 2009; Pinjari, et al., 2011; Paleti, et al., 2012). Rather than maximizing the utility of each choice domain separately and independently, it is more realistic to assume that individuals and households consider these choice domains jointly. Although the co-dependencies between residential choice, job choice, and commuting transportation mode choice have been widely acknowledged, most studies on multi-dimensional choice behavior have focused on two dimensions only (e.g., Paleti, et al., 2012). Studies focusing on more dimensions are relatively rare.

#### 1.2 Aim and objectives

The aim of this PhD study is to better understand the co-dependencies between residence, job and travel mobility decision. We present different models of residence, job and travel mobility decision that consider the interdependency of different domains. In addition, because individuals' preferences are heterogeneous in nature, both observed and unobserved preference heterogeneity are investigated in the context of these co-dependent mobility choices. Moreover, as these domain choices tend to be related to different life stages, we extend the life course modeling approach to better understand these interdependencies in the context of long-term decision processes. The life course approach acknowledges that life events may lead to a reconsideration of ones' current state in different life domains and possibly to decisions changing the current states. For instance, several studies found that child birth plays an important role in various life course mobility decisions due to the fact that an increase in the number of household members may create the need for a bigger house, or an additional or larger car (e.g., Dieleman and Mulder, 2002; Warner and Sharp, 2015). Similarly, commuting mode changes are primarily driven by job change and residential relocation (Clark, et al., 2016).

A limitation of many life course studies is the use of a static model specification. Most studies only capture a snapshot of behavior but ignore that the codependencies between life events themselves may change over time. Until recently, an emerging body of studies on the dynamics in household and mobility decisions suggests that the co-dependencies between decisions on different dimensions may stretch across multiple years. Long-term mobility decisions are not instantaneous decisions but may need time to adapt. Specifically, people may adapt their preference not only based on their current state, but also their historical experiences and future expectations. Therefore, a dynamic life course analysis is introduced in this study to examine the temporal interdependencies between various life domains.

Lastly, research has conventionally viewed individuals as the decision-making unit, ignoring decision roles of other family members. Because family members share various household resources, long-term mobility decisions such as residential mobility and car ownership change are often a decision made jointly by multiple household members. However, both theoretical and empirical analyses of household mobility decisions remain very limited in number. Thus, in order to fill this research gap, this study examines interdependencies between various life events and mobility decisions from both an individual and household level.

#### 1.3 Organization of the thesis

The thesis is organized into eight chapters. Following this introductory chapter, Chapter 2 summarizes existing theories and reviews the relevant literature. First, it reviews the literature on multidimensional choice behavior related to residence, job and travel mobility decision. Second, it reviews the life course approach. Lastly, to investigate intra-household interactions between various mobility decisions, the household decision making process is reviewed.

Chapter 3 explains the survey design. A stated choice experiment and a retrospective survey are implemented in this study. This chapter provides the details of the design and administration of the overall survey design used to collect the data. Features of the questionnaire, properties of the stated choice experiment, the choice of study area, and respondent recruitment are explained and motivated in this chapter.

Chapter 4 reports the results of the analyses based on discrete choice models. To investigate the behavior of people in the context of multidimensional decisions about residence, job and transportation mode, three discrete choice models are discussed in this chapter: a mixed logit model, an error component mixed logit model, and a latent class model.

Chapter 5 discusses how residential mobility change, job change and car ownership change are interrelated, and influenced by the occurrence of key life events such as marriage and child birth. Clearly, attitudes and lifestyle preferences play an important role in shaping the decision-making processes. These lifestyle preferences and attitudes constitute unobserved factors that simultaneously affect choices in different life domains. Considering this, based on the retrospective data, a simultaneous equations model is formulated to estimate the effects of both observed and unobserved factors influencing various mobility decisions in different life domains.

To understand the interdependencies among various life domains and the timing of these decisions, as two main methods, hazard models and Bayesians network approach have been applied. However, a limitation of the existing approaches is the time-invariant nature of the causal structure between life trajectory events. In light of this limitation, a dynamic Bayesian network is introduced in Chapter 6. While Chapter 6 is based on individual trajectories, Chapter 7 examines temporal interdependencies between various life course mobility decisions at the household level.

The last chapter summarizes the main conclusions of this study, discusses their implications for transportation and urban policy, and provides recommendations for future research.

# 2

### **Literature Review**

In this chapter, we will review the literature on the choices of various life domains in residence, job and travel mobility decision. First, we review the literature on multidimensional choice behavior related to residence, job and travel mobility decision. Then, we briefly review the life course approach because this approach constitutes an integral framework, taking temporal effects into account. In this context, individuals/households make their decisions not only based on their current state but also consider their previous experiences and future expectations. Lastly, because many long terms decisions in dual-earner households tend to be household decisions involving more than one family member, the literature on household decision making process is briefly summarized.

## 2.1 Multidimensional choice of residence, job and travel mobility decision

The transportation literature is extremely rich of transportation mode choice models. The vast majority of these models assume that individuals maximize the utility they derive from the attributes of the transportation mode, considering trip characteristics. Thus, these models involve a uni-dimensional choice problem. Consequently, any predictions based on these models are necessarily refined to the uni-dimensional domain captured by the model. This is not a problem if the management or policy issue concerns transportation mode choice, and does not affect space-time prisms of individuals. However, if the policy would open up new opportunities, individuals may reconsider the intricate relationship between residence, job and transportation mode.

During the last decades, many studies have emphasized the interdependencies between residential mobility and commuting mode (e.g., Desalvo and Hug, 2005; Bhat and Guo, 2007; Salon, 2009; Pinjari, et al., 2011; Guerra, 2015). As an example, Handy, et al. (2005) found significant changes in travel mode and car travel distances after residential relocation. In the job domain, key events such as first-time entry into the labor market, job change, income change and retirement were found to influence travel mobility change (e.g., Dargay, 2001; Dargay and Hanly, 2007; Scheiner and Holz-Rau, 2013b). In turn, commuting mode preferences also influence residential and/or job location choice. For example, commuters who use public transportation may find the egress time important and consciously choose to live/work near a transit station (Bhat and Guo, 2007). By contrast, people living in sprawling areas rely more on cars to conduct their daily activities (Khattak and Rodriguez, 2005; Schwanen and Mokhtarian, 2005a; 2005b). Additionally, increasing travel time may trigger people to reconsider and possibly change their residence or job. Ettema (2010) pointed out that an excessive commute distance may prompt the relocation process and trigger people to move to a new dwelling; Similarity, Rashidi, et al. (2011) found that longer commuting time or activity duration may accelerate relocation decisions. This emphasis on interdependency has led to the joint choice modeling of residential/job location and commuting behavior (e.g., Kim, et al., 2003; Ng, 2008).

Although the interdependencies between residential and job choice and travel mobility decision has been acknowledged in the literature, most studies considered only two dimensions. Studies focusing on all three dimensions are relatively scare. Several decades ago, Lerman (1976) developed a multinomial logit model that combined multiple dimensions (residential location, automobile ownership, and commute mode choice). Rich and Nielsen (2001) presented a micro-econometric model for forecasting long-term travel demand considering residential location, house type and choice of job location. Salon (2006) explored the relationship between the transportation and land use system in New York City by modeling a multinomial logit model of the joint choices of residential location, car ownership, and commute mode of New Yorkers. Likewise, using data from the San Francisco Bay Area, Pinjari et al. (2011) formulated a joint model of residential location, car ownership, bicycle ownership, and commute tour mode choice. They found that these aspects are interrelated and one choice dimension is not exogenous to the others, but endogenous to the system as a whole. In another

publication, Paleti et al. (2012) developed an integrated econometric model system that simultaneously considers six different choice dimensions: residential location, job location, commute distance, household vehicle ownership, commuting mode and number of stops of commute trips. Similarly, using multinomial discrete choice model, Vietnam, Tran, et al. (2016) jointly modelled the choices of residence, workplace, and commuting modes in Hanoi and confirmed significant interdependencies between these choice dimensions.

Although these multidimensional choice models consider long-term decisions and short-term decisions as a single integrated choice, unfortunately, only a limited number of variables were chosen in the multidimensional choice model, which may oversimplify the choice problem and make it difficult to depict the decision-making process in the real world.

#### 2.2 Life course approach

The life course approach, which originates in sociology and psychology field has introduced into the mobility research since 1980s, whereas the life events have been recognized a central role for a variety of economic and demographic triggers of mobility decisions (Elder, 1994, 1998). According to the life course approach, life events cover several domains (household, employment, education and residence). These events run in parallel and are associated as events in a particular domain. This may lead people to reconsider their status in that domain and other life domains.

Over the past decades, research examining various mobility decisions has been enriched by applying the life course approach. Empirically, research based on the life course approach has unfolded along two lines. First, a substantial body of research has demonstrated that changes in household composition trigger residential/job relocation and/or mobility tools possession choice. For example, Rossi (1980), as one of the earliest studies, argued that changes in family size may render the current dwelling inadequate, thus creating dissatisfaction with the current dwelling. Habib et al. (2011) examined the effects of household structure change on job mobility decisions, and found that an increase in household size may increase household member propensity to change job. Similarly, using the German Socioeconomic Panel data, Prillwitz et al. (2006) confirmed the key role of marriage and child birth on car ownership decisions. Similar findings were reported elsewhere in the literature (e.g., Verhoeven, et al., 2005; Lanzendorf, 2010; Beige and Axhausen, 2012).

Second, another stream of research has analyzed the impact of residential move and/or job relocation on mobility tools possession choice. For instance, Prillwitz et

al. (2006) found that there is significant car ownership growth because of the residential change, suggesting that a change in residential location could be a main cause of transportation mode choice. Additional evidence about change in car ownership conditioned on residential relocation has been reported in other studies (e.g., Prillwitz et al., 2006; Choocharukul et al., 2007; Kim, 2008; Scheiner and Holz-Rau, 2013a). In addition, several studies have investigated the impact of a change in job location on car ownership decisions. As an example, Lanzendorf (2010) concluded that a new job may stimulate individuals/households to purchase a car to save commuting time without changing their place of residence. Similar findings were reported in Prillwitz et al. (2006), Habib et al. (2011), Rashidi et al. (2011), and Yang et al. (2017) to name a few.

In fact, most long-term mobility decisions are not instantaneous decisions. Earlier decisions may cause individuals to reconsider their current behavior, but in some cases with a time lag (Yamamoto, 2008; Fatmi and Habib, 2016). In other words, people may need time to adapt, which implies a lagged response to the changing needs. Similarly, long-term decisions may also depend on individuals' anticipation of certain life events such as expected marriage and child birth (Oakil, et al., 2014; Yu, et al., 2017).

Empirically, to incorporate the temporal dependencies between various life domains, two different types of modelling approaches have been applied. The first one is hazard-based modeling. These models predict the interval times of life course events and their transitions as a function of the time elapsed since the last occurrence of the event and a set of covariates. For example, using a hazard model, Beige and Axhausen (2008) utilized a hazard model to compare different durations of residence, education, employment, and ownership of mobility tools. A major limitation is that these models cannot fully capture the complex direct and indirect relationship between the life course events and mobility decisions.

The second is known as the Bayesian network. A Bayesian network (BN) offers some advantages over econometric approaches in analyzing complex interdependencies among a set of variables. First, a BN has the ability to deal with uncertain and complex relationships hidden in the data. Especially in the context of life course mobility decisions in which state changes in one domain can lead to the changes in other domains, the BN approach has a clear advantage to find the complex interdependencies. Second, a BN can incorporate different types of information, including empirical data, theoretical relationships, and expert knowledge. It is an ideal representation for combining prior research knowledge and data (Heckerman, 1995). When testing a proposed theory using a BN, one does not have to rely only on statistical evidence that may be biased, particularly for small samples.

Bayesian networks, however, have some disadvantages. The most important of these is the treatment of time, which is an important dimension in detecting the temporal dependencies among different life domains. The problem is that Bayesian networks were not designed to explicitly model temporal relationships. To incorporate time element into the model, two types of Temporal Event Bayesian network models contribute to modeling the temporal effects, Temporal Nodes Bayesian Network (TNBN) and Network of Probabilistic Events in Discrete Time (NPEDT). TNBN is composed of a set of temporal nodes, which represents an event or a state change. For each node, time is discretized in a finite number of intervals. Verhoeven, et al. (2007) applied a TNBN to model and simulate the effects of life trajectory events on transportation mode choice decisions. Similar to TNBN approach where each variable represents an event that can occur only once, NPEDT differs from TNBNs in the temporal intervals. While TNBNs assumes the interval of each temporal node is relative to its parent nodes, NPEDT assumes that time is absolute and each value of a variable represents the instant at which a certain event may occur, without a dependency on its parent nodes. Based on this approach, Oakil, et al. (2011) found evidence of 1-year lagged effects of change in employment status on the probability to move house. Similarly, birth of the first child was found to have a 2-year lagged effect on residential relocation. In another work, Oakil, et al. (2014) built a framework investigating residential relocation, employer change and change in car ownership level and temporal dependencies among these long-term decisions and other household decisions. Similarly, using the NPEDT, Wang, et al. (2018) confirmed that interdependencies exist among the long-term and mid-term life domains, and evidenced the reactive and proactive behavior of individuals and households in the context of various life events over the life course.

As for the treatment of time, these studies either treat the probability of a state of certain nodes at the previous year as an independent node or transform two events at different times as one event (e.g. defined as increase or decrease), regardless of the contextual dependencies with other nodes which should be incorporated at the same (previous) time slice. In addition, treating the dynamic dependency in this static way may involve extra complexity in network structure learning. To improve the modeling process, a dynamic Bayesian network approach may be applied.

#### 2.3 Household decision process

Although the life course approach provides a better understanding of the interdependencies in long-term decision processes, individuals are conventionally assumed to be the decision-making unit. However, because household members physically share household resources, long-term decisions affect all household members. Moreover, especially if both partners have a job, the decision-making process is much more complicated under such circumstances because both partners have to decide jointly about their choice of residence and their choice of jobs. Long-term decisions, hence, tend to be made at the household level (Timmermans, et al., 1992; Borgers and Timmermans, 1993; Roorda, et al., 2009; Timmermans and Zhang, 2009).

Taking household as the decision unit, Scheiner (2014) studied changes in travel mode specific trip rates after life course events. Significant effects were found for some key events, and some effects differed distinctly between men and women, suggesting that men and women are differently affected by life course events. Similarly, Oakil (2016) provided empirical evidences that life events such as birth of the first child, residential relocation and job change only significantly affect women's decisions to get full-access to a car.

Although some studies focused on specific aspects, intra-household analyses in long-term mobility decisions remain very limited in the existing literature; this study contributes to provide a theoretical and empirical framework of intra-household interactions on various household mobility decisions.

#### 2.4 Conclusions

Residential and job choice are critical in understanding individuals' activity-travel behavior. The modeling and prediction of transportation mode choice depends on other choice dimensions. However, this review of the literature suggests that few studies have reported individuals' preferences for the interdependent choice of house, job and transportation mode. Ignoring the effects of residential and job choice on transportation mode decisions may lead to biased estimation results and therefore misleading policy recommendations or assessment of policy impacts. In order to address the relative paucity of this kind of research, multidimensional choice model needs to be further developed.

Life course approach provides a rich framework to study these decision processes. Existing studies that investigated the temporal effects between life events and long-term mobility decisions tend to oversimplify the decision process over a life trajectory. The interrelations between various mobility decisions need to be examined further from a dynamic perspective. Moreover, relative to the individuals' perspective, the literature of household decision makings on these mobility decisions is relatively scarce. Household level studies would be an important addition to the state-of-the art.

# 3

### Survey Design and Data Collection

#### 3.1 Introduction

The aim of this PhD thesis is to analyze the co-dependency of long-term residential and job decisions and short-term travel behavior using a variety of different models. In order to analyze these choices in different life domains, a stated choice experiment was developed. In addition, to allow a life course analysis, a retrospective survey that recorded demographic transitions, housing and job careers was administered.

The next sections will outline how these different parts of the survey were developed in order to collect data about the various concepts introduced in the last chapter. Next, the field work is discussed. The chapter ends with a summary and conclusions.

#### 3.2 Survey design

The questionnaire was developed and conducted through a platform (Pauline questionnaire system) that has been developed by our group to generate Internetbased questionnaires. The questionnaire requirements for this study consist of two segments: a stated choice experiment and a retrospective survey, which are further explained below.

Name	Attribute	Attribute level
	Tenure	Rent, Buy
	Cost (yuan/ month)	500, 1600, 2200, 3800
	Size (m²)	30, 65, 90, 125
Residence	Distance to the metro station	-75%, -25%, 25%, 75% variation from the current
	Distance to the bus stop	-75%, -25%, 25%, 75% variation from the current
	Distance to shopping mall	-75%, -25%, 25%, 75% variation from the current
	Home location	Central of the city, Surrounding area
	Working type	Government, Institution, Enterprise, Self-employed
	Flexibility	Yes, No
	Salary (yuan/year)	30k, 80k, 130k, 200k
Job	Work environment	Very good, Good, Poor, Very poor
	Colleague relationship	Very good, Good, Poor, Very poor
	Easy to find a similar or not	Yes, No
	Job location	Central of the city, Surrounding area
Transportat	ion mode	
	Travel cost (yuan)	1.4, 7, 12.6, 21
Car	Travel time (min)	4.5, 18, 31.5, 45
	Congestion time (min)	-75%, -25%, 25%, 75% variation from the current
	Travel cost (yuan)	2, 4
Matua	Travel time (min)	5, 20, 35, 50
Metro	Out-of-vehicle time (min)	10, 20, 30, 40
	Have seats or not	Yes, No
	Travel cost (yuan)	1, 2
	Travel time (min)	8, 29, 50, 71
Bus	Out-of-vehicle time (min)	5, 10, 15, 20
	Congestion time (min)	-75%, -25%, 25%, 75% variation from the current
	Have seats or not	Yes, No
Bike	Travel time (min)	9, 54, 84, 120
Walk	Travel time (min)	30, 144, 252, 360

#### Table 3.1 Attributes and attribute levels of residence, job and transportation mode

	Current Situation	Move House				(	Change Jol	b			
Tenure	Buy	Rent									
Cost (yuan/month)	3800			500							
Size (m <sup>2</sup> )	80			65							
Distance to the metro station (km)	3			3.75			Your current house				
Distance to the bus station (km)	1			0.25							
Distance to the shopping mall (km)	2			2.5							
Home location	Central of the city		Cer	tral of the	city						
Work type	Government							Se	If-employ	ed	
Flexibility	Yes						No				
Salary (yuan/year)	8w						18w				
Work environment	Not good	Your current job					Good				
Colleague relationship	Good	,					Poor				
Easy to find a similar job or not	Yes						No				
Job location	Central City						Surrounding area				
Transportation mode	Bus	Car	Metro	Bus	Bike	Walk	Car	Metro	Bus	Bike	Walk
Travel Cost (yuan/time)	1	1.4	2	1			7	2	1		
Travel time (min)	40	4.5 5 8 9 30				18	20	29	54	144	
Out-of-vehicle time(min)	5	10 5				20	5				
Congestion time (min)	10	7.5 7.5			17.5		7.5				
Have seats or not	No		Yes	Yes				Yes	No		
Your choice											

Figure 3.1 The pivot choice design

#### 3.2.1 Stated choice experiment design and implementation

A stated choice experiment was designed to measure the preferences for multidimensional residential choice, job choice and travel mobility decision. A stated choice experiment was chosen because it gives researchers more control over the variation of attribute levels, making the approach more appropriate for theoretical studies.

Attributes that potentially influence the choice of resident, job and transportation mode were selected based on the results of literature review. The examination of the relevant literature suggests that residential choice is primarily affected by tenure, monthly costs, housing size, distance to metro station, bus stop, shopping center and the location of the dwelling (e.g., Louviere and Timmermans, 1990; Timmermans, et al., 1992; Bagley, et al., 2002; Walker and Li, 2007; Balbontin, et al., 2015). Similarly, based on the existing literature, job situations were represented in terms of type of work, flexibility of the work schedule, income, work environment, relationship with co-workers, easy to find a similar job and job location (Timmermans, et al., 1992; Tran, et al., 2016). Commuting mode behavior is based on travel costs, travel time, out-of-vehicle time, congestion time and having seats or not (Van Ommeren, et al., 1999; Bagley, et al., 2002). Hence, the experiment involves a total of (7+7+5) = 19 attributes.

Some researchers have expressed concern about the administration of such complex experiments with so many attributes and have advocated the use of simple experiments with a very small number of attributes. In contrast, our contention is that, as a guiding principle, experiments should capture the complexity of real-world decision-making. A technical problem is that when modelling multi-dimensional choice phenomena, choice sets may explode in size due to the large number of selected attributes. To limit the problem, we separately generated home-job profiles and mobility profiles and randomly paired these profiles. For the home-job design, five attributes were systematically varied in terms of two levels, while another ten attributes had four levels. In the mobility experimental design, five transportation modes were included and shown to respondents. These modes include car, metro, bus, bike and walk. Table 3.1 lists the attributes and their levels used in the experimental design.

The choice profiles were generated using a Bayesian D-efficient optimal design by Ngene. The objective of producing efficient designs is to minimize the asymptotic standard errors obtained from models estimated from data collected from sampled individuals (Rose, et al., 2008). Prior expectations of the size and sign of the effects of selected attributes are provided based on the findings of previous research to increase the efficiency of the designs. If no previous research existed, parameters are assumed.

Moreover, instead of all respondents facing the same choice situations, we developed a pivoted-efficient design, which presents changes relative to the respondents' current situation (Hensher, et al., 2015). We created this optimal stated choice experiment using five selected variables (distance to the bus station, metro station and shopping center from home; congestion time of car and bus) as the pivoting attributes. We pivoted these five attributes (-75%, -25%, 25%, 75% difference from the current) around personally experienced values, thus representing more realistic hypothetical profiles specific for each respondent.

The inclusion of attributes such as travel times in stated choice experiments tends to be challenging. The design of experiments typically independently varies the attributes levels of the choice alternatives. Consequently, this procedure may result in highly unrealistic travel time differences between modes. For example, the bus may be substantially faster than the private car. It may lead respondents to doubt the professionalism of the research and credibility of the survey. In part, such unrealistic differences may be explained away by pointing at dedicated bus lanes and similar traffic control measures. Nevertheless, such explanations introduce other latent attributes, the effect of which is unknown.

Condition	Car	Metro	Bus	Bike	Walk
Short-distance	4.5	5	8	9	30
Medium 1-distance	18	20	29	54	144
Medium 2-distance	31.5	35	50	84	252
Long-distance	45	50	71	120	360

Table 3.2 Conditions of travel time of different transportation modes

To avoid this problem, a different procedure was applied in the present study. Rather than independently varying travel times across transportation models, a vector of travel times for the different modes was systematically varied. This vector reflects underlying speed differences between transportation modes. Four levels and hence four vectors were used. We first assumed four distance categories with constant intervals: 5, 20, 35, 50 km. Then, these distances were converted into travel times by dividing them by the typical speeds of the different transportation modes for the study area. Small variations around the speeds were assumed: (1) the travel times of the bus was assumed less than proportional to the travel times of the car because the number of stops tends to be less for the longer trips by bus; (2) the travel times by bike and walking were assumed disproportionally higher due to increasing fatigue. Note that the different speeds destroy the equation intervals of distance. This is not considered an issue because we created a D-optimal rather than an orthogonal fractional factorial design anyhow.

Table 3.2 displays the four vectors (conditions) that were used in the experimental design. Note that for some conditions, the travel times by bike and especially walking become very high and therefore probably prohibitive to choose the corresponding transportation mode. We argue this is a non-issue as the estimated models will predict approximately zero choice probabilities for the concerned transportation modes under these circumstances. If the selected algorithm would not converge, the usual technical tricks can be applied to solve non-convergence.

To be more specific, we developed two designs, one for the long-term residential and job dimension, and the other for the short-term transportation mode dimension. Each design has 128 profiles. By pairing these profiles between two designs, we generated the combination of paired profiles in the sense that each paired profile represents a choice set. Each respondent received eight choice sets that were randomly selected from the full list of paired profiles, which means each respondent needed to

finish the choice experiments eight times. In total, 3,208 observations were available from 401 effective interviews.

We created the full set of multidimensional attribute profiles, covering the residential, job and commute mode choice. As shown in Figure 3.1, the choice task includes eleven labelled alternatives: three long-term decisions (status quo, move house or change job), and for the latter two options, the choice among five transportation modes. Respondents were told that at any moment in time they may become aware of a new job opportunity and/or a vacant house. They were asked whether they would change job, move house or do not take any action if faced with these options. If they moving house or changing job, they should also indicate which transportation mode they would choose for the commute trip.

Although the survey was administered on a face-to-face basis, we used a stand-alone version of the Web-based survey system. The system offers many features not available in commercial software, particularly with respect to stated preference and choice experiments. Because different interviewers conducted the interviews, relatively strict controls were maintained in the process. For each variable, a range of feasible responses was defined. If a respondent violated the specified range, the respondent was asked whether the answer was correct and given the opportunity to change the response. In addition, except for some variables such as income that are known to lead to missing values, respondents could only continue with the questionnaire if they provided a response to every question. Finally, logical relationships between different variables were specified and answers were checked for consistency and feasibility. Again, if an answer did not satisfy the conditions we imposed, the inconsistency was signaled to the respondent, who was then asked to provide a new answer.

The relatively strict controls also served as a data cleaning tool. The set of collected data automatically satisfies all data range constraints and conditions on the allowed interdependencies of the data. Although we did not quantify the number of activations of the controls, debriefing of the volunteers gave the impression it was not activated many times, except in those cases where respondents simply forgot answering a question or made mistakes in using the platform. Despite the automated controls, a set of analysis was completed, including identifying and deleting respondents with invariant response patterns. Ultimately, 401 out of 450 were used in the analyses.

#### Survey Design and Data Collection

	Married	Divorce	Birth of Child
First time	Make a choice •	Make a choice •	Make a choice •
Second time	Make a choice •	Make a choice •	Make a choice •
Third time	Make a choice •	Make a choice	Make a choice •

#### Please fill in your current and historical residential information.

	Current	Previous h1	Previous h2	Previous h3	Previous h4	Previous h5
Residential location	Make a choi	Make a choi	Make a choi	Make a choi	Make a choi 🔻	Make a choi 🔻
When do you move to this house	Make a choi 🔻					

#### Please fill in your current and historical work information.

	Current	Previous w1	Previous w2	Previous w3	Previous w4	Previous w5
Work Location	Make a ci 🔹	Make a cl 🛛 🔻	Make a ct 🔹	Make a cł	Make a ct 🔹	Make a ct 🔹
Job starting year	Make a ci 🔻	Make a cl 🔻	Make a ct 🔻	Make a ct 🔻	Make a ct 🔻	Make a ct 🔻
Type of work contract	Make a ci 🔹	Make a cl 🔻	Make a ct 🔻	Make a ct 🔹	Make a ct 🔹	Make a ct 🔻
Type of company	Make a ci 🔹	Make a cl 🔻	Make a ct 🔹	Make a cł 🔻	Make a ct 🔻	Make a cł 🔻
Work status	Make a ci 🔹	Make a cl 🔻	Make a ct 🔻	Make a ct 🔹	Make a ct 🔹	Make a ct 🔹
Salary	Make a ci 🔹	Make a cl 🔻	Make a ct 🔹	Make a ct 🔻	Make a ct 🔻	Make a ct 🔻

#### Figure 3.2 Historical mobility decisions

Please indicate the travel environment of your partner's historical work.

	Current	Previous w1	Previous w2	Previous w3	Previous w4	Previous w5
Mode	Make a ci 🔹	Make a cl 🔻	Make a ct 🔻			
Travel time	Make a ci 🔻	Make a ci 🔻	Make a ct 🔻			

Please fill in the current and historical work information of your partner.

	Current	Previous w1	Previous w2	Previous w3	Previous w4	Previous w5
Work Location	Make a ci 🔹	Make a ci 🔹	Make a cł	Make a ct 🔹	Make a cł	Make a ct 🔹
Work starting year	Make a ci 🔹	Make a ci 🔹	Make a ct 🔹	Make a ct 🔹	Make a ct 🔻	Make a ct 🔻
Work status	Make a ci 🔹	Make a ci 🔹	Make a ct 🔻			
Type of work contract	Make a ci 🔻	Make a ci 🔻	Make a ct 🔻			
Type of company	Make a ci 🔹	Make a ci 🔹	Make a ct 🔻			
Salary	Make a cl 🔹	Make a cl 🔻	Make a ct 🔻			

Figure 3.3 Partner's historical job mobility decisions

#### 3.2.2 Retrospective survey

For testing the temporal causal relationships, longitudinal data are needed. Although panel data is the best option for the purpose of the study, obtaining enough information about long-term behavioral changes in a panel survey is both time and resource consuming. Thus, as an alternative to the panel survey, a retrospective approach was used, which asked respondents to recall their historical mobility decisions. Previous studies (e.g., Verhoeven, et al., 2005; Beige and Axhausen, 2008; Oakil, et al., 2014) based on retrospective surveys indicate that retrospective surveys can provide reliable information about the temporal events as long as respondents tend not to easy forget the measured events. We contend that demographic events, housing and job history have such a pregnant impact on people's life that they can relatively recall these events and provide reliable and valid responses.

Respondents were asked to provide a wide range of longitudinal information of their life course events. The retrospective data was collected from five different sections: 1) information of household members. Questions asked concern age, gender, highest education, etc.; 2) household structure biography (i.e., get married and child birth); 3) historical residential mobility (i.e., the year move, historical residential location, etc.); 4) historical employment mobility (i.e., the year of changing job, historical annual salary and job locations); and 5) historical car ownership change and corresponding historical commuting time. Figure 3.2 shows the organization and structure of the retrospective survey.

A web-based retrospective survey was designed and implemented with the assistance of interviewers. Respondents were asked to continue only if they provided the detailed information of all mobility decisions in the past. A potential problem of retrospective data is that richer life experiences may take substantial time. Thus, in order to reduce respondent burden, respondents were asked to provide information about the life events for only the last five times it occurred. In many cases, this maximum of five still covers the full trajectory of life events in a particular domain, particularly for younger respondents.

#### 3.3 Data collection

The survey was conducted in Shenyang, China, between September and November, 2016. Shenyang has a total area of 3495 km2 and 8.3 million people. As the capital city of Liaoning province, Shenyang is an important industrial city and a hub for transportation and commercial activities. The city is located in the northeast of China,

covering five main districts in the central city (Heping, Shenhe, Huanggu, Dadong, Tiexi) and four districts in the surrounding areas (Hunnan, Yuhong, Shenbei, Sujiatun). The surrounded four districts are relatively newly developed areas. The location of nine districts is shown in Figure 3.4. Population densities of the nine districts are shown in Table 3.3. The interviews were conducted in the nine districts based on a spatially stratified sample. Considering the aim of this study, we only interviewed respondents who have a job.

Volunteers assisted respondents to complete the questionnaire. These were mainly master students and trained to provide good instructions to respondents. First, after a general introduction about the aims and objectives of the study, and the methodology used, they were asked to complete the questionnaire themselves and identify questions they did not understand. This step also served as another pilot of the survey. The original questionnaire was designed in English by the authors. Several rounds of improvements were made until a final survey instrument that did not create obvious problems was developed. This version was then translated into Mandarin, and the student volunteers served as the trial respondents to check the wording, flow and explanations of the questionnaire. Their feedback was discussed within the group of volunteer interviewees, and final changes were made. Individual interviewees were monitored during the data collection process and further instructions and feedback were given on a daily basis, if necessary.

Respondents were given small gifts of appreciation as the survey was 24 web pages long and its completion took over 50 minutes. The response rate is about 16%, which is satisfactory, considering the length of the survey.

District	Heping	Shenhe	Huanggu	Dadong	Tiexi
Population	651,557	711,914	818,015	681,607	908,652
Area (km2)	59	60	66	100	286
Population density (person/ km2)	10,952	11,961	12,360	6,807	3,177
District	Hunnan	Yuhong	Shenbei	Sujiatun	
Population	333,563	445,834	320,337	427,158	
Area (km2)	734	499	884	782	
Population density (person/ km2)	455	893	362	546	

#### Table 3.3 Population density of nine districts


Figure 3.4 Survey area in the Shenyang city

# 3.4 Conclusions

In this chapter, the design and administration of the survey was presented. The survey aims at collecting data on people's preferences when they jointly consider various key variables influencing residential, job and transportation mode choice for commute trips and on key life events and mobility decisions. The data collection was implemented in Shenyang, China, from September to November 2016. Respondents were selected at random from five main districts in the central city and four other districts in the surrounding area. It was a successful experiment and the response rate is satisfactory, considering the length of the survey.

# 4

# **Co-Dependent Choice of Residence, Job and Travel Mobility Decision**<sup>1</sup>

# 4.1 Introduction

Individuals' travel mobility decision highly depends on the choice of where to live and where to work. An examination of the literature reveals that existing studies mainly focus on two choice dimensions only. In an attempt to modeling people's co-dependent choices concerning residence location, job and transportation mode, we assume that individuals/households attempt to find the combination of these three dimensions that maximizes their utility.

However, individuals' preferences are heterogeneous in nature, and the needs and preferences differ between respondents. Hence, in this chapter, preference heterogeneity in the described multi-dimensional choice behavior will be investigated.

<sup>&</sup>lt;sup>1</sup> This chapter is based on the articles:

*Guo J., Feng T., Timmermans H.T.P. (2020). Modeling co-dependent choice of workplace, residence and commuting mode using an error component mixed logit model, Transportation: Vol. 47, 911-933.* 

Guo J., Feng T., Timmermans H.T.P. (2020). Co-dependent workplace, residence and commuting mode choice: results of a multi-dimensional mixed logit model with panel effects, Cities: Vol. 96, 102448.

To specify heterogeneity, first, we consider a range of demographic variables, to examine differences in sensitivity due to observed household/individual factors. We estimate covariate-dependent effects to account for heterogeneity around the mean of taste parameter distributions. Second, a latent class model is formulated to account for individuals' preference heterogeneity. Lastly, to allow for the possibility that unobserved preferences for transportation modes depend on long-term choice behavior, long-term choice specific error components are identified and the variance of these error components is estimated through parameterization of their heteroscedasticity (Hensher and Greene, 2003). Thus, we estimate an error component mixed logit model to identify random and systematic long-term choice specific heterogeneity. As an extension of the standard mixed logit model, this error component approach includes parameter estimates for latent error component effects (Greene and Hensher, 2007). Specifically, the error component consists of IID and non-IID components, and the non-IID component part associates the unobserved variance nests with socio-demographic characteristics (gender, income and numbers of workers in the households).

The remainder of this chapter is structured as follows. In the following sections, the data analysis regarding a subset of samples will be presented first. Results of the model estimation are presented in the next section. Various sources of observed and unobserved heterogeneity in the multidimensional choice behavior are reported afterwards. Finally, the chapter is concluded by summarizing the major conclusions and illustrating the future research directions.

#### 4.2 Data analysis

The data used for this chapter is obtained from the multidimensional stated choice experiment. In our questionnaire, respondents were asked to report their current residential situation, job, and travel behavior, as well as personal and household sociodemographic characteristics. In total, 401 out of 450 valid data were used. The descriptive statistics of the main socio-demographic variables and other characteristics of respondents are reported in Table 4.1.

It shows that 63.8% of the respondents reside in the central city and 36.2% in the surrounding areas. Similarly, regarding job location, 65.3% are employed in the central city, 34.7% work in the surrounding areas and other cities. 48.6% of the respondents is male, 51.4% is female. People participating in the survey are younger than average. Owing to the purpose of our research, only respondents having a job were taken considered, which explains this finding.

Variable	Classification	# of Cases	Percentage (%)
Gender	Male	195	48.6
	Female	206	51.4
Age	18-40	281	70.0
	>40	120	30.0
Marital status	Couple with children	145	36.2
	Couple without children	38	9.5
	Single	218	54.4
Number of workers	dual-earn workers	240	59.9
	Single worker	161	40.1
Annual income	<60,000	243	60.6
	>=60,000	158	39.4
Tenure	Buy	337	84.0
	Rent	64	16.0
Living area	Central city	256	63.8
	Surrounding area	145	36.2
Type of work unit	Government	35	8.7
	Public institutions	74	18.5
	Joint venture, private firms	201	50.1
	Self-employed	91	22.7
Working area	Central city	262	65.3
	Surrounding area	139	34.7
Commute mode	Car	106	26.4
	Metro	65	16.2
	Bus	96	23.9
	Shuttle	32	8.0
	Bike	75	18.7
	Walk	27	6.7

 Table 4.1 Descriptive analysis of the survey attributes

The distribution of marital status shows that singles represent 36.2% of the sample, while couples without children make up only 9.5%. It suggests that 50% of the respondents has one or more children. The mean annual income is 60,000 RMB with a standard deviation of 39,000 RMB. Additionally, of the 401 surveyed respondents, 240 belong to a dual worker household, while 161 belong to a single-worker household. As for commuting, about 58% of the respondents mentioned they do not have a car in

their household. Public transportation is the dominant travel mode, 65 and 96 cases for metro and bus respectively. 26.4% of the respondents driving to work for their current job, while the percentage of slow traffic (bike and walk) still remains at a high rate in our sample (25.4%).

#### 4.3 Methods and results

Before estimating the model, all attributes were effect-coded. We first estimated a basic Multinomial Logit model and then a mixed logit model and a latent class model to uncover the unobserved heterogeneity of the taste variation. Finally, an error component mixed logit model was developed to estimate the unobserved heterogeneity through the selected taste parameters and the choice dependent heteroscedasticity in error component variance.

#### 4.3.1 Capturing taste variations through mixed logit model

We first estimated a basic Multinomial Logit model and then a Mixed Logit model, assuming a normal distribution for five selected variables. Three residential attributes and two job attributes were selected to define the unobserved heterogeneity of the selected random parameters. Alternative-specific constants were set in our model, unobserved heterogeneity of these constants was also taken into account. Furthermore, to uncover the unobserved heterogeneity, socio-demographic attributes were introduced. Potentially, there are many interaction effects, but only those interaction effects that were significant were included in the final specification of the model. The reported estimates are based on these 500 draws.

In our experiment, each participate responds to the choice 8 times. If we ignore the repeated measurement nature of the data, biased parameter estimates may be produced. Furthermore, the model will underestimate the standard errors of the parameters, thus t-values will be inflated. In turn, this may lead researchers to falsely decide some effects are significant for the current sample size. Thus, we estimated these panel effects.

Estimation results are presented in Table 4.2. The overall fit of the multinomial logit model is good (McFadden's rho squared is 0.560, adjusted rho squared is 0.539). The adjusted rho squared for the basic multinomial logit model is 0.488. Moreover, most main and interaction effects are significant.

#### Interpretation of the residential choice component

The coefficients for the residential attributes lead to some interesting interpretations. Effects of tenure and housing size are significant, which implies that people prefer to buy their own house with large space, suggesting that buying house with large space gives people more security and stability. Moreover, it is found that standard deviations of these two variables are significant, indicating that significant unobserved factors influence individuals/households' sensitivity of residential decision making. Furthermore, the results show that old people are generally reluctant to rent house and more likely to live in a very large house but more reluctant to live in a very small house. Housing ownership is generally the largest part of the household's wealth. It shows that older people prefer owning a property, considering it an asset for the future. Similarly, the results show that households with one or more children are more likely to consider buying a house rather than renting one, relative to households without children. These results are in line with the life course analyses, which indicates that buying a house is closely linked to marriage and the birth of children (Kulu, 2008; Feijten, et al., 2008; Wang, et al, 2018). Results also suggest that people prefer to live in the central city rather than in the surrounding area. Life in city centers differs from residential environments in the surrounding areas notably in the sense that cities have a high level of amenities. For instance, city centers create more opportunities for work and offer a wide range of facilities for leisure activities and social interaction. This makes the city center an attractive place to live. However, city centers have a high density of buildings and consequently make the city a crowded place, while surrounding areas tend to have more spacious and greener living environments. In addition, the interactions are insignificant; indicating that marital status in this study is not influencing taste variation. Moreover, although not all levels of distance to the metro station, bus station and shopping mall are significant, the estimation results still indicate that people tend to prefer living in an area close to these public facilities.

#### Interpretation of the job choice component

In terms of the job domain, our results indicate that people prefer to work in public institutions and government rather than in private firms or be self-employed. The reason may be that, in Shenyang city, stability is one of the main factors for individuals to choose job type. As expected, income has a positive impact on people's job choices. Moreover, the results show that males more strongly consider high income than females.

		MNL		MMNL	
Variable	Description	Estimate	S.E.	Estimate	S.E.
Random parameters					
Tenure	Buy 1; Rent -1	1.07***	.14	1.79***	.24
Housing size	Small	-1.44***	.19	-2.03***	.30
	Medium 1	13	.13	31	.13
	Medium 2	.26**	.13	.61***	.00
	Large	1.31***	.14	1.72***	.24
Home location	Central 1; Surrounding area -1	.17**	.07	.26**	.12
Salary	Low	-1.70***	.10	-2.83***	.21
	Medium 1	51***	.08	61***	.00
	Medium 1	.86***	.10	1.21***	.00
	High	1.35***	.10	2.24***	.19
Job location	Central 1; Surrounding area -1	.07	.06	.08	.09
Constant 1	Move house: by car	-2.41***	.19	-3.45***	.34
Constant 2	Move house: by metro	-2.91***	.21	-4.25***	.11
Constant 3	Move house: by bus	-3.38***	.23	-4.45***	.40
Constant 4	Move house: by bike	-4.10***	.27	-5.35***	.44
Constant 5	Move house: by walk	-3.91***	.25	-5.18***	.41
Constant 6	Change job: by car	-1.68***	.14	-2.46***	.25
Constant 7	Change job: by metro	-2.04***	.16	-2.81***	.27
Constant 8	Change job: by bus	-2.92***	.18	-3.68***	.27
Constant 9	Change job: by bike	-3.35***	.19	-4.56***	.33
Constant 10	Change job: by walk	-3.28***	.18	-4.30***	.31
Non-random parameters					
Housing price	Low	.81***	.11	1.35***	.17
	Medium 1	.04	.11	09	.16
	Medium 2	.06	.11	09	.15
	High	91***	.15	-1.18***	.21
Distance to metro station	Small	.36***	.14	.23	.21
	Medium 1	.25**	.12	.30	.19
	Medium 2	22	.14	18	.22
	Large	38**	.19	65	.32
Distance to bus stop	Small	.48***	.19	.66**	.28
	Medium 1	.63***	.19	.89***	.27

### Table 4.2 Estimation results of MNL model and mixed logit model

	Medium 2	33	.26	45	.36
	Large	78**	.36	-1.10	.51
Distance to shopping mall	Small	05	.18	15	.27
	Medium 1	.22	.16	.21	.24
	Medium 2	.19	.17	.53*	.28
	Large	36	.27	60	.41
Working type	Government	.42***	.09	.61***	.13
	Institution	.21***	.08	.32***	.12
	Company	18***	.07	29***	.11
	Self-employed	45***	.08	64***	.14
Flexible working time	Yes 1; No -1	01	.04	02	.07
Working environment	Very poor	78***	.11	-1.17***	.15
	Poor	10	.09	23*	.13
	Good	.41***	.07	.64***	.12
	Very good	.47***	.08	.76***	.14
Colleague relationship	Very poor	90***	.11	-1.30***	.16
	Poor	46***	.10	59***	.14
	Good	.68***	.08	.87***	.12
	Very good	.68***	.08	1.03***	.13
Easy to change job	Yes 1; No -1	05	.04	18	.07
Cost-car	Low	26	.28	04	.46
	Medium 1	.09	.20	11	.32
	Medium 2	.31	.21	.37	.37
	High	14	.22	22	.37
Travel time-car	Low	.03	.19	.05	.34
	Medium 1	.34**	.16	.59**	.28
	Medium 2	13	.18	04	.34
	High	30	.26	61	.45
Congestion time-car	Low	.37***	.10	.43***	.13
	Medium 1	07	.10	.00	.14
	Medium 2	14	.10	11	.15
	High	16	.11	32***	.16
Cost-metro	Low 1; High -1	.44***	.13	.58***	.20
Travel time-metro	Low	23**	.22	08	.33
	Medium 1	.21*	.11	.11	.15
	Medium 2	.22**	.11	.27*	.16
	High	20	.11	-1.23	.15

Out-of-vehicle time-metro	Low	.64***	.09	.91***	.12
	Medium 1	.13	.10	.18	.13
	Medium 2	16	.11	14	.14
	High	62***	.13	95***	.16
Have seats or not-metro	No 1; Yes -1	04	.06	11	.08
Cost-bus	Low 1; High -1	.23**	.09	.39***	.13
Travel time-bus	Low	.21	.17	.21	.22
	Medium 1	.83***	.12	.99***	.17
	Medium 2	.27**	.13	.51***	.18
	High	-1.31***	.22	-1.71***	.30
Out-of-vehicle time-bus	Low	.40***	.12	.41***	.16
	Medium 1	07	.11	02	.15
	Medium 2	08	.12	.05	.16
	High	25**	.13	44***	.18
Congestion time-bus	Low	.30**	.15	.33*	.22
	Medium 1	.07	.13	22	.16
	Medium 2	27**	.13	28	.18
	High	10	.13	.17	.17
Have seats or not-bus	No 1; Yes -1	07	.07	08	.09
Travel time-bike	Low	1.81***	.15	2.81***	.23
	Medium 1	.52**	.21	.46*	.279
	Medium 2	-1.14***	.36	-1.55***	.51
	High	-1.19***	.34	-1.72***	.57
Travel time-walk	Low	1.72***	.12	2.80***	.21
	Medium 1	40	.28	65	.41
	Medium 2	53*	.28	82**	.41
	High	78***	.28	-1.32***	.42
Interaction effects					
Age	<40 1, Older than 40 -1	.18**	.09	.15	.15
Gender	Male 1; Female -1	.12*	.07	.22**	.11
Marital status 1	Couple with children	08	.09	18	.16
Marital status 2	Couple without children	09	.12	23	.21
Tenure * Age		25**	.12	43**	.18
Tenure * Age Tenure * Marital status 1		25** 12	.12 .13	43** 35*	.18 .20
Tenure * Age Tenure * Marital status 1 Tenure * Marital status 2		25** 12 .28	.12 .13 .17	43** 35* .48*	.18 .20 .28
Tenure * Age Tenure * Marital status 1 Tenure * Marital status 2 Housing size 1 * Age		25** 12 .28 .51***	.12 .13 .17 .18	43** 35* .48* .35	.18 .20 .28 .28

Housing size 3 * Age	.13	.12	.12	.19
Housing size 4 * Age	29**	.12	13*	.19
Home location * Marital status 1	.02	.09	08	.14
Home location * Marital status 2	.05	.126	.08	.19
Salary 1 * Age	13	.09	24	.18
Salary 2 * Age	.03	.08	.26*	.14
Salary 3 * Age	12	.09	12	.13
Salary 4 * Age	.22**	.10	.10	.16
Salary 1 * Gender	.04	.07	.00	.14
Salary 2 * Gender	24***	.07	44***	.13

Standard deviation of the random parameters

Tenure		.52***	.20
Housing size 1		.87***	.28
Housing size 2		1.27***	.18
Housing size 3		.58***	.20
Housing size 4		1.52***	.34
Home location		.48***	.15
Salary 1		1.69***	.15
Salary 2		1.14***	.15
Salary 3		.48**	.22
Salary 4		1.11***	.19
Job location		.02	.20
Constant 1 (Move house by car)		1.31***	.19
Constant 2 (Move house by metro)		1.25***	.27
Constant 3 (Move house by bus)		1.30***	.39
Constant 4 (Move house by bike)		.84	.54
Constant 5 (Move house by walk)		.59	.43
Constant 6 (Move house by car)		2.15***	.17
Constant 7 (Change job by metro)		1.50***	.16
Constant 8 (Change job by bus)		.81***	.17
Constant 9 (Change job by bike)		1.14***	.19
Constant 10 (Change job by walk)		1.25***	.38
Rho2	0.505	0.560	
Rho2 adjusted	0.488	0.539	

\*\*\*, \*\*, \* means significance at 1%, 5% and 10% level.

Chapter 4
-----------

Mode	Car	Metro	Bus			
VoT (CNY / hour)	27.96	18.48	17.23			
VoT (€/ hour)	3.78	2.49	2.33			

Table 4.3 Value of time of different transportation modes

Similarly, a job with a very high salary attracts young people more, while jobs with an extremely low salary are less attractive for young people. Moreover, the coefficients of working environment and co-worker relationship indicate that people have a strong preference for a good relationship with co-workers and working environment. It indicates that not only a high salary, but also good harmony with co-workers will influence people's job choice. The coefficient for flexible work time is shown insignificant, which indicates that compared to Western cities, workers in China care less about the working hours flexibility. Lastly, although insignificant, the results show that in general, individuals prefer to work in the central city. The standard deviation of the random coefficient of job location is insignificant, suggesting that there is no significant unobserved variance in the population in the sensitivity of job location in the job choice process. Couples without children are found to be more sensitive to finding a job in the central city.

#### Interpretation of the travel mobility decision component

As expected, time-related variables indicate excess commute time/congestion time/ out of vehicle time have significant negative influence on the travel mobility decision of different transportation modes. Furthermore, results indicate that sensitivity to travel cost of different transportation modes is not the same. In general, individuals are less sensitive to travel cost when commuting by car. Conversely, opposite effects are found for the public transportation modes. Lastly, taking having seats or not into consideration, although not significant, the negative sign reveals that improvement of comfort on public transportation might attract individuals to metro and/or bus as the commute mode.

Considering the importance of practical relevance of the estimation results, the values of time are calculated (see Table 4.3). The value of commuting time is about 27.96 CNY per hour (3.78  $\in$ / hour) for car, 18.48 CNY per hour (2.49  $\in$ / hour) for metro, and 17.23 CNY per hour (2.33  $\in$ / hour) for bus. Most studies are central to the evaluation of transportation in the context of short-term mobility decisions. In turn, the values of time for all transportation modes in this long-term residential and job mobility

context are quite low. The implications are in line with previous research, focusing on long-term mobility decisions (Pérez, et al., 2003; Kim, et al., 2005; Tillema, et al., 2010; Peer, et al., 2015; Dubernet and Axhausen, 2016; Beck, et al., 2017), where commuting time is found less important in the context of long-term decisions.

Comparing the alternative-specific constants, the base utility of the 'keeping the current' alternative has the highest utility compared to the other alternatives. It suggests that the respondents are relatively satisfied with their current situation. Only the job/house exceeds their expectation thresholds, individuals are more likely to consider to change. Findings also show that respondents are less inclined to move house relative to change job. Taken all alternative-specific constants as random parameters, unobserved heterogeneity is shown significant for all base preferences.

In general, coefficients of various dimensions in the MNL and mixed logit model have the same signs. Interestingly, some significant coefficients in the MNL model became insignificant in the mixed logit model. The reduction in the number of parameters may be due to the relative increase in the number of parameters to estimate as well as the introduction of more random parameters.

#### 4.3.2 Capturing taste variation through a latent class model

To examine heterogeneity in preferences, a latent class model was estimated. In order to identify the optimal number of classes, the AIC values for the models without membership specification were calculated. The LCM considered 2 to 5 classes. However, due to the limitation in sample size and relatively large number of attributes, models with more than three classes could not be estimated. Table 4.4 reports the AIC and BIC values for 2 and 3 classes. Although the AIC of the 3-class model is smaller than that of the 2-class model, the difference is small. The model with two classes was therefore chosen for further analysis.

Indicators	2 classes	3 classes
AIC	5470.3	5369.3
AIC/N	2.399	2.355
BIC	5505.9	5417.0
BIC/N	2.415	2.376

Table 4.4 AIC and BIC	value of the l	base models with	different number	of classes
-----------------------	----------------	------------------	------------------	------------

The results of LCM and MNL are presented in Table 4.5. It confirms that the LCM has a higher goodness-of-fit than the MNL model, where the adjusted Rho-squared increases from 0.484 for the base MNL model to 0.518. On the whole, class 1 contains 56.7% of the cases, while class 2 contains 43.3%. Because the second class is chosen as the reference, positive parameters of the membership variables relate to class 1 while negative parameters relate to class 2.

An important issue when specifying a latent class choice model is the specification of the variables in the class membership. To identify the membership of each class, socio-demographic characteristics were introduced in the membership functions, which included age, gender and income. Results of the estimation (Table 4.5) indicate that income has a significant effect on membership probability at the 95% level, providing evidence that high-income people are more likely to belong to class 2.

Taking a closer look at the differences and similarities between the two classes, we see that the effects of housing-related attributes are homogeneous between the two classes. Both housing price and housing size have a substantial impact on households' residential mobility in the sense that people prefer larger and cheaper houses. In addition, individuals/households in both classes prefer to buy a house to renting one. The taste variation between the two classes is reflected in the factors of the built environment. The effect of 'distance to bus stop' in the first class is larger than that in the second class, which indicates that lower-income households may take shorter trips and are therefore more likely to use buses. Similarly, the results indicate that people in class 1 (mostly lower income) put more weight on locations that provide greater accessibility to shopping malls; it may because that individuals with a higher income buy more often online than individuals with a lower income, which motivates the former make less shopping trips. Moreover, variety of travel modes are used for shopping trips. High-income households are more likely to own and use a private car than low-income households. Consequently, people in class 2 may be less likely to consider the distance to shopping malls. On the other hand, the attribute 'distance to metro station' shows a larger effect in the second class, indicating that housing with good accessibility to a metro station is more important for people in the second class. It is understandable that in contrast with 'distance to metro station', there is a strong evidence that proximity to metro stations can uplift house price (Du and Mulley, 2006), which may cause households with low income being unable to buy such houses.

		MNL		LCM Class 1		LCM Class 2	
Variable	Description	Estimate	S.E.	Estimate	S.E.	Estimate	S.E.
Tenure	Buy 1; Rent -1	.78***	.07	.77***	.20	1.02***	.19
Housing price	Low	.79***	.10	.97***	.28	.86**	.34
	Medium 1	.04	.10	.03	.21	.14	.54
	Medium 2	.06	.11	05	.25	.07	.80
	High	89***	.15	95***	.33	-1.08***	.33
Housing size	Small	-1.13***	.14	-1.37***	.45	-1.35**	.62
	Medium 1	36***	.10	41**	.19	27	.31
	Medium 2	.30***	.10	.20	.23	.53	.28
	Large	1.19***	.12	1.58***	.32	1.08***	.28
Distance to metro	Small	.36***	.13	.31	.27	.82***	.30
station	Medium 1	.24**	.12	.44	.32	.25	.24
	Medium 2	19	.14	57	.36	.15	.35
	Large	41**	.18	18**	.33	-1.23	.57
Distance to bus	Small	.43**	.18	1.11*	.61	01	.45
stop	Medium 1	.55***	.18	.81	.71	.76	.58
	Medium 2	31	.25	75	.85	20	.76
	Large	66*	.34	-1.17	1.37	54	.98
Distance to	Small	07	.17	44	.67	.36	.51
shopping mall	Medium 1	.23	.16	.57	.55	17	.46
	Medium 2	.20	.17	.32	.45	07	.66
	Large	36	.26	44	1.09	12	1.04
Home location	Central 1; Surrounding area -1	.13**	.06	.26	.17	.05	.16
Working type	Government	.40***	.08	.43**	.19	.73***	.22
	Institution	.26***	.08	.18	.18	.28	.19
	Company	20***	.07	08	.12	42**	.17
	Self-employed	47***	.08	53***	.15	59**	.19
Flexible working time	Yes 1; No -1	02	.04	13	.20	.06	.10
Salary	Low	-1.64***	.09	-2.15***	.26	-2.58***	.29
	Medium 1	52***	.08	.19**	.08	66***	.16
	Medium 1	.87***	.09	.91***	.21	1.26***	.21
	High	1.29***	.09	1.11***	.23	1.98***	.26
Working	Very poor	79***	.11	-1.05***	.25	83***	.27

#### Table 4.5 Estimation results of the MNL model and the latent class model

environment	Poor	10	.09	15	.15	21**	.09
	Good	.40***	.07	.40***	.15	.36*	.20
	Very good	.48***	.08	.79***	.18	.67***	.22
Colleague	Very poor	88***	.11	-1.50***	.27	68**	.30
relationship	Poor	45***	.10	53***	.18	48*	.27
	Good	.65***	.08	.93***	.16	.64***	.24
	Very good	1.15***	.08	1.10***	.15	.52*	.21
Easy to change job	Yes 1; No -1	04	.04	.09	.08	20	.20
Job location	Central 1; Surrounding area -1	.05	.04	.01	.09	.09	.12
Cost-car	Low	13	.28	62	.65	3.12***	.62
	Medium 1	.07	.19	.56	.58	-1.24***	.42
	Medium 2	.22	.21	.22	.78	58*	.35
	High	16	.21	16	.65	-1.30***	.54
Travel time-car	Low	.14	.19	.20	.55	1.15**	.48
	Medium 1	.34**	.15	.14	.56	1.55***	.29
	Medium 2	04	.18	01	.52	1.05***	.29
	High	44*	.26	33	.58	-3.75***	.63
Congestion time-	Low	.40***	.09	.71**	.32	.40*	.22
car	Medium 1	08	.10	.43	.40	.35*	.18
	Medium 2	11	.10	.54*	.28	38	.26
	High	21**	.10	-1.67***	.42	37*	.20
Cost by metro	Low 1; High -1	.39***	.13	.11	.21	49	.50
Travel time-metro	Low	.21**	.11	.39**	.19	.82**	.32
	Medium 1	.20*	.11	.00	.21	.19	.37
	Medium 2	17	.10	.19*	.19	39	.27
	High	25	.22	58*	.31	62	.83
Out-of-vehicle	Low	.59***	.09	.28	.19	1.11***	.27
time-metro	Medium 1	.13	.01	.11	.20	.30	.24
	Medium 2	16	.11	07	.19	36	.52
	High	56***	.12	33	.20	-1.04**	.48
Have seats-metro	No 1; Yes -1	03	.06	00	.11	03	.16
Cost-bus	Low 1; High -1	.25***	.09	.29*	.17	22	.34
Travel time-bus	Low	.14	.17	02	.26	.20	.92
	Medium 1	.78***	.12	.66***	.20	1.39***	.48
	Medium 2	.29**	.13	.22	.22	.11	.69
	High	-1.22***	.21	85**	.34	-1.71**	.95

Out-of-vehicle	Low	.42***	.11	.60***	.18	.11	.44	
time-bus	Medium 1	10	.11	28	.20	.29	.42	
	Medium 2	10	.12	01	.19	12	.45	
	High	22*	.13	31	.21	29	.48	
Congestion time-	Low	.31**	.15	.22	.22	.91	.82	
bus	Medium 1	.09	.13	.13	.23	26	.50	
	Medium 2	26**	.13	31	.22	32	.50	
	High	15	.13	04	.23	33	.69	
Have seats- bus	No 1; Yes -1	06	.07	08	.14	05	.27	
Travel time-bike	Low	1.78***	.15	2.07***	.33	2.05***	.52	
	Medium 1	.48**	.20	.60	31	54	1.51	
	Medium 2	-1.12***	.35	-1.53	1.65	40	1.64	
	High	-1.14***	.34	-1.15	1.41	-1.11	2.71	
Travel time-walk	Low	1.69***	.12	2.54***	.38	1.86***	.31	
	Medium 1	40	.28	62	1.47	49	1.36	
	Medium 2	54**	.28	95	2.27	33	1.82	
	High	78**	.28	97	1.83	-1.04	2.54	
Constant 1 (Move h	ouse by car)	-2.37***	.17	-3.60***	.40	-1.21***	.38	
Constant 2 (Move h	ouse by metro)	-2.82***	.20	-3.42***	.51	-2.02***	.45	
Constant 3 (Move house by bus)		-3.32***	.22	-3.56***	.47	-2.41***	.61	
Constant 4 (Move house by bike)		-4.02***	.25	-4.12***	.49	-3.95***	.90	
Constant 5 (Move house by walk)		-3.84***	.24	-4.33***	.53	-3.79***	.82	
Constant 6 (Change	ijob: by car)	-1.60***	.13	-3.91***	.38	-1.20***	.31	
Constant 7 (Change	i job by metro)	-1.92***	.15	-2.37***	.33	-1.65***	.40	
Constant 8 (Change job by bus)		-2.84***	.16	-3.32***	.35	-2.31***	.48	
Constant 9 (Change	i job by bike)	-3.25***	.18	-4.01***	.38	-2.85***	.51	
Constant 10 (Chang	e job by walk)	-3.19***	.17	-4.54	.53	-2.52	.45	
Rho2		.498	.498 .547		47			
Rho2 adjusted		.484		.518				
Membership variables (Latent class 1)		Estima	te		S	.Е.		
Constant		.36* .21		21				
Gender (Male 1, Female -1)		15	.15 .23					
Age (Young 1, Old -1)		47	47 .29					
Income (Low 1; High -1)		.92*** .26		26				

\*\*\*, \*\*, \* means significance at 1%, 5% and 10% level.

As for job-related variables, results show that people in the second class with higher income are more concerned about salary. People in the first class favour good co-worker relationships and working environments more than people in the second class. One explanation may be that lower income workers may have lower aspirations about a high salary because these jobs are limited. Moreover, attitudes may induce job self-selection in the sense that people who expect more opportunities for promotion may consciously choose to work with more duties and responsibilities, and thus get better paid.

As for the utility of transportation modes, time-related variables show different effects between the two classes. Some existing studies have confirmed that people with higher income have shorter travel time and prefer to live closer to their work places, while other studies reported the opposite findings and argued that commute distance tends to be positively correlated with earnings. Wealthier people have longer commuting times (Blumenberg, 2004; Sandowa and Westin, 2010), since their desire for a better quality of life and larger houses can be more easily fulfilled by living far from the city center and their workplaces (Dargay and van Ommeren, 2005). In line with the former argument, results of the current study indicate that the second class has larger coefficients for travel time than the first class. This means that people with higher income are more sensitive to travel time than people with lower income. In this regard, once provided by a dream job offer, low-income people may consider accepting it, even if it may take longer time to commute.

Similarly, regarding out-of-vehicle time, results of metro related attributes show that people in class 2 (mostly higher income) consider metro more important than people with lower income. However, an opposite effect is found for the bus mode. In other words, people in class 2 are more sensitive to the distance to metro stations, but less sensitive to the distance to bus stops compared to people in class 1. In addition, travel cost by car is found significant only in the second class, which means that people with higher income may be concerned more with the travel cost by car. However, such difference between the two classes is not found for public transportation modes.

# 4.3.3 Capturing taste variations through an error component mixed logit model

To reflect the structure of the collected data and incorporate different sources of heterogeneity, we formulated an error component mixed logit model. The model follows the principles discussed in Greene and Hensher (2007). Assume that individual n faces

the choice among multidimensional alternatives in choice situation *t*. The multidimensionality of the alternatives relates to the residential environment, job and a set of transportation modes for the work commute. Individuals are assumed to derive from these multidimensional profiles a certain utility. The utility of alternative *i* is assumed to be stochastic and consists of a deterministic utility,  $V_{nit}$ , and a random error,  $\varepsilon_{nit}$ , such that

$$U_{nit} = V_{nit} + \varepsilon_{nit} \tag{1}$$

The deterministic utility is assumed to be a linear function of observed attributes. Hence, the utility function may be expressed as

$$U_{nit} = \beta_{io} + \sum_{k=1}^{K} \beta_k x_{nikt} + \varepsilon_{nit}$$
<sup>(2)</sup>

where,  $\beta_k$  is the parameter for attribute k,  $x_{nit}$  is an explanatory variable, related to attributes of the residence, job or transportation modes for commuting, and  $\varepsilon_{nit}$  is a random IID Gumbel distributed error term. We assume that individuals choose the multidimensional alternative that maximizes their utility.

We introduce various sources of preference heterogeneity in the model. First, we assume that people differ in terms of taste parameters  $\beta_k$ . In part, the unobserved heterogeneity in taste parameters can be accounted for by differences in socio-demographics, and in part there is pure error. Thus,

$$\beta_{nk} = \bar{\beta}_k + \sigma_k + \sum_{m=1}^{M} \theta_{km} z_{nm} , k = 1, ..., K$$
(3)

where  $\bar{\beta}_k$  is the mean of the random parameter of attribute *k*.  $z_{nm}$  is the  $m^{\text{th}}$  sociodemographic characteristic of individual n,  $\theta_{km}$  is a parameter to be estimated, and  $\sigma_k$  denotes the standard deviation of random parameter  $\beta_{nk}$ . Then, the utility expression becomes

$$U_{nit} = \beta_{io} + \sum_{k=1}^{K} (\bar{\beta}_k + \sigma_k + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{nikt} + \varepsilon_{nit}$$
(4)

Our experiment induced respondents to choose among three long-term life trajectory decisions, i.e. status quo (keep current), move house or change job. To allow for the possibility that unobserved heterogeneity underlying the choice of these long-term decisions may affect preferences for transportation modes, we additionally allow for three additional error components in each nested alternative, defined by the outcome of the long-term decision. The error components,  $Q_{n1}$ ,  $Q_{n2}$  and  $Q_{n3}$ , denote

the grouped error component of keep current, move house and change job. Thus, the utility function becomes

Status quo:

$$U_{n1t} = \beta_{1o} + \sum_{k=1}^{K} (\bar{\beta}_k + \sigma_k + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{n1kt} + \lambda_1 Q_{n1} + \varepsilon_{n1t}$$

Move house:

$$U_{n2t} = \beta_{2o} + \sum_{k=1}^{K} (\bar{\beta}_{k} + \sigma_{k} + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{n2kt} + \lambda_{2} Q_{n2} + \varepsilon_{n2t}$$
  
...  
$$U_{n6t} = \beta_{6o} + \sum_{k=1}^{K} (\bar{\beta}_{k} + \sigma_{k} + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{n6kt} + \lambda_{2} Q_{n2} + \varepsilon_{n6t}$$

Change job:

...

$$U_{n7t} = \beta_{7o} + \sum_{k=1}^{K} (\bar{\beta}_k + \sigma_k + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{n7kt} + \lambda_3 Q_{n3} + \varepsilon_{n7t}$$

$$U_{n11t} = \beta_{11o} + \sum_{k=1}^{K} (\bar{\beta}_k + \sigma_k + \sum_{m=1}^{M} \theta_{km} z_{nm}) x_{n11kt} + \lambda_3 Q_{n3} + \varepsilon_{n11t}$$
(5)

We assume that these error components  $Q_{nl}$  are independent and follow a standard normal distribution, with  $Q_{nl} \sim N[0,1]$ .  $\lambda_l$  is a parameter to be estimated for the error component *l*.

Finally, we specify the variance heterogeneity of the error component as

$$\operatorname{Var}[Q_{nl}] = \left[\lambda_l \exp \sum_{m=1}^{M} \tau_{lm} \, z_{nm}\right]^2 \tag{6}$$

where  $\tau_{\mathit{lm}}$  is a parameter that needs to be estimated.

The conditional choice probabilities then take on the logit form:

$$p_{nt}(i|\boldsymbol{Q}_{n}) = \frac{\exp(\beta_{io} + \sum_{k=1}^{K} (\bar{\beta}_{k} + \sigma_{k} + \sum_{m=1}^{M} \theta_{km} \, z_{nm} \,) \, x_{nikt} + \lambda_{l} Q_{nl} \,)}{\sum_{i=1}^{l} \exp(\beta_{io} + \sum_{k=1}^{K} (\bar{\beta}_{k} + \sigma_{k} + \sum_{m=1}^{M} \theta_{km} \, z_{nm} \,) \, x_{nikt} + \lambda_{l} Q_{nl} \,)}$$
(7)

Conditioned on the error components, the unconditional probability of a choice for alternative *i* for individual n is

$$p_{nit} = \int_{E_n} (p_{nt}(i|\boldsymbol{Q}_n) f(\boldsymbol{Q}_{nl})) \, d\boldsymbol{Q}_{nl}$$
(8)

Integrating the taste variation of random parameter  $\beta_{nk}$ , the unconditional choice in equation (7) denotes as

$$p_{nit} = \int_{\boldsymbol{Q}_{nl}} \int_{\boldsymbol{\beta}_n} \left( (P_{nit} | \boldsymbol{Q}_{nl}, \boldsymbol{\beta}_n) f(\boldsymbol{Q}_{nl}, \boldsymbol{\beta}_n | \boldsymbol{z}_n) \right) d\boldsymbol{Q}_{nl} d\boldsymbol{\beta}_n$$
(9)

In our experiment, respondents provide multiple responses. Thus, we estimated these panel effects. Consequently, the full log-likelihood function can be written as follows

$$LL = \sum_{n=1}^{N} \log \int_{\boldsymbol{Q}_{nl}} \int_{\boldsymbol{\beta}_{n}} \prod_{t=1}^{T} \left( (P_{nit} | \boldsymbol{Q}_{nl'} \boldsymbol{\beta}_{n}) f(\boldsymbol{Q}_{nl'} \boldsymbol{\beta}_{n} | \boldsymbol{z}_{n}) \right) d\boldsymbol{Q}_{nl} d\boldsymbol{\beta}_{n}$$
(10)

Table 4.7 presents the results of the error component mixed logit model. Because we have a large number of attributes, it is unrealistic for the given sample size to estimate random effects for all attributes. From a behavioral point of view, understanding and modelling residential and job location choice behavior is a primary concern for urban planners, policymakers, and researchers (Schirmer, et al., 2014; Dubernet and Axhausen, 2016). Thus, residential and job location were selected as the random attributes. In addition, not only the main effects but also several sociodemographic interaction effects were estimated.

Nlogit software was used to estimate the error component mixed logit model. Both random variables and error components were assumed to be normally distributed. To get stable estimates, we systematically checked the convergence of the parameter estimates from 100 to 5000 random Halton draws. The final estimates reported in this study are based on 500 Halton draws since increasing the number of draws did not largely improve the estimates. Overall, the estimated error component logit model has a good fit. The error component model improves the adjusted Rho-squared from 0.495 for the base random parameter model to 0.53.

Results in Table 4.7 indicate that parameter signs for both the error component mixed logit model and the base random parameter model are almost the same. Additionally, compared with the base random parameter mixed logit model, attributes were found less statistically significant for the error component mixed logit model. Since most taste parameters in the error component model have the same direction as in the base multinomial logit model and mixed logit model, we discuss the results by focusing mainly on the error components.

Results of the three alternative-specific error components (keep current, move house or change job) show that the standard deviation of moving house and changing job are statistically significant. It suggests that unobserved heterogeneity exists in preferences for transportation modes, dependent upon moving house and changing job decisions that are not fully captured by the alternative-specific parameter estimates of the attributes of the different transportation modes. This evidence indicates that a 'nested structure' exists in the process of transportation mode choice when people decide to move house and change job. Furthermore, we allowed for heteroscedasticity in the variance of the error components as a function of socio-demographics. Because the number of parameters relative to sample size does not allow estimating too many interaction effects, after exploring different combinations of socio-demographic variables, as listed in Table 4.7, gender, income and numbers of workers in the households were taken into consideration.

In case of moving house, the negative value of income indicates that as income decreases, the standard deviation of the error component decreases as well, leading to a reduction in preference heterogeneity from these unobserved effects for individuals with higher salary when they decide to move house. Moreover, the number of workers in the household shows a significant effect on the heterogeneity of moving house. Specifically, results indicate that households with single workers are more heterogeneous when they decide to move house. Similarly, as to job choice, income and numbers of workers in the household have a significant influence on preference heterogeneity for this long-term decision. Our results show that as the income and/or number of workers increases, the standard deviation of the error component decreases, implying a reduction in preference heterogeneity in these unobserved effects.

		MNL		ECL	
Variable	Description	Estimate	S.E.	Estimate	S.E.
Random parameter					
Home location	Central 1; Surrounding area-1	.23**	.09	.11	.14
Job location	Central 1; Surrounding area -1	.09	.07	.09	.10
Non-Random parameters					
Tenure	Buy 1; Rent -1	.94***	.09	1.36***	.18
Housing cost	Low	.97***	.12	1.26***	.22
	Medium 1	04	.12	.03	.21
	Medium 2	04	.12	13	.22
	High	89***	.17	-1.15***	.23
Housing size	Small	-1.37***	.17	-2.15**	.46
	Medium 1	38***	.12	52**	.21
	Medium 2	.39***	.12	.60***	.21
	Large	1.36***	.14	2.07***	.30
Distance to metro station	Small	.40**	.16	.20	.32
	Medium 1	.17	.14	.48	.30
	Medium 2	22	.16	24	.33
	Large	34	.21	45	.53
Distance to bus stop	Small	.43**	.21	.60	.40
	Medium 1	.62***	.21	.77*	.44
	Medium 2	38	.28	45	.52
	Large	67*	.41	92	.83
Distance to shopping mall	Small	.09	.20	.01	.46
	Medium 1	.23	.18	.27	.40
	Medium 2	.30	.19	.50	.45
	Large	62	.30	78	.88
Work type	Government	.45***	.10	.55***	.15
	Institution	.27***	.09	.29**	.15
	Company	23***	.08	27**	.14
	Self-employed	48***	.09	.57***	.14
Flexible working time	Yes 1; No -1	02	.05	05	.08
Salary	Low	-1.87***	.11	-2.88***	.21
	Medium 1	58***	.09	36***	.13
	Medium 1	1.01***	.10	1.38***	.13

#### Table 4.7 Estimation results of the MNL and error components mixed logit model

	High	1.44***	.11	1.85***	.15
Work environment	Very poor	91***	.12	-1.11***	.19
	Poor	14	.10	27*	.14
	Good	.45***	.09	.61***	.15
	Very good	.59***	.09	.77***	.14
Colleague relationship	Very poor	-1.09***	.13	-1.36***	.18
	Poor	49***	.11	66***	.16
	Good	.81***	.09	.99***	.15
	Very good	.76***	.09	1.04***	.14
Easy to change job	Yes 1; No -1	04	.05	04	.09
Cost-car	Low	06	.33	01	.64
	Medium 1	.14	.22	12	.49
	Medium 1	.19	.24	.44	.58
	High	27	.25	32	.58
Travel time-car	Low	.18	.22	.13	.56
	Medium 1	.31*	.19	.49	.47
	Medium 1	02	.22	27	.54
	High	48	.32	35	.64
Congestion time-car	Low	.40***	.10	.40***	.13
	Medium 1	05	.11	01	.15
	Medium 1	11	.11	07	.15
	High	25**	.12	32	.18
Cost-metro	Low 1; High -1	.22*	.12	.23	.27
Travel time-metro	Low	.28**	.12	.58**	.23
	Medium 1	.20	.12	.60**	.26
	Medium 1	21	.12	.06	.17
	High	27	.25	-1.23*	.65
Out-of-vehicle time-metro	Low	.66***	.10	.76***	.13
	Medium 1	.14	.10	.09	.15
	Medium 1	17	.11	19	.17
	High	63***	.13	66***	.18
Have seats or not-metro	No 1; Yes -1	05	.06	10	.09
Cost by bus	Low 1; High -1	.28***	.10	.26	.18
Travel time-bus	Low	.14	.18	.07	.25
	Medium 1	.88***	.13	1.12***	.22
	Medium 1	.33**	.14	.51*	.25
	High	-1.35***	.23	-1.7***	.39

Out-of-vehicle time-bus	Low	.43***	.12	.54***	.19
	Medium 1	.09	.12	.10	.23
	Medium 1	07	.13	16	.22
	High	45*	.14	47*	.24
Congestion time-bus	Low	.32*	.17	.16	.28
	Medium 1	.11	.14	.26	.22
	Medium 1	29**	.14	24	.22
	High	14	.14	18	.26
Have seats or not-bus	No 1; Yes -1	05	.08	16	.13
Travel time-bike	Low	2.02***	.17	2.80***	.45
	Medium 1	.56**	.23	.66	.48
	Medium 1	-1.27***	.401	-1.84	1.89
	High	-1.31***	.379	-1.62	1.39
Travel time-walk	Low	1.91***	.14	1.19***	.14
	Medium 1	41	.30	18	.26
	Medium 1	61**	.30	40	.30
	High	90***	.31	61*	.30
Constant 1	Move house by car	-2.50***	.19	-2.96***	.38
Constant 2	Move house by metro	-3.00***	.22	-3.19***	.50
Constant 3	Move house by bus	-3.51***	.24	-4.15***	.43
Constant 4	Move house by bike	-4.32***	.28	-5.28***	.60
Constant 5	Move house by walk	-4.12***	.26	-5.54***	.66
Constant 6	Change job by car	-1.64***	.15	-2.12***	. 31
Constant 7	Change job by metro	-1.99***	.17	-2.23***	.37
Constant 8	Change job by bus	-2.91***	.18	-3.55***	.36
Constant 9	Change job by bike	-3.41***	.20	-4.37***	.45
Constant 10	Change job by walk	-3.34***	.19	-4.73***	.59
Heterogeneity around m	nean				
Home location: number of	workers	01	.08	02	.13
Job location: number of workers16** .07			13	.10	
Standard deviation of ra	indom parameters				
Home location	Central 1; Surrounding area -1	.68***	.12	.01	.62
Job location	Central 1; Surrounding area -1	.82***	.08	.49**	.19
Error components for al	ternatives and nests of alternative	es parameters	5		
Standard deviationCurrent			.32	.40	
Standard deviationMove house			2.08***	.48	

Standard deviationChange job				.31
Heterogeneity around standard deviation of error components effect				
Current	Gender		.10	.15
	Income		.36	.34
	Numbers of workers		.13	.17
Move house	Gender		20	.14
	Income		43**	.18
	Numbers of workers		. 29*	.15
Change job	Gender		04	.11
	Income		30**	.14
	Numbers of workers		.30**	.14
Rho <sup>2</sup>		0.51	0.55	
Rho <sup>2</sup> adjusted		0.450	0.53	

\*\*\*, \*\*, \* means significance at 1%, 5% and 10% level.

#### 4.4 Conclusions and discussion

This chapter is based on the contention that individuals and households in response to dramatic change in their decision context maximize the combined utility of housing, job and commuting trips as opposed to maximizing their utility of one of these domains, treating the other dimensions as given, as many models assume. To support this contention, a stated choice experiment was designed to mimic the multidimensional choice behavior of interest. Furthermore, because a considerable amount of observed and unobserved preference heterogeneity normally exists in multi-dimensional choice behavior, this study investigated the heterogeneous behavior of people in the context of co-dependent decisions about residence, job and transportation modes using the data collected from a pivoted stated choice experiment. A mixed logit model, a latent class model and an error component model were estimated to identify the taste variation.

Results of mixed logit model estimation show that different choice dimensions considered in this paper are interrelated, both through directly observed relationships and through correlations across unobserved factors (error terms) affecting multiple choice dimensions. In terms of the housing dimension, tenure, size, price, distance to the bus stop and location were found to be important characteristics that help explaining the residential mobility choice process. In general, the old generation and households with one or more children are more likely to consider buying their own house rather than renting one. Moreover, relative to young people, old residents prefer larger houses instead of small houses. In terms of the job dimension, salary, job type, co-worker relationships and work environment were found to be the significant factors in the job mobility choice process. In general, males are more likely to choose jobs with a high salary (have the better paying jobs). Similarly, a job with a very high salary attracts young people more, while an extremely low salary makes young people more reluctant to accept it. Relative to other households, couples with no children prefer jobs in the central city. In case of the commute mode dimension, estimation results show significant presence of time-related factors affecting travel mobility dimension. Moreover, effects of commute cost on various transportation modes differ. Specifically, results indicate that public transportation modes are sensitive to commute cost but the car mode is not.

Results of the latent class model show that people with lower income are less likely to change their current house and/or job than people with higher income. In this regard, policies such as supplying affordable housing for low-income households or training working skills for non-professional workers will provide more alternative choices and increase their aspirations to find a better a house/job. Moreover, by exploring heterogeneity among individuals, results show that people with lower income have a higher probability to use public transportation for commuting than private cars, while a reverse effect is found for people with higher income. Increasing the accessibility of public transportation (especially for metro) in new areas would attract lower-income households more to the public transport system. Likewise, in terms of slow modes, results indicate that people with a lower income are more likely to cycle than walk, while a reserve effect is found for people with higher income. Thus, building bicyclefriendly environments may encourage people to commute more by bicycle.

Results of the error component mixed logit model show that not only the random parameters but also the variances associated with the long-term alternativespecific error components are significant. Results of the specified error components for the long-term decisions indicate that unobserved heterogeneity exists in transportation mode choice, dependent upon moving house and changing job decisions that are not fully captured by the alternative-specific attributes of different transportation modes.

Although the co-dependency of multidimensional long-term mobility choices has been addressed in this chapter, it should be noted that most long-term decisions are not static considering the various change over life courses. It still needs to further investigate the long-term mobility decisions from a longitudinal perspective. As a natural extension, the temporal interdependencies between different long-term mobility decisions will be examined in the next chapter.

# 5

# Simultaneous Mobility Decisions

#### 5.1 Introduction

To complement the dominant single-domain research in transportation, different models for the co-dependent choice of various life domains were proposed and estimated in previous chapters. Although the multidimensionality was considered in these choice models, the interdependencies between different mobility decisions were not examined. Therefore, the analysis reported in this chapter is motivated by the desire to better understand the temporal interdependencies between residential choice, job choice and car ownership change decision.

Moreover, there is a growing body of studies indicating that changes in the demographic profile of individuals/households, such as change in household composition, may trigger households to reconsider their status in each domain (Clark, et al., 2003; Lanzendorf, 2010; Prillwitz, et al., 2006; Habib, et al., 2011; Oakil, et al., 2014; Zhao and Zhang, 2018). Consequently, incorporating life trajectory choices in the choice of car ownership is expected to provide a better understanding of the underlying decision processes.

During the last decades, many studies have indicated that qualitative views and desires (value orientations, individual-specific attitudes and lifestyle preferences) may simultaneously influence various long-term decisions. Whereas some authors try to

measure these concepts, others (e.g., Paleti, et al., 2012) argue they are difficult to quantify directly. Thus, simultaneously considering choices in various life dimensions increases the needs of controlling for these unknown factors and their correlation among different decision processes (Michielin and Mulder, 2008). For instance, Manaugh, et al. (2010) argued that commuting behavior is correlated with socio-demographic characteristics and unobserved factors (e.g., lifestyle, environmental concerns, perception of risk related to road accidents, etc.). These unobserved factors then affect both car travel distance and home-job location choice. Similarly, Eluru, et al. (2011) and Pinjari, et al. (2011) proposed integrated multi-dimensional choice models that tie together long-term location choices and short-term activity-travel choices, and suggested that the only way to accurately reflect their impacts and capture the 'bundling' of choices is to model the choice dimensions together in a joint modelling equation framework that accounts for correlated unobserved effects.

Furthermore, it is assumed that most life course events have lasting effects on future decision-making, since these decisions involve investment of time and money (Beige and Axhausen, 2017). To understand how current states, past decisions and future expectations influence individual decisions, a life course approach will be introduced into this chapter to investigate the life course decisions from a long-term perspective.

Thus, following this stream of research, the objective of this chapter is threefold. First, it will look into the interdependencies between the decisions to change residence, job and car ownership. Second, to reflect the impacts of unobserved factors on the various mobility decisions, a simultaneous equation model is built and discussed in this chapter. Third, since most long-term mobility decisions are in adjustment to past events and/or in anticipation of future events, the concurrent, lagged and lead effects are incorporated in the model.

The remainder of this chapter first provides a short review of the relevant literature. Section 5.3 describes the characteristics of the retrospective data used in the analysis. Next, the simultaneous equation model is presented, while model estimation results and interpretations are presented in Section 5.5. Finally, conclusions are presented in the last section.

#### 5.2 Literature review

A growing body of studies in life course approach indicates that qualitative views and desires, such as individual attitudes and lifestyle references, may simultaneously

influence various long-term decisions, which increases the need of controlling for unobserved factors and their correlation among different decision-making processes. For instance, Kan (2000) developed an econometric model to study the intricate relationship between residential mobility and tenure choice. Four variables (current tenure choice, actual mobility, mobility expectation, and previous tenure mode) are modeled as a simultaneous equation system. It assumes that a tenure choice decision is observed only if there is a move, the decision depends on previous tenure status, and the tenure mode and future length of stay are determined simultaneously. Based on the Florida subsample of the 2009 National Household Travel Survey of the United States, Kortum, et al. (2012) applied a simultaneous equation model to explore the primary reason for moving. The results showed significant unobserved heterogeneity in the sensitivity to surroundings simultaneously affect the choice of residential locations and the decision to move.

In transportation, commuting behavior is often treated a simultaneous decision with other choice dimensions (Bhat and Guo, 2007; Eluru, et al., 2009; Manaugh, et al. 2010; Pinjari, et al., 2011; Tu Tran, et al., 2016), whereas the unobserved factors have been examined using simultaneous equation models. For example, Plaut (2006) developed a seemingly unrelated regression model to estimate commuting distance and time simultaneously. Based on an Origin-Destination Survey in Montreal, Canada, Manaugh, et al. (2010) suggested that unobserved factors (life style, environmental concerns, perception of risk concerned with accidents on roads, etc.) are found to simultaneously influence both travel distance and home-job location choice. In addition, based on the data set from the San Francisco Bay Area, Pinjari, et al. (2011) presented an integrated simultaneous multi-dimensional choice model of residential location, car ownership, bicycle ownership, and commute tour mode choices. The significant magnitude of common unobserved factors was found across multiple choice dimensions. Thus, ignoring any of these effects could result in biased estimation of other effects. More recently, Tran, et al. (2016) proposed a joint equation modeling framework for the choice of residential location, job location, and commuting mode. It was found that the effects of common unobserved factors on residential location, job location and commuting mode choices are all existed in different magnitudes.

In addition, evidences in favor of life course approach proposed that, most life course mobility decisions are choices of lifestyles. Lifestyle refers to an individual's way of living and is influenced by his or her outlook of life and motivations, including opinions, beliefs, interests and attitudes (Van Acker, et al., 2010), which may influence the choices in different life domains simultaneously. Although researchers have

recognized the importance of lifestyles and attitudes in life course analysis, to the best to our knowledge, there are little research which incorporates the effects of such factors on the choice of various long-term decisions. To fill this research gap and being in line with the life course approach, multiple long-term mobility decisions are modeled jointly whereas the unobserved common factors are assumed simultaneously impact residential, job and car ownership change.

### 5.3 Data Analysis

Longitudinal data was collected to analyze the complex temporal interdependencies among various long-term mobility decisions and life events. In addition to describing their socio-demographic profile, respondents were asked to provide data pertaining to their longitudinal mobility decisions, including household events (marriage and child birth), housing and employment mobility decisions, and car ownership change. After checking the integrity and accuracy of the retrospective data, data of 414 respondents were used in this analysis.

Life events and mobility decisions		# of Cases	Percentage (%)
Marriage	Getting married before	266	64.3
	Never marriage	148	35.7
Child birth	Birthing child before	232	56.0
	Never have child	182	44.0
Move house	Never moved house	94	22.7
	Move house once	180	43.5
	Move house two times	97	23.2
	Move house three times	31	7.5
	Move house more than three times	13	3.1
Change job	Never changed job	166	40.1
	Change job once	140	33.8
	Change job two times	77	18.6
	Chang job three times	21	5.1
	Change job more than three times	10	2.4
Car ownership change	Never have cars	174	42.0
	Have cars but never change	163	39.4
	Chang cars once	60	14.5
	Change cars more than once	17	4.1

Table 5.1 Descriptive statistics of life events and different mobility decisions

The descriptive statistics of the life events and mobility decisions in different life domains are summarized in Table 5.1. In 2016, about 35.7% people were married, while 44% of the households had no child; 56% of the households had one or more child. As for residential mobility decisions, 22.7% of the sample never moved house, 43.5% moved once, 23.2% moved twice, while only 10% people moved more than two times. In terms of job change, almost 60% of the respondents experienced employment relocation, with 33.8% changing once, 18.6% changing twice, and 7.5% changing more than two times. Regarding car ownership change, more than 42.0% of the respondent never owned a car (174 respondents), nearly 39.2% owned one but never changed, 14.5% changed once, while only 4.1% changed at least once.

## 5.4 Conceptual considerations and model estimation

#### 5.4.1 Conceptual considerations

The purpose of this analysis is to uncover the interdependencies between different long-term mobility decisions. Three life choice dimensions in residence, job and car ownership are considered. In this study, residential change denotes renting/purchasing the first apartment/house and/or moving to a new place; Job change denotes getting the first job or changing to a new job; car ownership change denotes purchasing or replacing cars. Household structure change includes getting married and the birth of a child.

According to the existing literature, we assume the long-term decisions of residential and job mobility are directly influenced by household structure. Residential mobility decision is influenced by job mobility decision, and the reverse effect also exists. In terms of car ownership change, besides household structure change, both residential and job mobility choices are considered as important determinants. The interdependencies between various life events and mobility decisions in different life domains are shown in Figure 5.1. Second, we included lagged and lead effects in the model. Lastly, we assume that the effects of unobserved factors such as attitudes and lifestyles preferences may simultaneously influence different mobility decisions.



Figure 5.1 Interdependencies between various long-term mobility decisions

#### 5.4.2 Model specification and estimation

Here, we use a simultaneous equation model to capture three binary responses. Different from treating each mobility decision independently, the simultaneous equation systems allow the error terms to be correlated across equations. This makes it possible to formulate model specifications that account for complex observed and unobserved interrelationships that exist among multiple dependent variables. The general specification for a multivariate probit model is,

$$\mathbf{y}_{D_{t0}} = \mathbf{c}_D + \mathbf{\beta}_D \mathbf{x} + \mathbf{\gamma}_{hDt} \mathbf{w}_{ht} + \mathbf{\gamma}_{mDt} \mathbf{w}_{mt} + \mathbf{\varepsilon}_r \tag{1}$$

where  $y_{D_{t0}}$  denotes the binary response of various mobility decisions in the current year  $t_0$ , which  $y_{D_{t0}} = 1$  if a mobility decision D at time  $t_0$  is made, and 0 otherwise.  $c_D$  denote the constant of mobility decisions D at time  $t_0$ ; x is a vector of sociodemographic characteristics with the coefficient vector  $\beta_D$ ; while  $w_{ht}$  denotes household structure change, including marriage and child birth at time t, where t refers to the time earlier or latter then or same as the current year  $t_0$ ;  $w_{mt}$  denotes a mobility decision m at time t related to residence, job and car ownership change, which influence the targeted mobility decision D at time  $t_0$ . Similarly, as a dummy variable, both  $w_{ht} = 1$  and  $w_{mt} = 1$  if mobility decisions are made at time t, and 0 otherwise.  $\gamma_{hDt}$  and  $r_{mDt}$  are vectors of coefficients to be estimated.  $\varepsilon_D$  is a multivariate normally distributed error term, representing the unobserved components of mobility decision D. More specifically, three equations represent the mobility decisions. For each decision, temporal effects are considered. Thus, the simultaneous equation model can be written as follows,

$$y_{r\ t0} = c_r + \beta_r x + \gamma_{hrt} w_{ht} + \gamma_{rrt} w_{rt} + \gamma_{irt} w_{it} + \varepsilon_r$$
<sup>(2)</sup>

$$\mathbf{y}_{j_t t 0} = \mathbf{c}_j + \mathbf{\beta}_j \mathbf{x} + \mathbf{\gamma}_{hjt} \mathbf{w}_{ht} + \mathbf{\gamma}_{rjt} \mathbf{w}_{rt} + \mathbf{\gamma}_{jjt} \mathbf{w}_{jt} + \mathbf{\varepsilon}_j$$
(3)

$$\mathbf{y}_{c_{t}t0} = \mathbf{c}_{c} + \boldsymbol{\beta}_{c} \mathbf{x} + \boldsymbol{\gamma}_{hct} \mathbf{w}_{ht} + \boldsymbol{\gamma}_{rct} \mathbf{w}_{rt} + \boldsymbol{\gamma}_{jct} \mathbf{w}_{jt} + \boldsymbol{\gamma}_{cct} \mathbf{w}_{ct} + \boldsymbol{\varepsilon}_{c}$$
(4)

In the equations above,  $\mathbf{y}_{r_{c}t0}$ ,  $\mathbf{y}_{j_{c}t0}$  and  $\mathbf{y}_{c_{c}t0}$  denote the binary response of residence, job and car ownership change at time  $t_0$ , respectively.  $c_r$ ,  $c_j$ , and  $c_c$  are the constants.  $\mathbf{x}$  are vectors of social-demographic attributes with corresponding coefficient vectors  $\boldsymbol{\beta}_r$ ,  $\boldsymbol{\beta}_j$ , and  $\boldsymbol{\beta}_c$  for residence, job, and car ownership change, respectively. The timing influences on different mobility choice are also considered, including the concurrent, lagged and lead effects.  $\mathbf{w}_{ht}$ ,  $\mathbf{w}_{rt}$ ,  $\mathbf{w}_{jt}$  and  $\mathbf{w}_{ct}$  are vectors of time-dependent attributes. To be more specific, for residential and job mobility decisions at time  $t_0$ , household structure change ( $\mathbf{w}_{ht}$ ), residential mobility decisions ( $\mathbf{w}_{rt}$ ) and job mobility decisions ( $\mathbf{w}_{jt}$ ) at time t are assumed influencing moving house and changing job at time  $t_0$ ; In terms of car ownership change at time  $t_0$  are assumed to have an influence, but also car ownership change ( $\mathbf{w}_{ct}$ ) at different time are assumed trigging individuals or households change their car ownership at time  $t_0 \cdot \mathbf{y}_{trt}$ ,  $\mathbf{y}_{trt}$ ,  $\mathbf{y}_{irt}$ ,  $\mathbf{y}_{ht}$ ,

 $\gamma_{rjt}$ ,  $\gamma_{jjt}$ ,  $\gamma_{hct}$ ,  $\gamma_{rct}$ ,  $\gamma_{jct}$  and  $\gamma_{cct}$  are vectors of coefficients to be estimated.  $\varepsilon_r$ ,  $\varepsilon_j$ , and  $\varepsilon_c$  are error terms representing the unobserved components of moving house, changing job and car ownership. The unobserved components are all assumed to be normally distributed with mean 0 and variance-covariance matrix  $V \{\varepsilon_r, \varepsilon_j, \varepsilon_c\} \sim N(0, V)$ . The variance covariance matrix is given by:

$$V = \begin{pmatrix} 1 & \rho_{rj} & \rho_{rc} \\ \rho_{jr} & 1 & \rho_{jc} \\ \rho_{cr} & \rho_{cj} & 1 \end{pmatrix}$$

The off-diagonal elements in the covariate matrix represent the unobserved correlation between the stochastic components of three choice dimensions.

## 5.5 Results and discussion

Table 5.2 presents the estimated coefficients of the simultaneous equation model. The log likelihood at the optimal value is -1429.14, while the initial log likelihood is -1592.32. The McFadden's rho-squared values is 0.102. Results indicate that although effects are different for different time slices, various mobility decisions are shown to be influenced by social-demographic characteristics and different life events. Moreover, error correlations between all three life domains are statistically significant, which indicates that one choice dimension is not exogenous to the other life choice, but is endogenous to the system as a whole.

### 5.5.1 Personal characteristics

Turning to the socio-demographic characteristics, the results indicate that people at different life stages have different willingness to move house, change job and/or change cars. Specifically, the results indicate that young people change job more frequently than old people. This is understandable because young people may need to change a job to explore their career goals, while older people are more possible to settle down into a steady career and prefer not to change jobs frequently. In contrast, it appears that individuals/households are more likely to plan to move up the housing ladder with age. For many movers, residential relocation denotes an opportunity to improve ones' living environment (better educational institution, hospital, etc.). In terms of car ownership change, age is an important factor influencing the decisions of buying/changing cars in a household. Looking at different age groups, the results show that people aged at 25-40 are more prone to buy/change cars compared to other groups of people.

In addition, results show that compared with the high salary group, people with low salary are less inclined to change job. One reason is that people with a fair and stable salary are more likely to pursue a better one when they go through a job ladder. Looking at the influence of income on car ownership change, the results show that, compared with other groups, individuals with very high salary are more incline to buy/change cars.

#### 5.5.2 Household structure change

Household structure changes are found having different impacts on various mobility decisions. Based on the life course approach, rather than studying the instantaneous effects, both the lagged and lead effects are addressed in this research. In terms of the residential mobility decisions, although coefficients at all time instances are not statistically significant, the signs indicate both marriage and child birth may positively influence individuals'/households' moving house decision. For instance, household structure change may induce housing characteristics (i.e. numbers of bedrooms) no longer meet their needs, such a mismatch may induce families change their house to achieve a better fit. Additionally, the results show that marriage and child birth are shown having different magnitude of effects on residential mobility decisions. Specifically, it shows that child birth in general has a larger effect on residential mobility than marriage. Taking a closer look at different temporal effects, the concurrent effect of marriage on residential mobility decisions are shown to be higher than both the lagged and lead effects. However, a reverse effect was found between child birth and residential mobility decisions where the concurrent effect is shown much smaller than other temporal effects. This is perhaps because life course events like a child birth may cause much mental/monetary/time cost and consequently, families need time to adapt to buy/change a house. In addition, household structure change (job change and getting married) is found to negatively influence job mobility decisions which means people are less likely to change job in occurrence of household structure change.

Lastly, with respect to the effects of household structure change on car ownership change, a body of research found that marriage or child birth may positively influence household car purchase in response to the increased household size (e.g. Verhoeven, et al., 2005; Beige and Axhausen, 2012). However, in this study, household structure change does not show a statistically significant effect on car ownership mobility decision.

### 5.5.3 Mobility decisions in different life domains

With respect to the bi-directional relationships between residential and job change, in line with our hypothesis, it shows that a job change may lead to a change to residential relocation. Similarly, residential mobility is shown positively associated with job mobility. It confirms the interplay between job and residential mobility decisions, because mobility decision in one life domain can affect mobility decision in other life domain.
Chapter 5

Looking at all temporal effects, it is found that bi-direction relationships between residential and job change both have the bigger concurrent effects than the lagged/lead effects. In terms of the 1-year and 2-year lagged effects, residential relocation is found to be positively affected by job change. This finding indicates that job change of one or more household members may trigger the residential relocation in the next 1 or 2 years. In contrast, 1 and 2-year lead effects of job change are found the likelihood of moving house, which means negatively affecting that individuals/households are prone not to relocate when they have the expectation to change jobs in the near future. In case of the temporal effects of residential relocation on job mobility decisions, a residential change in the next 1 and/or 2 years may increase the likelihood of changing job.

Regarding car ownership change, we analyzed the temporal effects of residential and job change on car ownership change. It shows that car ownership decision is greatly affected by residential mobility than job mobility. This suggests that individuals/households may be more likely to consider changing their travel behavior in face of a residential relocation rather than changing a job. Furthermore, effects at different time vary in size. Both lagged and lead effects of residential mobility are shown larger than the immediate effect on car ownership change; conversely, job mobility is shown largely increasing the likelihood to buy or change cars in the same year. This may be due to that residential mobility and car ownership change are costly, thus households need time to balance their goals. However, a workplace change may cause the change of travel time which increases the needs of households to buy/change cars directly. Regarding the changing frequency in a certain period of time, as expected, the negative signs of the parameters within each life domain shows that people prefer not to make a transition within 2 years regarding moving house, changing job or car ownership.

	Move ho	Move house		Change job		Car ownership change	
Variable	Estimate	S.E.	Estimate	S.E.	Estimate	S.E.	
Age1(<25)	08	.08	.85***	.10	04	.10	
Age2 (25-40)	.04	.07	.04	.10	.15**	.07	
Age3 (41-55)	.01	.09	35***	.12	20**	.10	
Income (low)	.01	.07	01	.08	06	.08	
Income (medium1)	.08	.07	16*	.08	11	.08	
Income (medium2)	09	.09	.14	.10	09	.10	

#### Table 5.2 Estimation results

Marriage (t-2)	38	.28	41	.28	.10	.22
Marriage (t-1)	.37	.24	02	.24	12	.26
Marriage (t0)	1.04***	.21	36	.27	.03	.26
Marriage (t+1)	.35	.25	.03	.22	.28	.23
Marriage (t+2)	02	.25	57**	.28	-1.50	1.54
Child birth (t-2)	.54***	.16	.08	.19	26	.24
Child birth (t-1)	.61***	.19	17	.22	.13	.20
Child birth (t0)	.13	.23	36	.24	03	.23
Child birth (t+1)	.33	.21	50*	.26	.23	.23
Child birth (t+2)	.27	.22	43*	.26	.38*	.22
Move house (t-2)	73***	.23	29	.23	.25	.16
Move house (t-1)	99***	.28	.02	.17	.32**	.15
Move house (t0)			.71***	.18	.08	.22
Move house (t+1)	44***	.14	.37***	.12	.32***	.12
Move house (t+2)	12	.12	.16	.13	.31***	.12
Change job (t-2)	.11	.15	26***	.15	.04	.15
Change job (t-1)	.21	.15	58***	.17	06	.16
Change job (t0)	.80***	.21			.44**	.20
Change job (t+1)	10	.19	49**	.21	.13	.17
Change job (t+2)	34*	.19	.14	.16	24	.20
Car ownership change (t-2)					60**	.23
Car ownership change (t-1)					67***	.22
Car ownership change (t0)						
Car ownership change (t+1)					57***	.19
Car ownership change (t+2)					67***	.21
Constant	-1.52***	.08	<b>-</b> 1.60***	.10	-1.40***	.09
Error correlations						
Correlation error correlationsmo		26**	.02			
Correlation error correlationsmo		.13*	.08			
Correlation error correlationsJob		14**	.05			
$LL(\beta)$					-1429.14	
LL( <i>0</i> )					-1592.32	
Rho-square					.102	

\*\*\*, \*\*, \* means significance at 1%, 5% and 10% level.

# 5.5.4 Common error components among multiple long-term mobility decisions

In this particular study, all error correlations are found to be statistically significant, indicating that the validity of the assumption that various mobility decisions should be modeled jointly. The interpretation of the error correlations is that unobserved attributes that affect one dimension are correlated with the unobserved attributes that affect other dimensions. In this regard, one choice dimension is not exogenous to the other life choice, but is endogenous to the system as a whole.

In addition, correlation of unobserved variables between residential and job change is higher than other two bundles (change of car ownership and residence, change of job and car ownership). Additionally, it shows that residential change is positively correlated with unobserved factors that contribute to car ownership change, while correlations between job mobility and other two mobility decisions are both negative. The positive sign of the parameter (0.13) suggests that unobserved factors that motivate individuals/households to relocate are likely to contribute to purchase/change cars. Conversely, there are negative covariances (-0.26 and -0.14), reflecting a negative disposition across the job mobility decision and other mobility decisions.

# 5.6 Conclusions and discussion

By explicitly dealing with the endogeneity issue among various mobility decisions, this chapter analyzed the temporal dependencies between various life choices using retrospective survey data. In an attempt to consider the common error components in residential mobility, job mobility and car ownership change, a simultaneous equation model was applied.

Results of the analyses identified the household structure change and various life course mobility decisions may influence the long-term mobility decisions. It is found that household structure change turns out to be an important factor for long-term residential and job mobility decisions. More specifically, both marriage and child birth are significant determinants for moving house, whereas child birth has a larger effect on residential mobility than getting marriage. Conversely, household structure change was found to negatively influence individuals' changing job. With respect to the household structure change effect on car ownership change, the findings indicate that both marriage and child birth have small and insignificant influences on car ownership change. Additionally, two-way relationships between residential and job mobility are

found, with the concurrent effect larger than the lagged/lead effects. The findings also suggest that the correlated unobserved factors simultaneously affect the decision to move house/job and the decision of purchasing/changing cars, which clearly points to the need to associate various life course decisions in a simultaneous system.

# 6

# Temporal Dependencies in Mobility Decisions over the Life Course<sup>2</sup>

# 6.1 Introduction

In Chapter 5, a simultaneous equation model was used to examine the concurrent, lagged and lead effects between various life events and mobility decisions. A limitation of the approach is the time-invariant nature of the causal structure of life trajectory events. In light of this consideration, a dynamic Bayesian network (DBN) is introduced in this chapter. Under the formalism of a DBN, initially, a causal model for a single time instance is built. Then, a copy of this model is generated for each time instance belonging to the temporal range of interest. Finally, links between nodes in adjacent networks are introduced. In this way, a DBN obeys the Markov property, with links connecting either nodes within the same time slice or between adjacent networks, and the value of each variable represents the state of a real-world property at each time instance. Therefore, DBNs are more appropriate to model the interdependencies

<sup>&</sup>lt;sup>2</sup> This chapter is based on the article:

Guo J., Feng T., Timmermans H.T.P. (2019). Time-varying dependencies among mobility decisions and key life course events: An application of dynamic Bayesian decision networks, Transportation Research Part A, Vol 130, 82-92.

between various life domains in the real-world, since they explicitly represent the state of the system at each time instance.

From the beginning of the 1990s, there are numerous studies focusing on the choice of workplace and place of residence. Some studies have investigated how decisions about job and residential locations are mutually influential (Gordon, et al., 1991; Abraham and Hunt, 1999). Although the interdependency between residential and job mobility decisions was examined in the previous chapter, the interaction between job location and residential location was not included in the analysis. Therefore, this chapter will explore whether people have the intention to move home closer to their workplace and vice versa.

Thus, the aim of this chapter is two-fold. First, it looks into the interdependencies between various long-term events: household events, residential choice, job choice and car ownership change. Second, as an extension of the time-invariant models, this chapter examines the complex temporal dependencies by developing a dynamic Bayesian network. The network creates an instance of each random variable for each point in time. Thus, concurrent, lagged and lead effects between various life course domains are examined in this research.

The remainder of this chapter is organized as follows: The next section presents the integrated framework. Following that, the Dynamic Bayesian network approach is presented. A brief description of the survey data is given in Section 6.3. Next, estimation results are discussed in Section 6.4. Conclusions and suggestions for future research are presented in the last section.



Figure 6.1 Consideration dependencies

# 6.2 Dynamic Bayesian networks

# 6.2.1 Conceptual considerations

The aim of this chapter is to uncover the interdependencies among long-term mobility decisions under the influence of various life course events. Based on the literature and available data, the life course events considered include change in household structure, residential change, job change and car change. Household structure change includes getting married and the birth of a child; residential mobility includes renting/purchasing the first apartment or house and moving house; changing job includes getting the first job and change of job; car change includes purchasing and/or replacing a car, regardless of buying the first/additional cars or going from one car to a different car. Besides, the influence of commuting time and socio-demographic characteristics on behavioral change is considered.

Decisions made during the current year are assumed to be influenced by past events, events in the same year and future events. Thus, the proposed Bayesian network examines concurrent, lagged and lead effects among various time-dependent events as illustrated in Fig. 6.1, where only the connections between different time instances are presented: the current (t0), previous (t-1) and future (t+1) situations. The rounded dot dashed lines present the one-year and two-year effects of future household structure change on current mobility decisions (lead/anticipation effects). Black bold lines represent the one-year effects of previous mobility decisions on current mobility decisions (lagged effects). Similarly, two-year lagged effects between household structure change and mobility decisions are shown by long dashed lines.

# 6.2.2 Model specification

Bayesian networks are attractive to flexibly examine the interdependency relationships between various life course domains. BNs belong to the family of probabilistic graphical models. They consist of a set of nodes and a set of arcs that form a directed acyclic graph (DAG). Each node represents a domain variable, whereas directed arrows between variables indicate dependence between them given that the values of their parents are known. Let  $\mathbf{X} = \{X_1, X_2, ..., X_n\}$  denote a set of random variables, while **Pa** ( $X_i$ ) denote the parent node of  $X_i$  in the DAG. The conditional probability distribution of  $X_i$  is denoted by  $P(X_i | Pa(X_i))$ . Then, the joint probability distribution can be represented as follows

$$P(X_1, X_2, ..., X_n) = \prod_{i=1}^{n} P(X_i | \mathbf{Pa}(X_i))$$
(1)

Any probability of interest can be computed from this joint probability distribution.

While Bayesian networks are powerful models for examining the interdependencies between various household long-term decisions, the Bayesian Networks were originally not designed to explicitly model temporal relationships; they are static models. As a temporal extension of a Bayesian network, dynamic Bayesian network has been developed to introduce a temporal dimension to BNs. Therefore, this chapter represents an extension of previous work on life trajectory decisions using static Bayesian networks (e.g. Verhoeven, et al., 2005; Wang, et al., 2018). It offers an appropriate approach to explore the dynamic dependencies between various life course domains into an integrated model, in which the state of a variable at one time instance depends on one or more states at other time instances.

DBNs can be defined as a pair of BNs  $(B, B_{\rightarrow})$ , where *B* represents the initial distribution  $P(X_t)$  and  $B_{\rightarrow}$  is a two-slice temporal Bayes net, which contains an instance of each variable at time m+m' and m. The probability of node *i* over two time slices is defined as,

$$P(X_{m+m'}^{i}|X_{m}^{i}) = \prod_{t=m}^{m+m'} P(X_{t}^{i}|\boldsymbol{P}\boldsymbol{a}(X_{t}^{i}))$$

$$\tag{2}$$

Where  $X_t^i$  is the *i*th node at time instance *t*, and *Pa* ( $X_t^i$ ) is the set of parent nodes  $X_t^i$ ,  $t \in [m', m]$ .

#### 6.2.3 Learning

Similar to a BN, a useful property of DBNs is their ability to learn from observations. Learning a DBN involves two steps: finding the network structure (structure learning), and finding the parameters that best describe the data, given a network structure (parameter learning).

#### 6.2.3.1 Structure learning

Structure learning aims to figure out a proper DAG, and confirm the association relationship between nodes. DBNs consider not only the dependencies between variables at one time instance but also the dependencies existing between several time instances. The dependent and causal directions between time nodes can be determined based on the mutual information between a pair of nodes. Structure learning of DBNs remains an active research field with several algorithms being proposed (Murphy,

2003). Even though the MCMC (Markov chain Monte Carlo) and Structural EM algorithms (Expectation Maximization) can be applied in dynamic Bayesian network structure learning, their applicability in the current study is limited due to too many nodes and number of time slices. In this regard, based on *a priori* knowledge, some constraints were enforced to reduce the degrees of freedom in learning the structure of the network. That is, semi-structure learning is adopted in the current study, where the structure within the same time slice is learned from the data, while the temporal dependency structure between different time instances is constructed based on a priori knowledge.

Considering the complexity of the structural learning process in DBNs, especially in treating the constrained casual relationships between different time instances, the network structure was learned automatically from the data only for the base network. The temporal dependency was determined in accordance with the conceptual framework. More specifically, a Bayesian search algorithm was used for structure learning for a single time instance. For a more detailed explanation on the search algorithm, readers are referred to Cooper and Herkovits (1992) and Heckerman (1995). In addition, some constraints were set to simplify structure learning and deny some unreasonable causal relationships learned from the data in the first instance of time. Setting these constraints will not affect the validity of the analysis.

- Age, as a socio-demographic characteristic, is closely related to longer-term decisions. For example, older people are less likely to move house than younger people. Similarly, family structure change, residential and job mobility, as well as car ownership change might be all influenced by the age of individuals. On the other hand, it is impossible that any life course events influence age. Thus, such effects are unidirectional between state/events and age.
- 2) Family structure change (getting married and child birth) may affect change of residence, job, and car ownership, while the opposite process is less likely to happen. Therefore, the effects of mobility decisions on family structure change are ignored in this study.

As discussed above, the network structure can be extracted from empirical data or simply derived from domain knowledge. Due to limited computational power, the dependencies between life course events across different time instances are based on the literature. First, family structure change is assumed to have both forward and backward effects on mobility decisions. Second, residential and job mobility decisions are assumed to have concurrent/lagged/lead effects on car ownership change. Third, in

terms of the residential/job location change, the current location choice is assumed to be a major factor influencing future location choice. Moreover, considering that residences and workers might relocate or purchase a car to avoid excess commuting time, travel time is assumed to affect various mobility decisions.

# 6.2.3.2 Parameter learning

With the given network structure, parameter learning entails finding the optimal parameters for a given network structure and determining the conditional probability distribution of each observation variable. Parameter learning in DBN is similar to learning in BN, except that parameters must be tied across time-slices. The expectation Maximization (EM) algorithm is capable of learning parameters from datasets that contain missing values. Thus, the EM algorithm was adopted in this study.

Because longitudinal data cover a long period of time, some life course events data are missing. The EM algorithm solves this problem by iterating between an Expectation step (E-step) and a Maximization step (M-step). Each E-step estimates expectations over the hidden variables. Next, parameters are updated in the M-step using the expectations of the hidden variables obtained in the E-step. The updated model is then used in the next E-step to obtain more accurate estimates of the hidden variables. In this way, estimated parameters are improved in every iteration and eventually converge to an optimum.

# 6.3 Data

In order to test the concurrent, lagged and lead effects between different mobility decisions and life events, data which were collected through the retrospective survey were used. This chapter uses data from four dimensions, family structure biography, residential history, job history, and car ownership biography. The sample size adopted in this chapter is the same as in Chapter 5. Data of 414 respondents are used in this analysis.

# 6.4 Results and analysis

## 6.4.1 Results of structure learning

The Bayesian search algorithm was used in the structure learning process within one time slice. The prior link probability was set at the default value of 0.001 for the structure learning process. The learned causal dependencies between the variables

within one time-slice are shown in Figure 6.2. Household mobility decisions include two life course events within households: getting married (marry) and child birth (birth); life course mobility decisions include three life domains: residential mobility (changeH), job mobility (changeW), and car ownership change (changeC). Residential location and workplace are represented as locationH and locationW, respectively. As expected, family structure change is closely associated with other life course mobility decisions. The results clearly present a direct concurrent relationship between car ownership change and residential and job change, which is in line with previous studies. Moreover, life course mobility decisions have concurrent effects on state variables, i.e. travel time (tt). Lastly, the socio-demographic variable 'age' is directly linked with various life course mobility decisions.

The causal relationships with temporal dependencies are presented in Figure 6.3. Orange numbers in rectangles represent one and two years lagged effects between various mobility decisions, while green numbers represent previous location choice influencing people's future location choice. In the unrolling structure of the DBN model, the same network is copied to each time instance. Here, the lead effects between life course events and various mobility decisions are added manually to the unrolling structure network. In total, five time slices are constructed based on five years of continuous data.



Fig 6.2 Structure of concurrent effects



Fig 6.3 Structure of lagged effects (left) and lead effects (right)

#### 6.4.2 Results of parameter learning

Based on the network structure presented above, the EM algorithm was used to estimate the conditional probability tables (CPTs) for each time node based on the retrospective data. Bayesian networks can portray the direct and indirect dynamic effects between the nodes of interest based on provided evidence, and compare the change in probabilities with and without an event. The evidence means that the probability of a state of a certain node is known or instantiated. When providing evidence for certain nodes, the probability of the states of others nodes may be updated depending on their structural connection. To analyze the positive or negative causal effects between different mobility decisions and states given evidence for other nodes, both the updated probability and the relative probability differences are calculated. The relative probability differences are defined as the difference between the updated and prior probability divided by the prior value. Assume the prior probability of a certain state of a variable x is  $p_0$ , the updated probability is p', the relative probability difference as follows,

$$p_{diff} = \frac{p_0 - p'}{p_0}$$
(3)

The relative probability difference can be negative or positive, which means that the probability of the predicted event decreases or increases given the evidence of certain variables/events. A value close to 0 means that the predicted event is not influenced by the given variables/events. The conditional probabilities in DBN reflect the effects of state variables between time slices. The relevant conditional probability tables, which describe the temporal interdependencies between different life domains, are shown in the following sections.

#### 6.4.2.1 Mobility decisions change for different age categories

Taking age as the parent node, both the updated probability and the relative probability differences of various mobility changes (shown in parentheses) are presented in Table 6.1. It is found that age has a direct impact on residential and job mobility decisions as well as car ownership change. As a socio-demographic variable, 'age' was categorized into four categories: under 25, 25-34, 35-50, and over 50. For the category age under 25, the probability of change is positive for all mobility decisions, indicating that young people are more likely to change their current residence, job and car ownership. The biggest relative probability difference among these mobility decisions is for the job domain, which suggests that the probability of changing job is bigger for younger generations. In contrast, people aged 25-34 and 35-50 are more likely to have a stable life and are less likely to change their current status. Our results indicate that people aged over 50 also change residence, job and car. The reason may be that, compared to those aged 25-50, people aged over 50 may have more money and time, thus having the capability and willingness to choose their favorable house/job or to buy a new car. However, the size of change is much smaller than that of young people, which suggests that young people (aged under 25) are instable in both the labor and housing market, and change more frequently than other age groups. This result is consistent with the findings of other studies, including Habib, et al. (2011), which also found that younger people are more likely to change job than older people.

#### 6.4.2.2 Mobility decisions change given different levels of family structure change

Given family structure change as the parent node, the updated probability and the relative probability differences of various mobility are presented in Table 6.2. Here, '*t*' indicates the year when a certain event occurred; '*t*-1, *t*-2' indicate one-year and two-years ahead of the event, and '*t*+1, *t*+2' indicate one-year and two-year later than the event. It shows that various life course mobility decisions are strongly influenced by family structure change. This finding emphasizes the importance of family structure change as a trigger of people's relocation and car ownership change.

Age	Residential mobility	Job mobility	Car ownership change		
No evidence	8%	9%	13%		
<25	19% (141.7%)	38% (322.2%)	33% (153.3%)		
26-34	7% (-9.1%)	11% (22.2%)	12% (-6.7%)		
35-50	7% (-9.1%)	7% (-22.2%)	12% (-6.7%)		
>50	10% (25.0 %)	11% (22.2%)	19% (46.7%)		

**Table 6.1** Conditional probability distribution of various mobility decisions for different age categories

Table 6.2 Transition probability of various mobility change given evidence of life events

Family structure change		Residential mobility (t	Job mobility ( <i>t</i> )	Car ownership change ( <i>t</i> )
No change		8%	9%	13%
Getting married	<i>t</i> -2	17% (112.5%)	14% (55.6%)	37% (184.6%)
	<i>t</i> -1	18% (125.0%)	24% (166.7%)	36% (176.9%)
	t	41% (412.5%)	21% (133.3%)	37% (184.6%)
	<i>t</i> +1	19% (125.0%)	23% (155.6%)	30% (130.8%)
	<i>t</i> +2	17% (112.5%)	21% (133.3%)	31% (138.5%)
	<i>t</i> -2	20% (150.0%)	18% (100.0%)	29% (123.1%)
	<i>t</i> -1	23% (187.5%)	17% (88.9%)	28% (115.4%)
Child birth	t	16% (100.0%)	15% (66.7%)	31% (138.5%)
	<i>t</i> +1	24% (200.0%)	17% (88.9%)	32% (146.2%)
	<i>t</i> +2	18% (125.0%)	17% (88.9%)	30% (130.8%)

The conditional probability table shows that both marriage and child birth have a positive effect on residential mobility decisions, which is consistent with existing findings. However, effects differ in size at different time instances. In general, child birth has larger effects on residential mobility than getting married. Obviously, the birth of a child implies new responsibilities. As a consequence, household may be more likely to find a larger house in response to the increase in household size. The biggest probability change occurs for getting married at time t (dramatic increase from 8% to 41%), which indicates that the concurrent effect of getting married is larger than both the lagged and lead effects. However, a reverse effect was found between child birth and residential mobility, with lagged and lead effects both being larger than the concurrent effects. This means people are more likely to move house in the year of marriage, but less likely when a child is born. This is understandable considering that

arrangements for having a child and moving house in the same year may not be easy. Additionally, the comparison between probabilities change across different time slices shows that one-year temporal effects are in general larger than two-year temporal effects for both lagged and lead effects.

In terms of job mobility, concurrent, lead, and one-year lagged effects (*t*-1) of a child are found in general smaller than the effects of getting married. In addition, the comparison between different time instances shows that both lagged and lead effects of having a child (17%) are larger than the effect at time *t* (15%). However, the same effects are not found for getting married.

Table 6.3	Transition	probability	of car	<sup>r</sup> ownership	change	given	evidence	of residen	tial
and work c	hange								

Mobility decision	Car ownership change (t)	
No change		13%
	<i>t</i> -2	29% (123.1%)
Residential mobility	<i>t</i> -1	29% (123.1%)
	t	31% (138.5%)
	<i>t</i> +1	19% (46.2%)
	<i>t</i> +2	17% (30.8%)
	<i>t</i> -2	27% (107.7%)
Table and billing	<i>t</i> -1	30% (130.8%)
JOD MODILITY	t	28% (115.4%)
	<i>t</i> +1	19% (46.2%)
	<i>t</i> +2	16% (23.1%)

**Table 6.4** Transition probability of residential/work location at time t given location at time t-1 and mobility decisions

		Residential location (t)	Job location ( <i>t</i> )
Mobility decisions		Change home	Change job
No evidence		62%	69%
Residential location (t-1)	Job location ( <i>t</i> -1)	(central city)	(central city)
Central city	central city	73% (17.7%)	88% (27.5%)
Central city	surrounding area of the city	68% (9.7%)	75% (8.7%)
Surrounding area of the city	central city	60% (-3.2%)	47% (-31.9%)
Surrounding area of the city	surrounding area of the city	21% (-66.1%)	22% (-68.1%)

Mobility decisions		Residential mobility(t)		Job	mobility ( <i>t</i> )	Car ownership change (t)		
Commuting time ( <i>t</i> -1)	No evidence	8%			9%		13%	
	<20 min	8%	(0.00%)	8%	(-11.10%)	12%	(-6.67%)	
	20-40 min	7%	(-16.67%)	10%	(11.10%)	12%	(-6.67%)	
	41-60 min	9%	(8.33%)	10%	(11.10%)	14%	(6.67%)	
	>60 min	9%	(16.67%)	10%	(11.10%)	14%	(6.67%)	

**Table 6.5** Transition probability of commuting time at time *t* given commuting time at time *t*-1 and various mobility decisions

#### 6.4.2.3 Mobility decisions change given different levels of commuting time

Considering the role of commuting time in different mobility decisions, Table 6.5 presents the probability change of various mobility decisions at time t given evidence of commuting time at time t-1. Commuting time is categorized into four levels: less than 20 minutes, 20-40 minutes, 41-60 minutes, and longer than one hour. Results show that commuting time is a factor influencing people to change residence, job and car ownership. In general, the longer the commuting time, the higher the probability people move house, change job or car. More specifically, the probability of moving house or changing car increases when commuting time is longer than 40 minutes. For changing job, the probability increases the probability of changing job. This conclusion is in line with the previous findings by Habib, et al., (2011). However, it should be noted that, compared with the results of various life course mobility decisions, effects of commuting time are relatively small. This probably indicates that these long-term decisions are more dependent on anticipated life course events than on travel-related attributes like commuting time.

# 6.5 Conclusions and discussion

Research on long-term mobility decisions has received increasing interest lately. People hold a series of aspirations related to different life domains. In an attempt to better understand the decision-making process, this chapter proposed an integrated framework to comprehensively explore the interdependencies between various life domains. In addition, because people have different needs at different life stages, interdependencies between life events and mobility decisions may change over time. To incorporate the time dimension into life course analysis, a dynamic Bayesian network was estimated.

The model results evidence the existence of concurrent and temporal interdependencies between different mobility decisions. Both getting married and child birth are found to increase the probability of residential/job/car ownership change. Likewise, both residence and job relocation are found to have positive effects on car ownership change, indicating that either moving house or changing job may increase people' needs to purchase/replace a car.

In terms of the relevance of long-term policy interventions, people's mobility decisions over the life course cannot be ignored. Results also show that temporal one-year effects are larger than two-year effects for both lagged and lead effects.

Despite the relevance of the approach, a limitation of the current analysis is its focus on the individual as the decision-making unit. However, in reality, it is more likely that life trajectory decisions of households, who physically share resources such as house, income and cars (Roorda, et al., 2009; Timmermans and Zhang, 2009), are made at the household level. To contribute to the further development of the relatively thin line of research on household decision-making, temporal interdependencies between various life course mobility decisions for household members are examined in the next chapter.

# 7

# Temporal Dependencies in Mobility Decisions: A Household-level Analysis<sup>3</sup>

# 7.1 Introduction

In the previous two chapters, a life course approach has been introduced to examine the interdependencies between key life events, long-term mobility decisions such as residential move, job change and change in vehicle possession. However, to deepen the understanding of the temporal interdependencies between different life domains, further analyses are necessary. The previous two chapters were conducted based on individuals, without considering the possible influence of other household members. However, as members in a household physically share various household resources, some of these long-term decisions affect them equally, while job change affects them differently because their job location likely differs. Thus, as a natural extension to the previous chapter, this chapter contributes to the literature by modelling long-term

<sup>&</sup>lt;sup>3</sup> This chapter is based on the article:

Guo J., Feng T., Zhang, J., Timmermans H.T.P. (2020). Temporal interdependencies in mobility decisions over the life course: a household-level analysis using dynamic Bayesian networks, JournalJournal of Transport Geography, Vol 82.

mobility of dual-earner households, looking into the temporal dependencies between various life domains: child birth, residential change, job change, and car ownership change. Moreover, gender analyses will be conducted in this chapter. We analyze to what extent child birth affects life trajectories in other domains of wives and husbands, and how the status of wives and husbands affects household decisions.

The remainder of this paper is organized as follows. Section 7.2 presents the structure of the integrated model that explicitly incorporates time-dependent dependencies within and between different life domains, using households as the observed decision-making unit. Section 7.3 discusses the retrospective survey that was used to collect the life trajectory data of households. The model results are presented and interpreted in Section 7.4. Finally, concluding remarks are made regarding results, policy implications and future research.

# 7.2 Methodology

#### 7.2.1 Background

Long-term decisions such as residential and job choice are high involvement decisions that have long-term repercussions on people's daily life. These decisions tend to codepend on past and future decisions across different life domains. The aim of this chapter is to uncover these dynamic interdependencies between household long-term mobility decisions in various life domains. In particular, we focus on residential mobility (moving house), job mobility (changing job) and car ownership mobility (change in car possession). We use one year as the time unit. If any event occurs more than once in a single year, we take the last event into account. For example, if individuals reported they changed job three times within one year, we take the state of the last event as input to the Bayesian decision network.



Figure 7.1 Network structure

## 7.2.2 Network structure

The network structure is shown in Figure 7.1. As discussed, the framework includes dynamics, time dependence, and interrelationships between long-term decisions. With respect to temporal interdependencies between different long-term mobility decisions, it is assumed that child birth may have both 1 and 2 years lagged and lead effects on the long-term mobility decisions. Thus, for example, mobility events such as residential and job relocation may have lagged, concurrent and lead effects on car ownership change. Lastly, mobility states such as income and commuting time are assumed to have direct impacts on various long-term mobility decisions. For example, people may take different actions such as changing job, moving house, and purchasing/changing cars to avoid excessive commuting time.

# 7.2.3 Learning

Given a dataset, developing a DBN requires two steps: learning the network structure and estimating the parameters of the learned network structure. A semi-structure learning process was adopted in this study. A more detailed explanation of the learning process is discussed in section 6.2.3.

Moreover, to simplify the network structure and avoid unrealistic relationships, plausible constraints were set. In particular, child birth may affect residential change, job change for both wives and husbands, and car ownership change. However, the reversed effects are not allowed. Thus, unidirectional effects were set between child birth and various long-term decisions.

# 7.3 Data analysis

## 7.3.1 Data collection

In order to analyze the temporal effects between various life course household mobility decisions, retrospective life course data were used in this research. Except for sociodemographic information of each household member, data about life course events were collected. In particular, the following events were included in the data collection: child birth, residential moves, car ownership change and job change of each spouse. In addition, because these were assumed to be important influential variables, annual income/and commuting time of both spouses were collected. A single respondent provided the data of both spouses. This may be less ideal, but arranging interviews with two spouses would have been much more demanding and not worthwhile. In order to reduce respondent burden, respondents were asked to provide information about the life events for only the last five times it occurred. In many cases, this maximum of five still covers the full trajectory of life events in a particular domain, particularly for younger respondents.

Considering the aim of the study, we restricted our sample to respondents belonging to dual earner households. Ultimately, 266 respondents were used in the analyses. The response rate is 16%, which is satisfactory, considering the unannounced contacting of possible respondents, the high percentage of non-eligible respondents.

## 7.3.2 Sample description

As shown in Table 7.1, the sample consists of 138 (51.9%) females and 128 (48.1%) males. The average age is 38.2, while on average are younger than their husbands. Average household size is 3.2 persons. 12.8% of the couples have no children; 5.6% have two children, while the remaining 81.6% has one child. These statistics indicate that some households involved three generations.

Variable	Classification	# of Cases	Percentage (%)
Gender	Female	138	51.9
	Male	128	48.1
Age	<35	97	36.5
(Wife)	35-50	128	48.1
	>50	41	15.4
(Husband)	<35	66	24.8
	35-50	152	57.1
	>50	48	18.0
Marital Status	Couple with no child	34	12.8
	Couple with one child	217	81.6
	Couple with two children	15	5.6

#### Table 7.1 Sample description

Table 7.2 presents the descriptive statistics for the main life course events. 22.9% of the respondents never changed residence; 41.4% changed only once, 25.2% changed twice and only about 10.5% moved house more than 2 times. In case of car ownership change, 33.5% of the respondents never had a car in their household; 43.2% had one car but never changed; 14.4% reported to have changed cars once, while the remaining 5.7% changed cars more than once in the past. Moreover, data of job change were collected for both spouses. As shown in Table 7.1, wives changed job less often than husbands.

Variable	Classification	# of Cases	Percentage (%)
Residential change	Never moved house	61	22.9
	Moved house once	110	41.4
	Moved house twice	67	25.2
	Moved house three times	19	7.1
	Moved house more than three times	9	3.4
Job change	Never moved job	78	29.3
(Wife)	Moved job once	156	46.2
	Moved job twice	46	17.3
	Moved job three times	14	5.3
	Moved job more than three times	5	1.9
(Husband)	Never moved job	54	20.3
	Moved job once	130	48.9
	Moved job twice	58	21.8
	Moved job three times	18	6.8
	Moved job more than three times	6	2.3
Car ownership change	Never had any cars	89	33.5
	Always had the same vehicle	115	43.2
	Change cars once	47	14.4
	Change cars twice	14	5.3
	Change cars more than two times	1	0.4

#### Table 7.2 Descriptive statistics of the life course events

# 7.4 Results and analysis

# 7.4.1 Results of structure learning

The Bayesian search algorithm was used in the structure learning process within one time slice. The prior link probability was set as the default value 0.001 for the structure learning process. The learned structure network for the one-time slice is shown in Figure 7.2 (left). Concurrent effects between various household mobility decisions are shown as two aspects. First, as we expected, child birth has direct effects on various household mobility decisions such as residential change, car ownership change, and job change for both husband and wife. Second, the learned structure network indicates that moving house and changing job for both wife and husband have direct effects on household car ownership change. Lagged and lead causal relationship between various household mobility decisions discussed in section 7.3 are represented in Figure 7.2 (right) and 7.3 respectively. 1-year temporal effects are shown by the blue lines, while 2-year temporal effects are shown by red lines.



# Figure 7.2 Learned network structure of concurrent effects (left) and lagged effects (right)



Figure 7.3 Network structure of lead effects

# 7.4.2 Results of parameter learning

Based on the network structure shown in Figure 7.1, the EM algorithm was applied to estimate the time-dependent conditional probability tables (CPTs) for all nodes and time slices. In this paper, both the updated conditional probability and relative probability differences are calculated.

## 7.4.2.1 Household mobility decisions given child birth

The effects of child birth on different household mobility decisions are presented in Tables 7.3 and 7.4. Tables 7.3 and 7.4 show that child birth positively influences different mobility decisions in households. However, the effects differ in size for different decisions. Looking at the conditional probability table, the effects on residential change and car ownership change are much stronger than the effect on job change for both wife and husband in dual-worker households. The birth of a child implies new responsibilities. Consequently, household are more likely to find a larger house/change car in response to an increase in household size.

Mobility decisions	Residential mobility	Car ownership change		
No child birth	9%	13%		
<i>t</i> -2	19% (111%)	24% (85%)		
<i>t</i> -1	20% (122%)	30% (131%)		
t	17% (89%)	32% (146%)		
<i>t</i> +1	27% (200%)	30% (131%)		
<i>t</i> +2	24% (167%)	28% (115%)		

**Table 7.3** Conditional probability and probability difference (in parentheses) of residential mobility and car ownership change given different levels of child birth

Table	7.4	Conditional	probability	and	probability	difference	(in	parentheses)	of	job
mobility	/ give	en different l	evels of chil	ld bir	th					

Mobility decisions	Job mobility (Wife)	Work mobility (Husband)
No child birth	10%	9%
<i>t</i> -2	18% (80%)	13% (44%)
<i>t</i> -1	12% (20%)	12% (33%)
t	13% (30%)	12% (33%)
<i>t</i> +1	12% (20%)	12% (33%)
<i>t</i> +2	14% (40%)	11% (22%)

In addition, the temporal effects of child birth are shown to differ in size between the long-term mobility decisions in different life domains. Specifically, results show that both the 1-year lagged and lead effect are much bigger than the 2-years temporal effects. In addition, the lead effects (t+1/t+2) are in general larger than the lagged effects (t-1/t-2), suggesting that residential change/car ownership change is more likely to occur in anticipation of a household structure change in the current study.

However, a reverse result is found for car ownership. A child birth increases the probability of a change in car ownership mostly in the same year. The two-year lag effect is much smaller than the 2-year lead effect, while the one-year lead and lag effects are the same. It suggests that child birth tends to immediately increase people's mobility needs and hence their tendency of purchasing an (additional) car.

In case of changing job, results show that both the lagged and lead effect as well as the concurrent effect of child birth on changing job for both wife and husband are much smaller than the effects on other life course mobility decisions. It means that childbirth tends to primarily increase the need for a larger dwelling and/or a more convenient transportation option. Although the set of responsibilities of both wife and husband may increase due to a new-born baby, there is no strong evidence that household members will change their job. Moreover, a noticeable finding is that both the 2-year lagged and lead effects for the wife are bigger than the 1-year and concurrent effects, indicating that wives need more time to change job before/after giving birth to a child as a similar result is not found for husbands.

#### 7.4.2.2 Car ownership change given residential and job change

Tables 7.5 and 7.6 show the results of the relative probability differences given evidence of various household mobility decisions. The lagged, lead and concurrent effects of residential and job change for both wife and husband are found to positively influence car ownership change. In general, concurrent effects are stronger than lagged and lead effects on car ownership change.

Taking a closer look at the different temporal effects on various mobility decisions, results show that the conditional probability of car ownership change is nearly double the prior probability. Moreover, the updated CPTs show that concurrent and lagged effects are slightly larger than the lead effects. Only minor differences are found between wife and husband. However, when both wife and husband change their job, the probability for households buying or changing car increases dramatically.

Mobility decision	Car ownership change	
No change	13%	
<i>t</i> -2	28% (115%)	
<i>t</i> -1	28% (115%)	
t	30% (131%)	
<i>t</i> +1	25% (92%)	
<i>t</i> +2	25% (92%)	

**Table 7.5** Conditional probability and probability difference (in parentheses) of car ownership change given different levels of residential mobility

Table	7.6	Conditional	probability	and	probability	difference	(in	parentheses)	of	car
owners	hip c	hange given	different le	vels (	of job mobil	ity				

Mobility decision	Car ownership change			
No change	13%			
	Wife	Husband	Both wife and husband	
<i>t</i> -2	20% (54%)	21% (62%)	35% (169%)	
<i>t</i> -1	21% (62%)	22% (69%)	39% (200%)	
t	23% (77%)	23% (77%)	38% (192%)	
<i>t</i> +1	21% (62%)	22% (70%)	36% (177%)	
<i>t</i> +2	21% (62%)	21% (62%)	34% (162%)	

## 7.4.2.3 Household mobility decisions given wife and husband's commuting time

Our conceptual model assumed that commuting times of household members may influence long-term residential change, job change and car ownership change. Several interesting findings are found in this analysis (shown in Tables 7.7, 7.8 and 7.9). First, as expected, wife and husband's commuting times differently influence household residential change and car ownership change. In case husbands have excessive commuting times (more than 1 hour) while wives have a relative long commuting time (40-60 minutes), residential change and car ownership change increase to 14% and 18%, respectively. However, when the wife has more than 1 hour travel time, while the husband has 40-60 minutes commuting time, the probability of change is small (only 5% and 12%, respectively). This means that the probability of residential change and car ownership change increase when wife and husband both face relative long commuting times. However, the magnitude of this effect varies between household members.

Mobility decision		Residential mobility
No change		9%
Wife	Husband	
< 20 min	< 20 min	11% (26%)
	20-40 min	9% (0%)
	41-60 min	11% (21%)
	>60 min	14% (53%)
20-40 min	< 20 min	8% (-11%)
	20-40 min	6% (-32%)
	41-60 min	7% (-21%)
	>60 min	10% (5%)
41-60 min	< 20 min	10% (16%)
	20-40 min	10% (11%)
	41-60 min	12% (32%)
	>60 min	14% (58%)
>60 min	< 20 min	14% (58%)
	20-40 min	10% (5%)
	41-60 min	10% (5%)
	>60 min	13% (42%)

**Table 7.7** Conditional probability and relative probability differences (in parentheses) of residential mobility at time t given commuting time at time t-1 of the spouses

Second, in dual-earner household, the probability of moving house dramatically increases if the residential location is very close to one worker but quite far away from the other worker's job location. For example, as shown in Table 7.8, in case the commuting time is less than 20 minutes for the wife and more than one hour for the husband respectively, the probability of moving house is increased to 14%. Similar results can be found for car ownership mobility change. Lastly, if at least one household member has a relatively short commuting time (20-40 minutes), the probability to move house and change cars are relatively low.

Given different levels of commuting time for wife and husband, results show that excessive commuting times do increase the probability of job change. However, this effect differs between wives and husbands. As shown in Table 7.9, wives are more sensitive to long commuting times (more than 40 minutes) than husbands, and consequently more likely to change job. In this regard, wives and husbands may take different actions when faced with excessive commuting times. In case only husbands face an excessive commuting time, dual-worker households have a larger probability to move to a new house or switch to a more convenient transportation mode (53% and 35% increase of probability, respectively). However, if only the wife is facing an excessive commuting time, they are more likely to change job instead of moving house or changing car ownership.

Mobility decision		Car ownership change
No change		13%
Wife	Husband	
< 20 min	< 20 min	15% (12%)
	20-40 min	14% (8%)
	41-60 min	16% (20%)
	>60 min	18% (35%)
20-40 min	< 20 min	11% (-20%)
	20-40 min	10% (-27%)
	41-60 min	11% (-15%)
	>60 min	14% (4%)
41-60 min	< 20 min	15% (15%)
	20-40 min	14% (8%)
	41-60 min	15% (15%)
	>60 min	18% (35%)
>60 min	< 20 min	16% (23%)
	20-40 min	13% (0%)
	41-60 min	15% (12%)
	>60 min	18% (35%)

**Table 7.8** Conditional probability and relative probability differences (in parentheses) of car ownership change at time t given different levels commuting time at time t-1 of the spouses

**Table 7.9** Conditional probability and relative probability differences (in parentheses) of job mobility at time t given different levels commuting time at time t-1 of the spouses

Mobility decision	Job mobility (Wife)	Job mobility (Husband)
No change	10%	
< 20 min	4% (-56%)	6% (-38%)
20-40 min	9% (-11%)	8% (-25%)
41-60 min	12% (22%)	9% (-13%)
>60 min	14% (37%)	13% (34%)

Mobility decision		Residential mobility
No change		9%
Wife	Husband	
<40,000 yuan	<40,000 yuan	8% (-11%)
	50,000-80,000 yuan	7% (-21%)
	90,000-150,000 yuan	11% (21%)
	>150,000 yuan	14% (58%)
50,000-80,000 yuan	<40,000 yuan	8% (-11%)
	50,000-80,000 yuan	6% (-32%)
	90,000-150,000 yuan	9% (0%)
	>150,000 yuan	12% (32%)
90,000-150,000 yuan	<40,000 yuan	15% (68%)
	50,000-80,000 yuan	13% (42%)
	90,000-150,000 yuan	16% (74%)
	>150,000 yuan	19% (116%)
>150,000 yuan	<40,000 yuan	17% (90%)
	50,000-80,000 yuan	16% (79%)
	90,000-150,000 yuan	19% (116%)
	>150,000 yuan	22% (147%)

**Table 7.10** Conditional probability and relative probability differences (in parentheses) of residential mobility at time t given different levels of annual income at time t -1 of the spouses

## 7.4.2.4 Household mobility decisions given wife and husband's annual income

The difference in probabilities of various life course mobility decisions given different levels of annual income for both wife and husband are shown in Tables 7.10, 7.11 and 7.12. Results show that households with a high income have a larger probability to move house and change car. The probability of moving house and of changing car for both wives and husbands with the highest annual income (more than 150,000 yuan/year) are 8.7 times and 3.4 times higher than for households with the lowest annual income (less than 40,000 yuan/year for both wife and husband). Another notable finding is that the wife's income plays an important role in various household mobility decisions. Households with the wife having the lowest annual income (less than 40,000 yuan/year) and the husband having the highest annual income (more than 150,000 yuan/year) have a 14% probability of moving house. On the other hand, the probability of moving house for households with the same total annual income but

different contribution between members (i.e. wife has the highest and husband has the lowest annual income) is 17%. Similar results are found for car ownership change.

Table 7.11	Conditional	probability a	nd relativ	ve probabili	ity differe	ences (in	parenth	leses)
of car owner	ship change	at time t give	en differ	ent levels o	f annual	income a	at time <i>t</i>	t-1 of
the spouses								

Mobility decision		Car ownership change
No change		13%
Wife	Husband	
<40,000 yuan	<40,000 yuan	13% (-4%)
	50,000-80,000 yuan	11% (-15%)
	90,000-150,000 yuan	14% (10%)
	>150,000 yuan	17% (31%)
50,000-80,000 yuan	<40,000 yuan	12% (-8%)
	50,000-80,000 yuan	10% (-23%)
	90,000-150,000 yuan	12% (-8%)
	>150,000 yuan	10% (-23%)
90,000-150,000 yuan	<40,000 yuan	19% (46%)
	50,000-80,000 yuan	16% (19%)
	90,000-150,000 yuan	19% (46%)
	>150,000 yuan	22% (69%)
>150,000 yuan	<40,000 yuan	21% (62%)
	50,000-80,000 yuan	19% (46%)
	90,000-150,000 yuan	23% (73%)
	>150,000 yuan	24% (85%)

**Table 7.12** Conditional probability and relative probability differences (in parentheses) of job mobility at time t given different levels of annual income at time t -1 of the spouses

Mobility decision	Job mobility (Wife)	Job mobility (Husband)
No change	10%	9%
<40,000 yuan	17% (67%)	10% (8%)
50,000-80,000 yuan	9% (67%)	7% (25%)
90,000-150,000 yuan	8% (-22%)	3% (-63%)
>150,000 yuan	12% (22%)	2% (-75%)

Likewise, wife and husband's income have a direct influence on the decision of changing job. In general, the probability of changing job is relatively high for people with a relatively low income. Income has a different effect on the decision to change job for wives and husbands. When both household members have the lowest annual income (less than 40,000 yuan/year), wives have a larger probability of changing job (67% vs. 8% respectively). When having a high-paid job (150,000 yuan/year), the probability for husbands to change job is lower than for wives.

# 7.5 Conclusions and discussion

Life course analysis is a rich approach to analyze and model the interdependencies among a set of lifetime events. In contributing to this emerging field of study in transportation research, this chapter reports the main findings of a dynamic Bayesian Network derived from life trajectories of dual-earner households, considering lagged, concurrent and lead effects, in part separately for husbands and wives. Four life course domains (child birth, residential change, job change for both wife and husband, and car ownership change) are incorporated. In addition, commuting time and income of husband and wife are taken into consideration, in order to examine their potential influence on various household mobility decisions.

The model results point at some interesting findings. First, it suggests that wives need more time to change job before/after giving birth to a child. Second, the lagged, lead and concurrent effects of residential and job change for both wife and husband are found to positively influence car ownership change. Moreover, in case of the influence of changing job on car ownership change, only minor differences are found between husbands and wives. However, when both wife and husband change their job, the probability of buying or changing car increases dramatically.

These findings illustrate the richness of the suggested approach. Findings of this analysis in part confirm findings of earlier studies, mostly conducted in a European context. In addition, the differential effects complement earlier findings in life trajectory analysis, where the vast majority of studies did not involve households. Despite the convincing results, a caveat should be mentioned. Although time has been treated explicitly and some causal relationships have been constrained in the model, still Bayesian networks rely on observed co-occurrences in the data.

# 8

# **Conclusions and Discussions**

Residential and job choice are critical in understanding individuals' activity-travel behavior. The modeling and prediction of transportation mode choice depends on long-term life choices. However, few studies have reported individuals' preferences for the interdependent choice of house, job and transportation. Ignoring the effects of any long-term key events may lead to biased estimation results and therefore misleading policy recommendations or assessment of policy impacts. This PhD study therefore contributes to address an underdeveloped area of research on modeling the choice of multiple dimensions. The key assumption is that individuals and households consider different choice dimensions jointly.

This study provides insightful implication for urban planning and transportation study. First, transportation planners should realize that the decision on transportation may be considered jointly with other life domains. The results of our analyses have provided evidence that in the long-term decision context, travel behavior such as travel time appears to play a less important role. Second, people react differently to changes in life domains. Both observed and unobserved heterogeneities were observed. To the best of our knowledge, this is the first time to investigate heterogeneous preferences in the context of co-dependent mobility choices across different life domains. The results could provide useful information for planners and policy makers regarding the importance of dwelling and job market supplement of various segments of the population.

In addition, to broad our view from static to dynamic, this thesis develops an integrated framework for modelling the temporally interdependent choices related to residential change, job change and car purchasing decisions. According to our results, the proposed framework coincides with the life-oriented approach where the timing in mobility decisions play an important role. Individuals/households make their decisions based on their past experiences and future expectations. In terms of the dynamic interdependencies, concurrent and lagged effects were found larger than the lead effects, indicating that in general the probability of people taking an action to change before a triggering event is smaller than that after these events happen. Results also show that one-year lagged and lead effects are larger than two-year effects. Thus, this study sheds light on better understanding the mechanisms underlying long-term mobility decisions in response to various life course events.

Lastly, because people in a household physically share resources, most longterm decisions are household as opposed to individual decisions. In order to examine the potential influence of various household mobility decisions on wives and husbands, and how status of wives and husbands affects household decisions, a gender analysis of dynamics in life course decisions was conducted at the household level. The research contributes to the literature by providing insights into life course approach from a household perspective. To the best of our knowledge, this is the first time such a life course analysis is conducted at the households' level. Results added to the interpretability of the life course research findings in dual-earner households.

This PhD research addressed an underdeveloped area of research on modeling the co-dependent choice of multi-dimensional long-term mobility. As the complement to the single-domain oriented research, we jointly consider various key variables of different domains by presenting an integrated model between long-term residence choice, job choice, and short-term commute mode choices. However, there are still some rooms for extensions in further studies.

First, the current study is built based on the assumption that people will not change transportation modes only nor both houses or job simultaneously, but either change house or change job. We acknowledge that this assumption may not be strictly hold in actual situations since people may unavoidably change transportation mode only, and in rare cases both house and job with or without a time lag. It might be necessary to study exclusive choice options especially when the focus is more on the aspect of market shares.

Second, in the current study, we did not further examine to what extent the current situation influences the multidimensional choice decision. Individuals climb a ladder of a nicer house or job close to their expectation. Normally, people considered residential and job relocation carefully for process of moving house and changing job can be costly. Residential relocation may involve necessitate expenditure on selling their current house, removal old/purchase new furniture and redecoration of the new home. It also takes time to familiarize with the new working environment and colleagues. Furthermore, if people relocate to a remote area and lead to longer commute budget (both time and money), the process may involve family members changing job or children changing schools by force. In this context, if requisite gap (between the new residence/job and the current one) is not greater than the cost of changing, people do not consider move to the new house or change a new job. In other words, if the utility difference is too small, people will decline to change. Thus, in the future research, we plan to further examine to what extent the current situation influences the multidimensional choice decision.

Third, variance proportions explained by error terms in residential mobility, job mobility and car ownership change suggest that correlated unobserved factors simultaneously affect the decisions to move house, change job and the preferences to purchasing or changing cars, which clearly points to the need of incorporating various life course decisions in a simultaneous framework. In the future, more surveys and analyses are needed to capture the subjective factors such as attitudes and lifestyle preferences.

Fourth, related to the life-oriented approach, we have found that various life domains are interdependent. While future research may extend this analysis into different directions. If a larger sample can be obtained, the network can be realistically expanded with additional lifetime events, such as divorce, retirement, children enroll in school, etc. Further refinement can be obtained by including the aspects of daily activity-travel behavior. Moreover, rather than examining the mobility histories of dualearner households, other household types can be studied.

Finally, this study brings insight into the case of Shenyang city in China, which is relatively larger in size and population than many western cities. Some conclusions drawn may be generalized for other cities with a similar context in their policy decision making. More case studies in different cities need to be carried out in future.
# References

- Abraham, J.E. and Hunt, J.D. 1999. Policy Analysis Using the Sacramento MEPLAN Land Use–Transportation Interaction Model. Transportation Research Record: Journal of the Transportation Research Board, Vol. 1685, 199-208.
- Bagley, M.N., Mokhtarian, L.P. and Kitamura, R. 2002. A Methodology for the Disaggregate, Multidimensional Measurement of Residential Neighborhood Type. Urban Studies, Vol. 39, 689-704.
- Balbontin, C., Ortúzar, J. and Swait, J.D. 2015. A Joint Best-Worst Scaling and Stated Choice Model Considering Observed and Unobserved Heterogeneity: An Application to Residential Location Choice. Journal of Choice Modelling, Vol, 16, 1-14.
- Beck, M.J., Hess, S., Cabral, M.O. and Dubernet, I. 2017. Valuing Travel Time Savings: A Case of Short-term or Long-term Choices? Transportation Research Part E, Vol. 100, 133-143.
- Beige, S. and Axhausen, K.W. 2008. Long-term and Mid-term Mobility Decisions during the Life Course: Experiences with a Retrospective Survey. IATSS Research, Vol. 32, 16-33.
- Beige, S. and Axhausen, K.W. 2012. Interdependencies between Turning Points in Life and Long-term Mobility Decisions. Transportation, Vol. 39, 857-872.
- Beige, S. and Axhausen, K.W. 2017. The Dynamics of Commuting over the Life Course: Swiss Experiences. Transportation Research Part A, Vol. 104, 179-194.
- Bhat, C.R. 2000. Incorporating Observed and Unobserved Heterogeneity in Urban Work Travel Mode Choice Modeling. Transportation Science, Vol. 34, 228-238.

- Bhat, C.R. and Guo, J.Y. 2007. A Comprehensive Analysis of Built Environment Characteristics on Household Residential Choice and Auto Ownership Levels. Transportation Research Part B, Vol. 41, 506-526.
- Blumenberg, E. 2004. En-gendering Effective Planning: Spatial Mismatch, Low-Income Women, and Transportation Policy. Journal of the American Planning Association, Vol. 65, 269-281.
- Borgers, A.W.J. and Timmermans, H.J.P. 1993. Transport Facilities and Residential Choice Behavior: A Model of Multi-Person Choice Processes. Regional Science, Vol. 72, 45-61.
- Choocharukul, K., Tan Van, H. and Fujii, S. 2007. Psychological Effects of Travel Behavior on Preference of Residential Location Choice. Transportation Research Part A, Vol. 42, 116-124.
- Clark, B., Chatterjee, K. and Melia, S. 2016. Changes to Commute Mode: The Role of Life Events, Spatial Context and Environmental Attitude. Transportation Research Part A, Vol. 89, 89-105.
- Clark, W.A.V., Huang, Y. and Withers, S. 2003. Does Commuting Distance Matter? Commuting Tolerance and Residential Change. Regional Science and Urban Economics, Vol. 33, 199-221.
- Cooper, G.F. and Herskovits, E. 1992. A Bayesian Method for the Induction of Probabilistic Networks from Data. Machine Learning, Vol. 9, 309-347.
- Dargay, J. 2001. The Effect of Income on Car Ownership: Evidence of Asymmetry. Transportation Research Part A, Vol. 35, 807-821.
- Dargay, J.M. and van Ommeren, J.N. 2005. The Effect of Income on Commuting Time Using Panel Data. 45th Conference of the European Regional Science Association, Amsterdam.
- Dargay, J. and Hanly, M. 2007. Volatility of Car Ownership, Commuting Mode and Time in the UK. Transportation Research Part A, Vol. 41, 934-948.
- DeSalvo, J.S. and Huq, M. 2005. Mode Choice, Commuting Cost, and Urban Household Behavior. Journal of Regional Science, Vol. 45, 493-517.
- Dieleman, F.M. and Mulder, C.H. 2002. The Geography of Residential Choice. In: Aragonés, J.I., Francescato, G., Gärling, T. (Eds), Residential Environments: Choice, Satisfaction, and Behavior. Westport, CT and London: Bergin & Garvey, 35-54.

- Dubernet, I. and Axhausen, W.K. 2016. The Choice of Workplace and Residential Location in Germany. 16th Swiss transport research conference, Ascona, Switzerland, May 18-20.
- Du, H. and Mulley, C. 2006. Relationship between Transport Accessibility and Land Value: Local Model Approach with Geographically Weighted Regression. Transportation Research Record: Journal of the Transportation Research Board, Vol. 1977, 197-205.
- Elder, G.H. 1994. Time, Human Agency, and Social Change: Perspectives on the Life Course. Social Psychology Quarterly, Vol. 57, 4-15.
- Elder, G.H. 1998. The Life Course as Developmental Theory. Child Development, Vol. 69, 1-12.
- Eluru, N., Sener, I.N., Bhat, C.R., Pendyala, R.M. and Axhausen, K.W. 2009. Understanding Residential Mobility: Joint Model of the Reason for Residential Relocation and Stay Duration. Transportation Research Record, Transportation Research Record: Journal of the Transportation Research Board, Vol. 2133, 64-74.
- Eluru, N., Bhat, C.R., Pendyala, R.M., and Konduri, K.C. 2011. A Joint Flexible Econometric Model System of Household Residential Location and Vehicle Fleet Composition/Usage Choices. Transportation, Vol. 37, No. 4, 603-626.
- Ettema, D. 2010. The Impact of Telecommuting on Residential Relocation and Residential Preferences: A Latent Class Modeling Approach. Transportand Land Use, Vol. 3, 7-24.
- Fatmi, M.R. and Habib, M.A. 2016. A Longitudinal Vehicle Transaction Model: Assessing the Lead and Lagged Effects of Longer-term Changes and Life-cycle Events. Transportation Research Record: Journal of the Transportation Research Board, Vol. 2566. 11-21.
- Feijten, P., Hooimeijer, P. and Mulder, C.H. 2008. Residential Experience and Residential Environment Choice over the Life-course. Urban Studies, Vol. 45, 141-162.
- Gordon, P., Richardson, H.W. and Jun, M.J. 1991. The Commuting Paradox Evidence from the Top Twenty, Journal of the American Planning Association, Vol. 57, 416-420
- Grava, S. 2003. Urban Transportation Systems: Choices for Communities. McGraw-Hill, New York.

- Greene, W.H. and Hensher, D.A. 2007. Heteroscedastic Control for Random Coefficients and Error Components in Mixed Logit. Transportation Research Part E, Vol. 43, 610-623.
- Guerra, E. 2015. Has Mexico City's Shift to Commercially Produced Housing Increased Car Ownership and Car Use? Journal of Transport and Land Use, Vol. 8, 171-189.
- Habib, M.A., Miller, E.J. and Mans, B.T. 2011. Modeling of Job Mobility and Location Choice Decisions. Transportation Research Record: Journal of the Transportation Research Board, Vol. 2255, 69-78.
- Handy, S., Cao, X. and Mokhtarian, P. 2005. Correlation or Causality between the Built Environment and Travel Behavior? Evidence from Northern California. Transportation Research Part D, Vol. 10, 427-444.
- Heckerman, D. 1995. A Tutorial on Learning with Bayesian Networks. Technical Report MSR-TR–95–06, Microsoft Research.
- Heinen, E., Maat, K. and Wee, B. 2013. The Effect of Work-related Factors on the Bicycle Commute Mode Choice in the Netherlands. Transportation, Vol. 40, 23-43.
- Hensher, D., Rose, J.M. and Green, W. 2015. Applied Choice Analysis 2nd Edition. Cambridge University Press.
- Hensher, D. and Green, W. 2003. The Mixed Logit Model: The State of Practice. Transportation, Vol. 30, 133-176.
- Kan, K. 2000. Dynamic Modeling of Housing Tenure Choice. Journal of Urban Economics, Vol. 48. 46-69.
- Khattak, A.J. and Rodriguez, D. 2005. Travel Behavior in Neo-traditional Neighbourhood Developments: A Case Study in USA. Transportation Research Part A, Vol. 39, 481-500.
- Kim, C. 2008. Commuting Time Stability: A Test of a Co-location Hypothesis. Transportation Research Part A, Vol. 42, 524-544.
- Kim, J.H., Pagliara, F. and Preston, J. 2003. An Analysis of Residential Location Choice Behavior in Oxfordshire, UK: A Combined Stated Preference Approach. International Review of Public Administration, Vol. 8, 103-114.
- Kim, J.H., Pagliara, F. and Preston, J. 2005. The Intention to Move and Residential Location Choice Behavior. Urban Studies, Vol. 42, 524-544.

- Kortum, K., Paleti, R., Bhat, C.R. and Pendyala, R. 2012. A Joint Model of Residential Choice and Underlying Causal Factors. Transportation Research Record: Journal of the Transportation Research Board, Vol. 2303, No. 1, 28-37.
- Kulu, H. 2008. Fertility and Spatial Mobility in the Life course: Evidence from Austria. Environment and Planning A, Vol. 40, 632-652.
- Lanzendorf, M. 2003. Mobility Biographies: A New Perspective for Understanding Travel Behavior. Paper presented at the 10th Conference of the International Association of Travel Behavior Research, Lucerne, Switzerland.
- Lanzendorf, M. 2010. Key Events and Their Effect on Mobility Biographies: The Case of Child Birth. International Journal of Sustainable Transportation, Vol. 4, 272-292.
- Lerman, S.R. 1976. Location, Housing, Automobile Ownership and Mode to Work: A Joint Choice Model. Transportation Research Record: Journal of Transportation Research Board, No. 610, 6-11.
- Louviere, J. and Timmermans, H.J.P. 1990. Hierarchical Information Integration Applied to Residential Choice Behavior. Geographical Analysis, Vol. 22, 127-144.
- Manaugh, K., Miranda-Moreno, L. and EI-Geneidy, A. 2010. The Effect of Neighbourhood Characteristics, Accessibility, Home–work Location, and Demographics on Commuting Distances. Transportation, Vol. 37, 627-646.
- Meixell, M.J. and Norbis, M. 2008. A Review of the Transportation Mode Choice and Carrier Selection Literature. Journal of Logistics Management, Vol. 19, 183-211.
- Meyer, R.J., Levin, I.P. and Louviere, J.J. 1978. Functional Analysis of Mode Choice. Transportation Research Record, No. 673, 1-7.
- Michielin, F. and Mulder, C. 2008. Family Events and the Residential Mobility of Couples. Environment and Planning A, Vol. 40, 2770-2790.
- Murphy, K.P. 2003. Dynamic Bayesian Networks: Representation, Inference and Learning. Ph.D. Thesis, UC Berkeley, USA.
- Ng, C.F. 2008. Commuting Distances in a Household Location Choice Model with Amenities. Journal of Urban Economics, Vol. 63, 116-129.
- Oakil, A.T.M. 2016. Securing or Sacrificing Access to a Car: Gender Difference in the Effects of Life Events. Travel Behaviour and Society, Vol. 3, 1-7.
- Oakil, A.T.M., Ettema, D.F., Arentze, T.A. and Timmermans, H.J.P. 2011. Longitudinal Model of Longer Term Mobility Decisions: Framework and First Empirical Test. Journal of Urban and Development, Vol. 137, 220-229.

- Oakil, A.T.M., Ettema, D.F., Arentze, T.A. and Timmermans, H.J.P. 2014. Changing Household Car Ownership Level and Life Cycle Events: An Action in Anticipation or an Action on Occurrence. Transportation, Vol. 41, 889-904.
- Paleti, R., Bhat, C.R. and Pendyala, R. 2012. An Integrated Model of Residential Location, Work Location, Vehicle Ownership, and Commute Tour Characteristics. Transportation Research Record: Journal of the Transportation Research Board, Vol. 2382, 162-172.
- Paulssen, M., Temme, D., Vij, A. and Walker, J. 2014. Values, Attitudes and Travel Behavior: A Hierarchical Latent Variable Mixed Logit Model of Travel Mode Choice. Transportation, Vol. 41, 873-888.
- Peer, S., Verhoef, E.T., Knockaert, J., Koster, P. and Teseng, Y.Y. 2015. Long-run versus Short-run Perspectives on Consumer Schedeling: Evidence from A Revealed-preference Experiment among Peak-hour Road Commuters. International Economic Review, Vol. 56, 303-323.
- Pérez, P.E., Martínez, F.J. and Ortúzar, J.de D. 2003. Microeconomic Formulation and Estimation of a Residential Location Choice Model: Implications for the Value of Time. Journal of Regional Science, Vol. 43, 771-789.
- Pinjari, A.R., Pendyala, R.M., Bhat, C.R. and Waddell, P.A. 2011. Modeling the Choice Continuum: An Integrated model of Residential Location, Auto Ownership, Bicycle Ownership, and Commute Tour Mode Choice Decision. Transportation, Vol. 38, 933-958.
- Plaut, P.O. 2006. The Intra-household Choices Regarding Commuting and Housing. Transportation Research Part A, Vol. 40, 561-571.
- Prillwitz, J., Harms, S. and Lanzendorf, M. 2006. Impact of Life-course Events on Car Ownership. Transportation Research Record: Journal of the Transportation Research Board, Vol. 1986, 91-77.
- Quarmby, D.A. 1967. Choice of Travel Mode for the Journey to Work: Some Findings. Journal of Transport Economics and Policy, Vol. 1, 273-314.
- Rashidi, T.H., Mohammadian, A. and Koppelman, F.S. 2011. Modeling Interdependencies between Vehicle Transaction, Residential Relocation and Job Change. Transportation, Vol 38, 909-932.
- Rich, J. and Nielsen, O. 2001. A Microeconomic Model for Car Ownership, Residence and Work Location. European Transport Conference, PTRC, Cambridge, UK.

- Rodriguez, D.A. and Joo, J. 2004. The Relationship between Non-motorized Mode Choice and the Local Physical Environment. Transportation Research Part D, Vol. 9, 151-173.
- Roorda, M.J., Carrasco, J.A. and Miller, E.J. 2009. An Integrated Model of Vehicle Transactions, Activity Scheduling and Mode Choice. Transportation Research Part B, Vol. 43, 217-229.
- Rose, J.M., Bliemer, M.C.J, Hensher, D.A. and Collins, A.T. 2008. Designing Efficient Stated Choice Experiments in the Presence of Reference Alternatives. Transportation Research Part B, Vol. 42, 395-406.
- Rossi, P.H. 1980. Why Families Move. Sage Publications, Beverley Hills, CA.
- Salon, D. 2006. Cars and the City: An Investigation of Transportation and Residential Location Choices in New York City. Dissertation, University of California.
- Salon, D. 2009. Neighborhoods, Cars, and Commuting in New York City: A Discrete Choice Approach, Transportation Research Part A, Vol. 43, 180-196.
- Sandowa, E. and Westin, K. 2010. The Persevering Commuter Duration of Long-Distance Commuting. Transportation Research Part A, Vol. 44, 433-445.
- Scheiner, J. 2014. Gendered Key Events in the Life Course: Effects on Changes in Travel Mode Choice over Time. Journal of Transport Geography, Vol. 37, 47-60.
- Scheiner, J. and Holz-Rau, C. 2013a. A Comprehensive Study of Life Course, Cohort, and Period Effects on Changes in Travel Mode Use. Transportation Research Part A, Vol. 47, 167-181.
- Scheiner, J. and Holz-Rau, C. 2013b. Changes in Travel Mode Use after Residential Relocation: A Contribution to Mobility Biographies. Transportation, Vol. 40, 431-458.
- Schirmer, P. M., Van Eggermond. M. and Axhausen, K.W. 2014. The Role of Location in Residential Location Choice Models: A Review of Literature. Transport and Land Use, Vol. 7, 3-21.
- Schwanen, T. and Mokhtarian, P.L. 2005a. What Affects Commute Mode Choice: Neighbourhood Physical Structure or Preferences toward Neighborhoods? Transport Geography, Vol. 13, 83-99.
- Schwanen, T. and Mokhtarian, P.L. 2005b. What If You Live in the Wrong Neighbourhood? The Impact of Residential Neighbourhood Type Dissonance on Distance Traveled. Transportation Research Part D, Vol. 10, 127-151.

- Tillema, T., Van Wee, B. and Ettema, D. The Influence of (Toll-realted). 2010. Travel Costs in Residential Location Decisions of Households: A Stated Choice Approach. Transportaiton Research Part A, Vol. 44, 752-796.
- Timmermans, H.J.P, Borgers, A.W.J., Van Dijk, J. and Oppewal, H. 1992. Residential Choice Behaviour of Dual Earner Households: A Decompositional Joint Choice Model. Environment and Planning A, Vol. 24, 517-33.
- Timmermans, H.J.P. and Zhang, J. 2009. Modeling Household Activity Travel Behavior: Examples of State of the Art Modeling Approaches and Research Agenda. Transportation Research Part B, Vol. 43, 187-190.
- Tran, T.M., Zhang, J., Chikaraishi, M. and Fujiwara, A. 2016. A Joint Analysis of Residential, Work Location and Commuting Mode Choice in Hanoi, Vietnam. Transport Geography, Vol. 54, 181-193.
- Tu Tran, M., Zhang, J., Chikaraishi, M. and Fujiwara, A. 2016. A Joint Analysis of Residential Location, Work Location and Commuting Mode Choices in Hanoi, Vietnam. Journal of Transport Geography, Vol. 84, 181-193.
- Van Acker, V., Van Wee, B. and Witlox, F. 2010. When Transport Geography Meets Social Psychology: Toward a Conceptual Model of Travel Behaviour. Transport Reviews, Vol. 30, 219-240.
- Van Ommeren, J. 1999. Job Moving, Residential Moving, and Commuting: A Search Perspective. Journal of Urban Economics, Vol. 46, 230-253.
- Verhoeven, M., Arentze, T.A., Timmermans, H.J.P. and Van der Waerden, P.J.H.J. 2005. Modelling the Impact of Key Events on Long-term Transport Mode Choice Decisions: Decision Network Approach Using Event History Data. Transportation Research Record: Journal of the Transportation Research Board, Vol. 1926, 106-114.
- Verhoeven, M., Arentze, T.A., Timmermans, H.J.P. and Van der Waerden, P.J.H.J. 2007. Examining Temporal Effects of Lifecycle Events on Transport Mode Choice Decisions. International Journal of Urban Sciences, Vol. 11, 1-13.
- Wang, B., Rasouli, S., Timmermans, H.J.P. and Shao, C. 2018. Relationships between Consecutive Long-term and Mid-term Mobility Decisions over the Life Course: A Bayesian Network Approach. Paper presented at the Transportation Research Board 87th Annual Meeting, Washington, DC.
- Walker, J. and Li, J. 2007. Latent Lifestyle Preferences and Household Location Decisions. Journal of Geographical Systems, Vol. 9, 77-101.

- Warner, C. and Sharp, G. 2015. The Short-and Long-term Effects of Life Events on Residential Mobility. Advances in Life Course Research, Vol. 27, 1-15.
- Yamamoto, T. 2008. The Impact of Life-Course Events in Vehicle Ownership Dynamics: the Cases of France and Japan. IATSS Research, Vol. 32, 34-43.
- Yu, B., Zhang, J. and Li, X. 2017. Dynamic Life Course Analysis on Residential Location Choice. Transportation Research Part A, Vol. 104, 281-292.
- Zhao, P. and Zhang, Y. 2018. Travel Behaviour and Life Course: Examining Changes in Car Use after Residential Relocation in Beijing. Journal of Transport Geography, Vol. 73, 41-53.

# **Author Index**

### A

Abraham, J.E., 64

### В

Bagley, M.N., 15 Balbontin, C., 15 Beck, M.J., 33 Beige, S., 7, 8, 20, 50 Bhat, C.R., 1, 2, 6, 51 Blumenberg, E., 38 Borgers, A.W.J., 10

### С

Choocharukul, K., 8 Clark, B., 2 Clark, W.A.V., 49 Cooper, G.F., 67

# D

Dargay, J., 6, 38 DeSalvo, J.S., 6 Dieleman, F.M., 2 Dubernet, I., 33, 41 Du, H., 34

### Е

Elder, G.H., 7

Eluru, N., 2, 50, 51 Ettema, D., 6

# F

Fatmi, M.R., 8 Feijten, P., 27

# G

Gordon, P., 64 Grava, S., 1 Greene, W.H., 24, 38 Guerra, E., 6

### Н

Habib, M.A., 7, 8, 49, 71, 74 Handy, S., 6 Heine, D., 10, 67 Heinen, E., 1 Hensher, D., 16, 24

# K

Kan, K., 51 Khattak, A.J., 6 Kim, C., 8 Kim, J.H., 6, 33 Kortum, K., 51 Kulu, H., 27

### L

Lanzendorf, M., 7, 8, 49 Lerman, S.R., 6 Louviere, J., 15

# М

Manaugh, K., 50, 51 Meixell, M.J., 1 Meyer, R.J., 1 Michielin, F., 50 Murphy, K.P., 66

# Ν

Ng, C.F., 6

# 0

Oakil, A.T.M., 8, 9, 10, 20, 49

### Ρ

Paleti, R., 2, 7, 50 Paulssen, M., 1 Peer, S., 35 Pérez, P.E., 35 Pinjari, A.R., 2, 8, 52, 53 Plaut, P.O., 53 Prillwitz, J., 9, 10, 51

# Q

Quarmby, D.A., 1

# R

Rashidi, T.H., 6, 8 Rich, J., 6 Rodriguez, D.A., 1 Roorda, M.J., 10, 75 Rose, J.M., 16 Rossi, P.H., 7

# S

Salon, D., 6 Sandowa, E., 38 Scheiner, J., 6, 8, 10 Schirmer, P. M., 41 Schwanen, T., 6

# Т

Tillema, T., 33 Timmermans, H.J.P., 10, 15, 75 Tran, T.M., 7, 15, 51

# V

Van Acker, V., 51 Van Ommeren, J., 15 Verhoeven, M., 7, 9, 20,57, 66

### W

Wang, B., 9, 27, 66 Walker, J., 15 Warner, C., 2

# Y

Yamamoto, T., 8 Yu, B., 8

# Ζ

Zhao, P., 49

# Subject Index

### B

Bayesian network, 8, 9, 65, 66, 70, 90

### С

Car ownership, 3, 6, 7, 8, 9, 20, 49, 50, 51, 52, 53, 54, 55, 56, 57, 58, 60, 64, 67, 68, 69, 71, 73, 74, 75, 78, 79, 81, 82, 83, 84, 85, 86, 87, 89, 90, 111 Child birth, 2, 3, 7, 8, 20, 52, 54, 57, 60, 67, 69, 72, 75, 78, 79, 82, 83, 84, 90, 96, 119 Concurrent effect, 57, 58, 61, 69, 72, 82, 84, 90 Co-dependent, 2, 23, 46, 49, 91, 92, 109 Commuting mode, 2, 6, 7, 15, 51 Commuting time, 1, 6, 8, 20, 32, 33, 38, 65, 68, 74, 79, 85, 86, 87, 90, 109, 111 Conditional probability, 65, 68, 70, 71, 72, 83, 84, 85, 86, 87, 89

# D

Dynamic, 2, 3, 9, 11, 66, 70, 78, 79, 92, 111 Dynamic Bayesian network, 4, 9, 63,

64, 65, 66, 67, 74, 90, 111

Discrete choice model, 3, 7 Dual-earner household, 5, 78, 86, 90, 92, 111

# E

Error component, 3, 24, 26, 38, 39, 40, 41, 42, 43, 45, 46, 47, 60, 110 Experiment, 3, 13, 15, 16, 17, 18, 22, 24, 26, 39, 41, 46, 109 Expectation, 3, 5, 33, 50, 51, 58, 67, 68, 92, 93

# F

Family structure change, 67, 69, 71, 72

# Н

Heterogeneity, 2, 23, 24, 26, 33, 38, 39, 40, 42, 45, 46, 47, 51, 109, 110 Household, 2, 3, 5, 7, 8, 9, 10, 11, 20, 23, 24, 25, 26, 27, 34, 42, 46, 47, 49, 51, 52, 53, 54, 55, 56, 57, 58, 60, 64, 65, 66, 69, 72, 75, 77, 78, 79, 80, 81, 82, 83, 84, 85, 86, 87, 88, 90, 91, 92, 93, 109, 111

# I

Interdependency, 2, 6, 64, 65

### J

Job mobility, 7, 19, 32, 47, 53, 55, 57, 58, 60, 64, 67, 69, 71, 72, 73, 74, 78, 83, 85, 87, 89, 93, 110

### L

- Lag effect, 84
- Latent class model, 3, 24, 26, 33, 35, 46, 47, 110
- Lead effect, 50, 53, 55, 57, 58, 61, 63, 64, 65, 67, 68, 69, 70, 72, 73, 75, 79, 82, 84, 90, 92,110, 111
- Learning, 9, 66, 67, 68, 70, 79, 82, 83
- Life course, 2, 3, 4, 5, 7, 8, 9, 10, 13, 20, 27, 47, 50, 51, 52, 57, 60, 61, 63, 64, 65, 66, 67, 68, 69, 71, 74, 75, 77, 79, 81, 84, 88, 90, 92, 93
- Life events, 2, 3, 7, 8, 9, 10, 20, 22, 52, 53, 56, 63, 68, 72, 74, 77, 80
- Life trajectory, 4, 9, 39, 49, 63, 66, 75, 78, 90, 111
- Location choice, 6, 41, 50, 51, 68, 69
- Longitudinal, 20, 47, 52, 68,
- Long-term, 2, 3, 6, 7, 8, 9, 10, 13, 17, 18, 20, 24, 32, 33, 39, 42, 47, 49, 50, 51, 52, 53, 54, 60, 64, 65, 66, 74, 75, 77, 78, 79, 84, 85, 91, 92, 110, 111

### Μ

Marriage, 3, 7, 8, 27, 52, 54, 57, 59, 60, 72
Mixed logit model, 3, 24, 26, 28, 33, 38, 41, 42, 43, 46, 47, 110
Mobility decision, 2, 3, 4, 5, 6, 7, 8, 10, 11, 15, 19, 20, 22, 23, 32, 33, 47, 49, 50, 51, 52, 53, 54, 55, 55, 56, 57, 60, 63, 64, 65, 67, 68, 69, 70,

71, 72, 73, 74, 75, 77, 78, 79, 82, 83, 84, 85, 86, 87, 88, 89, 90, 92, 109, 110, 111 Multidimensional choice, 3, 5, 7, 10, 24,

46, 93, 109, 110

68, 70, 78, 79, 111

### R

Relative probability, 70, 71, 83, 84, 86, 87, 88, 89 Relocate, 58, 60, 68, 93 Residential mobility, 3, 6, 20, 34, 46, 51, 53, 55, 57, 58, 60, 65, 69, 72, 73, 74, 78, 83, 85, 86, 88, 93, 110 Retrospective, 3, 4, 13, 20, 50, 52, 60,

### S

- Simultaneous equation, 4, 50, 51, 54, 55, 56, 60, 63 Shenyang, 20, 22, 27, 93, 109 Short-term, 7, 13, 17, 32, 50, 92 Socio-demographic, 24, 26, 34, 42, 52, 56, 65, 67, 69, 71
- Stated choice, 3, 13, 15, 16, 24, 46, 109

### Т

Temporal effects, 5, 9, 10, 55, 57, 58, 73, 79, 82, 84

# U

Updated probability, 70, 71

Utility, 1, 2, 5, 23, 33, 38, 39, 40, 46, 93, 109

# Summary

#### Job-housing Co-dependent Mobility Decisions in Life Trajectories

An effective instrument of urban planning to alleviate congestion is to create a jobhousing balance, which depends on the co-localisation of housing and labour. Recently, the view has re-emerged that people's decisions with respect to different life-domains should be treated as a 'bundle' choice. In that sense, the job (location) choice, residential choice, commuting times, transportation mode choice and underlying vehicle possession are strongly interdependent. Rather than maximizing the utility of each of these choice facets separately and independently, it is more realistic to assume that individuals/households consider the multidimensionality and maximize the utility of the multidimensional profile (or apply another choice mechanism to the multidimensional profile).

To mimic the complexity of the decision-making process under investigation in the real world, a stated choice experiment was designed to mimic the multidimensional choice behavior of interest. Instead of creating an experimental design that is the same for all respondents, we generated an efficient design in which attribute values were pivoted based on the real-world data of individual respondents.

The data was collected in Shenyang, China, in 2016. Respondents were selected at random from five main districts in the central city and four other districts in the surrounding, new development areas. Considering the aim of the study, we only interviewed respondents who had a job.

Based on data collected in Shenyang, China, a Mixed Multinomial Logit (MMNL) model with panel effects, which allows for unobserved heterogeneity in individual preferences, was estimated to capture the effects of different residential, work and commuting attributes on the multidimensional choice, accounting for the panel nature of the data. Our findings indicate that, 1) Housing tenure, size, price, distance to the

#### Summary

bus stop, housing location are important housing characteristics that help explaining the residential mobility choice process. Salary, work type, co-worker relationships and work environment are significant factors in the job mobility choice process. 2) Time-related factors influence commute mode choice. Choice of public transportation modes is sensitive to commuting costs while car mode choice is not. 3) People are relatively satisfied with their current situation and do not frequently make changes. Furthermore, people are less inclined to move house relative to changing job. 4) Unobserved heterogeneity and demographic characters both affect the multiple dimensions of choices.

Moreover, a latent class model was adopted to account for individuals' preference heterogeneity which assume a finite mixture of the latent groups to explore preference heterogeneity. Results based on a two-class model show that income is the main variable explaining class membership. The results of the heterogeneous behavior indicate that people with lower income are less likely to change their current house and/or job than people with higher income. In addition, people with lower income prefer public transportation modes over private cars. In terms of slow modes, people with lower income are more likely to cycling than walking. While these effects are shown reversely for people with higher income. These findings could provide useful information for planners and policy makers regarding the importance of dwelling and job market supplement of various segments of the population.

Third, to allow for the possibility that unobserved preferences for transportation models depend on long-term choice behavior, specific error components are identified and the variance of these error components is estimated through parameterization of their heteroscedasticity. Thus, we estimate an error component mixed logit model to identify random and systematic long-term choice specific heterogeneity. The results of the estimated error component mixed logit model with panel effects indicate that most selected attributes of the residential environment, job profile and transportation mode are significantly related to individual differences in multidimensional choices. Moreover, the estimation of various sources of unobserved heterogeneity signals significant unobserved heterogeneity in selected taste parameters, and choice dependent heteroscedasticity in error component variance.

From a behavioral perspective, people's long-term mobility decisions may depend on their current situation, past experiences and/or future plans. Consequently, models of long-term mobility decisions should take lagged, concurrent and/or lead effects into account. In turn, long-term mobility decisions may be the consequence of particular events in other life domains. Contributing to the literature on long-term mobility analysis, this study develops an integrated framework for modeling the dynamic, interdependent choices related to residential move, job changes and mobility tools transactions. Using retrospective life trajectory data collected through a Webbased survey, a dynamic Bayesian network model is estimated. Results show that different life domains are highly interdependent. Concurrent, as well as lagged and lead effects of various mobility decisions are observed.

In addition, because people in multi-earner households share resources, most long-term decisions are made by more than one household member. Thus, to contribute to the further development of the relatively thin line of research in transportation studies, a dynamic Bayesian network approach is proposed from a household perspective. Results show that the effects of child birth are much larger on residential and car ownership change than on job change for both household heads in dual-earner households. Moreover, the probability of residential and car ownership change increases when both spouses have relatively long commuting times. In case only the husband faces an excessive commuting time, households have a larger probability of moving house or purchasing an additional car. By contrast, in case only the wife faces an excessive commuting time, she is more likely to change job rather than the household taking particular actions to adjust to the problematic situation.

# **Curriculum Vitae**

Jia Guo was born on June, 1987 in Shenyang, China. She completed her bachelor and master degree in Urban Planning and Design at Shenyang Architectural University in 2011 and 2015, respectively. After obtaining her master's degree, she joined the Urban Planning and Transportation Group of the Department of Built Environment at TU/e and started working on her PhD project. Her primary research interests include travel behavior analysis, transportation planning, residential location choice, and household long-term decisions.

# Acknowledgements

Four years ago, I decided to pursue a Ph.D. in the Netherlands. Four years later, I feel very grateful for that decision. Along this journey, I have not only built great passion for research but also developed more optimistic attitude towards life. If time flied back and I was allowed to choose again, I would definitely make the same decision.

I have got plenty of support along this journey. I would even wonder whether I can manage to go this long and accomplish my dissertation without all the help. First of all, I want to express my sincere thanks to my supervisor Prof. Harry Timmermans, who inspired this thesis, shared his expertise and ideas as well as gave me critical comments and advices. Working with him gives me a deep understanding about what makes a good researcher and I also found my role model for future career. Besides working, his attitudes towards life, optimism and great leadership also deeply influenced me.

My gratitude also goes to my second supervisor Tao Feng. This thesis wouldn't be possible without his help. I appreciate all his contribution of time and ideas on this thesis. He has given me plenty of suggestions to develop the research proposal, design the survey, and improve the models. Furthermore, Tao is also a good life guide. When I am confused about the future career, he was the one who shown me his experiences and provided very valuable suggestions.

My thankfulness is also extended to my committee members, Prof. Dr. Dick Ettema, Prof. Dr. Soora Rasouli, Dr. Eric Molin, and Dr. Dujuan Yang. Thank you for spending the time and efforts to read the manuscript of my thesis and giving feedback. I appreciate all your critical comments and suggestions, which helped me to improve my work.

#### Acknowledgements

I would particularly like to acknowledge my teachers, peer researchers, and friends in China who participated in the in-depth interviews and provided valuable advice and comments. In addition, I am especially grateful to all the interviewees and survey participants. Without their co-operation, none of this would have been possible.

I owe special thanks to my colleagues in the Urban Planning and Transportation group Aloys, Bilin, Bobin, Calvin, Chenchen, Chengyu, Chaoyu, Chenyu, Dujuan, Dong, Feixiong, Jiajia, Jing, Gaofeng, Guangde, Hexin, Juan, Marianna, Qianqian, Qiyao, Rainbow, Shangqi, Sunghoon, Valeria, Weiming, Wen, Wenshu, Wensi, Xiaochen, Xiaofeng, Xiaoming, Xiaoyue, Xueting, Yajie, Yanan, Yang, Yonglei, Zhengying, Zhihui, Zhong, and Zuoxian. I learned many things during working with them. Every time when I had a question there was always someone willing to help me out. They have all contributed to a pleasant and inspiring work environment. I really enjoyed the time we spent together during the four years. My appreciation also goes to Mandy, Marielle and Peter for their administrative and technical assistance.

Living abroad for four years is a challenging but beautiful journey. Thanks to all my friends, who made me enjoy my time in the Netherlands. I have received lots of encouragement and friendship, and turned my PhD study into a wonderful experience. I will remember your support and help in work and life during these years forever. Thank you for all the great moments. I am grateful to you all!

Last but not least, I sincerely thank my parents and relatives for their unconditional love, understanding of all my choices, and support in every aspect.

Thank you all for everything!

Jia Guo Eindhoven, June, 2020

# **Publication List**

#### **Journal papers**

- Guo, J., Feng, T. and Timmermans, H.J.P. 2018. Modeling Co-dependent Choice of Workplace, Residence and Commuting Mode using An Error Component Mixed Logit Model. Transportation, Vol. 47, 911-933.
- Guo, J., Feng, T. and Timmermans, H.J.P. 2020. Co-Dependent Workplace, Residence and Commuting Mode Choice: Results of a Multi-Dimensional Mixed Logit Model with Panel Effects. Cities, Vol. 96, 102448.
- Guo, J., Feng, T. and Timmermans, H.J.P. 2019. Time-varying Dependencies among Mobility Decisions and Key Life Course Events: An Application of Dynamic Bayesian Decision Networks. Transportation Research Part A, Vol 130, 82-92.
- Guo, J., Feng, T., Zhang, J. and Timmermans, H.J.P. 2020. Temporal Interdependencies in Mobility Decisions over the Life course: A Household-level Analysis using Dynamic Bayesian Networks. Journal of Transport Geography, Vol 82.

#### **Conference papers**

- Guo, J., Feng, T., Zhang, J. and Timmermans, H.J.P. 2019. Temporal Interdependencies in Mobility Decisions over the Life Course: A Family-Based Analysis Using Dynamic Bayesian Networks. 98th Annual Meeting of the Transportation Research Board (TRB), Washington DC, USA.
- Guo, J., Feng, T. and Timmermans, H.J.P. 2019. Taste Variation in Co-dependent Choice of Residence, Job and Transportation Mode. 98th Annual Meeting of the Transportation Research Board (TRB), Washington DC, USA.

- Guo, J., Feng, T. and Timmermans, H.J.P. 2018. Modeling Co-dependent Choice of Workplace, Residence and Commuting Mode Using an Error Component Mixed Logit Model. 97th Annual Meeting of the Transportation Research Board (TRB), Washington DC, USA.
- Guo, J., Feng, T. and Timmermans, H.J.P. 2018. Analysis of Temporal Dependencies in the Occurrence of Life Events Using Dynamic Bayesian Networks. hEART 2018 – 7th Symposium of the European Association for Research in Transportation. Technical University of Munich.
- Guo, J., Feng, T. and Timmermans, H.J.P. 2017. Co-Dependent Workplace, Residence and Commuting Mode Choice: Analysis of a Multi-Dimensional Pivoted Stated Choice Experiment. NECTAR 14th, Madrid.

**Bouwstenen** is een publicatiereeks van de Faculteit Bouwkunde, Technische Universiteit Eindhoven. Zij presenteert resultaten van onderzoek en andere activiteiten op het vakgebied der Bouwkunde, uitgevoerd in het kader van deze Faculteit.

**Bouwstenen** en andere proefschriften van de TU/e zijn online beschikbaar via: https://research.tue.nl/

Kernredactie MTOZ Reeds verschenen in de serie Bouwstenen

#### nr 1

**Elan: A Computer Model for Building Energy Design: Theory and Validation** Martin H. de Wit H.H. Driessen R.M.M. van der Velden

#### nr 2

#### Kwaliteit, Keuzevrijheid en Kosten: Evaluatie van Experiment Klarendal, Arnhem

J. Smeets C. le Nobel M. Broos J. Frenken A. v.d. Sanden

#### nr 3

Crooswijk: Van 'Bijzonder' naar 'Gewoon' Vincent Smit Kees Noort

nr 4 **Staal in de Woningbouw** Edwin J.F. Delsing

#### nr 5

Mathematical Theory of Stressed Skin Action in Profiled Sheeting with Various Edge Conditions Andre W.A.M.J. van den Bogaard

#### nr 6

Hoe Berekenbaar en Betrouwbaar is de Coëfficiënt k in x-ksigma en x-ks? K.B. Lub A.I. Bosch

nr 7 Het Typologisch Gereedschap: Een Verkennende Studie Omtrent Typologie en Omtrent de Aanpak van Typologisch Onderzoek J.H. Luiten

nr 8 Informatievoorziening en Beheerprocessen A. Nauta Jos Smeets (red.) Helga Fassbinder (projectleider) Adrie Proveniers J. v.d. Moosdijk nr 9

Strukturering en Verwerking van Tijdgegevens voor de Uitvoering van Bouwwerken ir. W.F. Schaefer P.A. Erkelens

#### nr 10

**Stedebouw en de Vorming van een Speciale Wetenschap** K. Doevendans

nr 11 Informatica en Ondersteuning van Ruimtelijke Besluitvorming G.G. van der Meulen

#### nr 12

Staal in de Woningbouw, Korrosie-Bescherming van de Begane Grondvloer Edwin J.F. Delsing

#### nr 13

Een Thermisch Model voor de Berekening van Staalplaatbetonvloeren onder Brandomstandigheden A.F. Hamerlinck

#### nr 14

De Wijkgedachte in Nederland: Gemeenschapsstreven in een Stedebouwkundige Context K. Doevendans R. Stolzenburg

#### nr 15

Diaphragm Effect of Trapezoidally Profiled Steel Sheets: Experimental Research into the Influence of Force Application Andre W.A.M.J. van den Bogaard

#### nr 16

Versterken met Spuit-Ferrocement: Het Mechanische Gedrag van met Spuit-Ferrocement Versterkte Gewapend Betonbalken K.B. Lubir M.C.G. van Wanroy

#### nr 17 **De Tractaten van** Jean Nicolas Louis Durand G. van Zeyl

nr 18 Wonen onder een Plat Dak: Drie Opstellen over Enkele Vooronderstellingen van de Stedebouw K. Doevendans

#### nr 19

Supporting Decision Making Processes: A Graphical and Interactive Analysis of Multivariate Data W. Adams

#### nr 20

Self-Help Building Productivity: A Method for Improving House Building by Low-Income Groups Applied to Kenya 1990-2000 P. A. Erkelens

nr 21 **De Verdeling van Woningen: Een Kwestie van Onderhandelen** Vincent Smit

#### nr 22

Flexibiliteit en Kosten in het Ontwerpproces: Een Besluitvormingondersteunend Model M. Prins

#### nr 23

**Spontane Nederzettingen Begeleid: Voorwaarden en Criteria in Sri Lanka** Po Hin Thung

#### nr 24

Fundamentals of the Design of Bamboo Structures Oscar Arce-Villalobos

#### nr 25 **Concepten van de Bouwkunde** M.F.Th. Bax (red.) H.M.G.J. Trum (red.)

nr 26 **Meaning of the Site** Xiaodong Li nr 27 Het Woonmilieu op Begrip Gebracht: Een Speurtocht naar de Betekenis van het Begrip 'Woonmilieu' Jaap Ketelaar

#### nr 28

Urban Environment in Developing Countries editors: Peter A. Erkelens George G. van der Meulen (red.)

### nr 29

Stategische Plannen voor de Stad: Onderzoek en Planning in Drie Steden prof.dr. H. Fassbinder (red.) H. Rikhof (red.)

#### nr 30

**Stedebouwkunde en Stadsbestuur** Piet Beekman

#### nr 31

**De Architectuur van Djenné: Een Onderzoek naar de Historische Stad** P.C.M. Maas

#### nr 32

**Conjoint Experiments and Retail Planning** Harmen Oppewal

#### nr 33

Strukturformen Indonesischer Bautechnik: Entwicklung Methodischer Grundlagen für eine 'Konstruktive Pattern Language' in Indonesien Heinz Frick arch. SIA

#### nr 34

Styles of Architectural Designing: Empirical Research on Working Styles and Personality Dispositions Anton P.M. van Bakel

# nr 35

**Conjoint Choice Models for Urban Tourism Planning and Marketing** Benedict Dellaert

nr 36 **Stedelijke Planvorming als Co-Produktie** Helga Fassbinder (red.) nr 37 **Design Research in the Netherlands** editors: R.M. Oxman M.F.Th. Bax H.H. Achten

nr 38 **Communication in the Building Industry** Bauke de Vries

nr 39 **Optimaal Dimensioneren van Gelaste Plaatliggers** J.B.W. Stark F. van Pelt L.F.M. van Gorp B.W.E.M. van Hove

nr 40 **Huisvesting en Overwinning van Armoede** P.H. Thung P. Beekman (red.)

nr 41 **Urban Habitat: The Environment of Tomorrow** George G. van der Meulen Peter A. Erkelens

nr 42 **A Typology of Joints** John C.M. Olie

nr 43 Modeling Constraints-Based Choices for Leisure Mobility Planning Marcus P. Stemerding

nr 44 Activity-Based Travel Demand Modeling Dick Ettema

nr 45 Wind-Induced Pressure Fluctuations on Building Facades Chris Geurts

nr 46 **Generic Representations** Henri Achten

nr 47 Johann Santini Aichel: Architectuur en Ambiguiteit Dirk De Meyer nr 48 **Concrete Behaviour in Multiaxial Compression** Erik van Geel

nr 49 **Modelling Site Selection** Frank Witlox

nr 50 **Ecolemma Model** Ferdinand Beetstra

nr 51 Conjoint Approaches to Developing Activity-Based Models Donggen Wang

nr 52 **On the Effectiveness of Ventilation** Ad Roos

nr 53 **Conjoint Modeling Approaches for Residential Group preferences** Eric Molin

nr 54 **Modelling Architectural Design Information by Features** Jos van Leeuwen

nr 55 **A Spatial Decision Support System for the Planning of Retail and Service Facilities** Theo Arentze

nr 56 Integrated Lighting System Assistant Ellie de Groot

nr 57 **Ontwerpend Leren, Leren Ontwerpen** J.T. Boekholt

nr 58 **Temporal Aspects of Theme Park Choice Behavior** Astrid Kemperman

nr 59 **Ontwerp van een Geïndustrialiseerde Funderingswijze** Faas Moonen nr 60 Merlin: A Decision Support System for Outdoor Leisure Planning Manon van Middelkoop

nr 61 **The Aura of Modernity** Jos Bosman

nr 62 **Urban Form and Activity-Travel Patterns** Daniëlle Snellen

nr 63 **Design Research in the Netherlands 2000** Henri Achten

nr 64 **Computer Aided Dimensional Control in Building Construction** Rui Wu

nr 65 **Beyond Sustainable Building** editors: Peter A. Erkelens Sander de Jonge August A.M. van Vliet co-editor: Ruth J.G. Verhagen

nr 66 **Das Globalrecyclingfähige Haus** Hans Löfflad

nr 67 **Cool Schools for Hot Suburbs** René J. Dierkx

nr 68 **A Bamboo Building Design Decision Support Tool** Fitri Mardjono

nr 69 **Driving Rain on Building Envelopes** Fabien van Mook

nr 70 **Heating Monumental Churches** Henk Schellen

nr 71 **Van Woningverhuurder naar Aanbieder van Woongenot** Patrick Dogge nr 72 **Moisture Transfer Properties of Coated Gypsum** Emile Goossens

nr 73 **Plybamboo Wall-Panels for Housing** Guillermo E. González-Beltrán

nr 74 **The Future Site-Proceedings** Ger Maas Frans van Gassel

nr 75 **Radon transport in Autoclaved Aerated Concrete** Michel van der Pal

nr 76

The Reliability and Validity of Interactive Virtual Reality Computer Experiments Amy Tan

nr 77 **Measuring Housing Preferences Using Virtual Reality and Belief Networks** Maciej A. Orzechowski

#### nr 78

**Computational Representations of Words and Associations in Architectural Design** Nicole Segers

#### nr 79

Measuring and Predicting Adaptation in Multidimensional Activity-Travel Patterns Chang-Hyeon Joh

nr 80 **Strategic Briefing** Fayez Al Hassan

nr 81 **Well Being in Hospitals** Simona Di Cicco

nr 82 Solares Bauen: Implementierungs- und Umsetzungs-Aspekte in der Hochschulausbildung in Österreich Gerhard Schuster nr 83 Supporting Strategic Design of Workplace Environments with Case-Based Reasoning Shauna Mallory-Hill

nr 84 ACCEL: A Tool for Supporting Concept Generation in the Early Design Phase Maxim Ivashkov

nr 85 Brick-Mortar Interaction in Masonry under Compression Ad Vermeltfoort

nr 86 **Zelfredzaam Wonen** Guus van Vliet

nr 87 **Een Ensemble met Grootstedelijke Allure** Jos Bosman Hans Schippers

nr 88 On the Computation of Well-Structured Graphic Representations in Architectural Design Henri Achten

nr 89 **De Evolutie van een West-Afrikaanse Vernaculaire Architectuur** Wolf Schijns

nr 90 **ROMBO Tactiek** Christoph Maria Ravesloot

nr 91 External Coupling between Building Energy Simulation and Computational Fluid Dynamics Ery Djunaedy

nr 92 **Design Research in the Netherlands 2005** editors: Henri Achten Kees Dorst Pieter Jan Stappers Bauke de Vries

nr 93 **Ein Modell zur Baulichen Transformation** Jalil H. Saber Zaimian nr 94 Human Lighting Demands: Healthy Lighting in an Office Environment Myriam Aries

nr 95 A Spatial Decision Support System for the Provision and Monitoring of Urban Greenspace Claudia Pelizaro

nr 96 **Leren Creëren** Adri Proveniers

nr 97 **Simlandscape** Rob de Waard

nr 98 **Design Team Communication** Ad den Otter

nr 99 **Humaan-Ecologisch Georiënteerde Woningbouw** Juri Czabanowski

nr 100 **Hambase** Martin de Wit

nr 101 Sound Transmission through Pipe Systems and into Building Structures Susanne Bron-van der Jagt

nr 102 **Het Bouwkundig Contrapunt** Jan Francis Boelen

nr 103 **A Framework for a Multi-Agent Planning Support System** Dick Saarloos

nr 104 **Bracing Steel Frames with Calcium Silicate Element Walls** Bright Mweene Ng'andu

nr 105 **Naar een Nieuwe Houtskeletbouw** F.N.G. De Medts nr 106 and 107 *Niet gepubliceerd* 

nr 108 **Geborgenheid** T.E.L. van Pinxteren

nr 109 **Modelling Strategic Behaviour in Anticipation of Congestion** Qi Han

nr 110 **Reflecties op het Woondomein** Fred Sanders

nr 111 **On Assessment of Wind Comfort by Sand Erosion** Gábor Dezsö

nr 112 Bench Heating in Monumental Churches Dionne Limpens-Neilen

nr 113 **RE. Architecture** Ana Pereira Roders

nr 114 **Toward Applicable Green Architecture** Usama El Fiky

nr 115 Knowledge Representation under Inherent Uncertainty in a Multi-Agent System for Land Use Planning Liying Ma

nr 116 Integrated Heat Air and Moisture Modeling and Simulation Jos van Schijndel

nr 117 **Concrete Behaviour in Multiaxial Compression** J.P.W. Bongers

nr 118 **The Image of the Urban Landscape** Ana Moya Pellitero

nr 119 **The Self-Organizing City in Vietnam** Stephanie Geertman nr 120 A Multi-Agent Planning Support System for Assessing Externalities of Urban Form Scenarios Rachel Katoshevski-Cavari

#### nr 121

Den Schulbau Neu Denken, Fühlen und Wollen Urs Christian Maurer-Dietrich

nr 122 **Peter Eisenman Theories and Practices** Bernhard Kormoss

nr 123 **User Simulation of Space Utilisation** Vincent Tabak

nr 125 **In Search of a Complex System Model** Oswald Devisch

nr 126 Lighting at Work: Environmental Study of Direct Effects of Lighting Level and Spectrum on Psycho-Physiological Variables Grazyna Górnicka

#### nr 127

Flanking Sound Transmission through Lightweight Framed Double Leaf Walls Stefan Schoenwald

nr 128 **Bounded Rationality and Spatio-Temporal Pedestrian Shopping Behavior** Wei Zhu

nr 129 **Travel Information: Impact on Activity Travel Pattern** Zhongwei Sun

nr 130 Co-Simulation for Performance Prediction of Innovative Integrated Mechanical Energy Systems in Buildings Marija Trčka

nr 131 Niet gepubliceerd nr 132

Architectural Cue Model in Evacuation Simulation for Underground Space Design Chengyu Sun

#### nr 133

Uncertainty and Sensitivity Analysis in Building Performance Simulation for Decision Support and Design Optimization Christina Hopfe

nr 134 Facilitating Distributed Collaboration in the AEC/FM Sector Using Semantic Web Technologies Jacob Beetz

#### nr 135

**Circumferentially Adhesive Bonded Glass Panes for Bracing Steel Frame in Façades** Edwin Huveners

#### nr 136

Influence of Temperature on Concrete Beams Strengthened in Flexure with CFRP Ernst-Lucas Klamer

nr 137 **Sturen op Klantwaarde** Jos Smeets

#### nr 139

Lateral Behavior of Steel Frames with Discretely Connected Precast Concrete Infill Panels Paul Teewen

nr 140

Integral Design Method in the Context of Sustainable Building Design Perica Savanović

#### nr 141

Household Activity-Travel Behavior: Implementation of Within-Household Interactions Renni Anggraini

nr 142 **Design Research in the Netherlands 2010** Henri Achten nr 143 Modelling Life Trajectories and Transport Mode Choice Using Bayesian Belief Networks Marloes Verhoeven

#### nr 144 Assessing Construction Project Performance in Ghana William Gyadu-Asiedu

nr 145 Fmpowering

Empowering Seniors through Domotic Homes Masi Mohammadi

### nr 146

An Integral Design Concept for Ecological Self-Compacting Concrete Martin Hunger

#### nr 147

**Governing Multi-Actor Decision Processes in Dutch Industrial Area Redevelopment** Erik Blokhuis

### nr 148

A Multifunctional Design Approach for Sustainable Concrete Götz Hüsken

#### nr 149

**Quality Monitoring in Infrastructural Design-Build Projects** Ruben Favié

#### nr 150

Assessment Matrix for Conservation of Valuable Timber Structures Michael Abels

#### nr 151

**Co-simulation of Building Energy Simulation and Computational Fluid Dynamics for Whole-Building Heat, Air and Moisture Engineering** Mohammad Mirsadeghi

#### nr 152

External Coupling of Building Energy Simulation and Building Element Heat, Air and Moisture Simulation Daniel Cóstola nr 153 Adaptive Decision Making In Multi-Stakeholder Retail Planning Ingrid Janssen

nr 154 **Landscape Generator** Kymo Slager

nr 155 **Constraint Specification in Architecture** Remco Niemeijer

nr 156 A Need-Based Approach to Dynamic Activity Generation Linda Nijland

nr 157

Modeling Office Firm Dynamics in an Agent-Based Micro Simulation Framework Gustavo Garcia Manzato

nr 158 Lightweight Floor System for Vibration Comfort Sander Zegers

nr 159 **Aanpasbaarheid van de Draagstructuur** Roel Gijsbers

nr 160 **'Village in the City' in Guangzhou, China** Yanliu Lin

nr 161 **Climate Risk Assessment in Museums** Marco Martens

nr 162 **Social Activity-Travel Patterns** Pauline van den Berg

nr 163 Sound Concentration Caused by Curved Surfaces Martijn Vercammen

#### nr 164

**Design of Environmentally Friendly Calcium Sulfate-Based Building Materials: Towards an Improved Indoor Air Quality** Qingliang Yu nr 165 Beyond Uniform Thermal Comfort on the Effects of Non-Uniformity and Individual Physiology Lisje Schellen

nr 166 **Sustainable Residential Districts** Gaby Abdalla

#### nr 167

Towards a Performance Assessment Methodology using Computational Simulation for Air Distribution System Designs in Operating Rooms Mônica do Amaral Melhado

#### nr 168

Strategic Decision Modeling in Brownfield Redevelopment Brano Glumac

#### nr 169

Pamela: A Parking Analysis Model for Predicting Effects in Local Areas Peter van der Waerden

#### nr 170

A Vision Driven Wayfinding Simulation-System Based on the Architectural Features Perceived in the Office Environment Qunli Chen

#### nr 171

Measuring Mental Representations Underlying Activity-Travel Choices Oliver Horeni

#### nr 172

Modelling the Effects of Social Networks on Activity and Travel Behaviour Nicole Ronald

#### nr 173

Uncertainty Propagation and Sensitivity Analysis Techniques in Building Performance Simulation to Support Conceptual Building and System Design Christian Struck

nr 174 Numerical Modeling of Micro-Scale Wind-Induced Pollutant Dispersion in the Built Environment Pierre Gousseau nr 175 Modeling Recreation Choices over the Family Lifecycle Anna Beatriz Grigolon

#### nr 176

Experimental and Numerical Analysis of Mixing Ventilation at Laminar, Transitional and Turbulent Slot Reynolds Numbers Twan van Hooff

#### nr 177

**Collaborative Design Support: Workshops to Stimulate Interaction and Knowledge Exchange Between Practitioners** Emile M.C.J. Quanjel

#### nr 178

Future-Proof Platforms for Aging-in-Place Michiel Brink

#### nr 179

Motivate: A Context-Aware Mobile Application for Physical Activity Promotion Yuzhong Lin

#### nr 180

Experience the City: Analysis of Space-Time Behaviour and Spatial Learning Anastasia Moiseeva

# nr 181

Unbonded Post-Tensioned Shear Walls of Calcium Silicate Element Masonry Lex van der Meer

#### nr 182

**Construction and Demolition Waste Recycling into Innovative Building Materials for Sustainable Construction in Tanzania** Mwita M. Sabai

nr 183 **Durability of Concrete** with Emphasis on Chloride Migration Przemysław Spiesz

### nr 184

Computational Modeling of Urban Wind Flow and Natural Ventilation Potential of Buildings Rubina Ramponi nr 185 A Distributed Dynamic Simulation Mechanism for Buildings Automation and Control Systems Azzedine Yahiaoui

#### nr 186

Modeling Cognitive Learning of Urban Networks in Daily Activity-Travel Behavior Şehnaz Cenani Durmazoğlu

#### nr 187

Functionality and Adaptability of Design Solutions for Public Apartment Buildings in Ghana Stephen Agyefi-Mensah

#### nr 188

A Construction Waste Generation Model for Developing Countries Lilliana Abarca-Guerrero

#### nr 189

Synchronizing Networks: The Modeling of Supernetworks for Activity-Travel Behavior Feixiong Liao

#### nr 190

**Time and Money Allocation Decisions in Out-of-Home Leisure Activity Choices** Gamze Zeynep Dane

### nr 191

How to Measure Added Value of CRE and Building Design Rianne Appel-Meulenbroek

#### nr 192

Secondary Materials in Cement-Based Products: Treatment, Modeling and Environmental Interaction Miruna Florea

#### nr 193

Concepts for the Robustness Improvement of Self-Compacting Concrete: Effects of Admixtures and Mixture Components on the Rheology and Early Hydration at Varying Temperatures Wolfram Schmidt nr 194 Modelling and Simulation of Virtual Natural **Lighting Solutions in Buildings** Rizki A. Mangkuto

#### nr 195

**Nano-Silica Production at Low Temperatures** from the Dissolution of Olivine - Synthesis, **Tailoring and Modelling** Alberto Lazaro Garcia

#### nr 196

**Building Energy Simulation Based** Assessment of Industrial Halls for **Design Support** Bruno Lee

#### nr 197

**Computational Performance Prediction** of the Potential of Hybrid Adaptable **Thermal Storage Concepts for Lightweight** Low-Energy Houses Pieter-Jan Hoes

#### nr 198 **Application of Nano-Silica in Concrete** George Quercia Bianchi

nr 199 **Dynamics of Social Networks and Activity Travel Behaviour** Fariva Sharmeen

#### nr 200

**Building Structural Design Generation and Optimisation including Spatial Modification** Juan Manuel Davila Delgado

#### nr 201

Hydration and Thermal Decomposition of **Cement/Calcium-Sulphate Based Materials** Ariën de Korte

#### nr 202

**Republiek van Beelden:** De Politieke Werkingen van het Ontwerp in **Regionale Planvorming** Bart de Zwart

#### nr 203

**Effects of Energy Price Increases on** Individual Activity-Travel Repertoires and **Energy Consumption Dujuan Yang** 

#### nr 204

**Geometry and Ventilation:** Evaluation of the Leeward Sawtooth Roof Potential in the Natural Ventilation of Buildings Jorge Isaac Perén Montero

#### nr 205

Computational Modelling of Evaporative Cooling as a Climate Change Adaptation Measure at the Spatial Scale of Buildings and Streets Hamid Montazeri

#### nr 206

Local Buckling of Aluminium Beams in Fire Conditions Ronald van der Meulen

#### nr 207

Historic Urban Landscapes: Framing the Integration of Urban and Heritage Planning in Multilevel Governance Loes Veldpaus

#### nr 208

Sustainable Transformation of the Cities: Urban Design Pragmatics to Achieve a Sustainable Citv Ernesto Antonio Zumelzu Scheel

#### nr 209

**Development of Sustainable Protective** Ultra-High Performance Fibre Reinforced Concrete (UHPFRC): Design, Assessment and Modeling Rui Yu

#### nr 210

Uncertainty in Modeling Activity-Travel **Demand in Complex Uban Systems** Soora Rasouli

#### nr 211

Simulation-based Performance Assessment of Climate Adaptive Greenhouse Shells Chul-sung Lee

#### nr 212

Green Cities: Modelling the Spatial Transformation of the Urban Environment using Renewable **Energy Technologies** Saleh Mohammadi
nr 213 A Bounded Rationality Model of Short and Long-Term Dynamics of Activity-Travel Behavior

lfigeneia Psarra

#### nr 214

Effects of Pricing Strategies on Dynamic Repertoires of Activity-Travel Behaviour Elaheh Khademi

## nr 215

Handstorm Principles for Creative and Collaborative Working Frans van Gassel

## nr 216

Light Conditions in Nursing Homes: Visual Comfort and Visual Functioning of Residents Marianne M. Sinoo

## nr 217

Woonsporen: De Sociale en Ruimtelijke Biografie van een Stedelijk Bouwblok in de Amsterdamse Transvaalbuurt Hüseyin Hüsnü Yegenoglu

#### nr 218

Studies on User Control in Ambient Intelligent Systems Berent Willem Meerbeek

#### nr 219

Daily Livings in a Smart Home: Users' Living Preference Modeling of Smart Homes Erfaneh Allameh

## nr 220

Smart Home Design: Spatial Preference Modeling of Smart Homes Mohammadali Heidari Jozam

nr 221 **Wonen:** Discoursen, Praktijken, Perspectieven Jos Smeets

## nr 222

Personal Control over Indoor Climate in Offices: Impact on Comfort, Health and Productivity Atze Christiaan Boerstra nr 223 Personalized Route Finding in Multimodal Transportation Networks Jianwe Zhang

nr 224 **The Design of an Adaptive Healing Room for Stroke Patients** Elke Daemen

#### nr 225

Experimental and Numerical Analysis of Climate Change Induced Risks to Historic Buildings and Collections Zara Huijbregts

## nr 226

**Wind Flow Modeling in Urban Areas Through Experimental and Numerical Techniques** Alessio Ricci

### nr 227

Clever Climate Control for Culture: Energy Efficient Indoor Climate Control Strategies for Museums Respecting Collection Preservation and Thermal Comfort of Visitors Rick Kramer

#### nr 228

Fatigue Life Estimation of Metal Structures Based on Damage Modeling Sarmediran Silitonga

#### nr 229

A multi-agents and occupancy based strategy for energy management and process control on the room-level Timilehin Moses Labeodan

#### nr 230

Environmental assessment of Building Integrated Photovoltaics: Numerical and Experimental Carrying Capacity Based Approach Michiel Ritzen

## nr 231

Performance of Admixture and Secondary Minerals in Alkali Activated Concrete: Sustaining a Concrete Future Arno Keulen nr 232 World Heritage Cities and Sustainable Urban Development: Bridging Global and Local Levels in Monitoring the Sustainable Urban Development of World Heritage Cities Paloma C. Guzman Molina

### nr 233

Stage Acoustics and Sound Exposure in Performance and Rehearsal Spaces for Orchestras: Methods for Physical Measurements Remy Wenmaekers

#### nr 234

Municipal Solid Waste Incineration (MSWI) Bottom Ash: From Waste to Value Characterization, Treatments and Application Pei Tang

## nr 235

**Large Eddy Simulations Applied to Wind Loading and Pollutant Dispersion** Mattia Ricci

## nr 236

Alkali Activated Slag-Fly Ash Binders: Design, Modeling and Application Xu Gao

## nr 237

Sodium Carbonate Activated Slag: Reaction Analysis, Microstructural Modification & Engineering Application Bo Yuan

nr 238 **Shopping Behavior in Malls** Widiyani

#### nr 239

Smart Grid-Building Energy Interactions: Demand Side Power Flexibility in Office Buildings Kennedy Otieno Aduda

nr 240

Modeling Taxis Dynamic Behavior in Uncertain Urban Environments Zheng Zhong nr 241 Gap-Theoretical Analyses of Residential Satisfaction and Intention to Move Wen Jiang

# nr 242

**Travel Satisfaction and Subjective Well-Being: A Behavioral Modeling Perspective** Yanan Gao

## nr 243

Building Energy Modelling to Support the Commissioning of Holistic Data Centre Operation Vojtech Zavrel

## nr 244

Regret-Based Travel Behavior Modeling: An Extended Framework Sunghoon Jang

### nr 245

Towards Robust Low-Energy Houses: A Computational Approach for Performance Robustness Assessment using Scenario Analysis Rajesh Reddy Kotireddy

## nr 246

Development of sustainable and functionalized inorganic binder-biofiber composites Guillaume Doudart de la Grée

#### nr 247

A Multiscale Analysis of the Urban Heat Island Effect: From City Averaged Temperatures to the Energy Demand of Individual Buildings Yasin Toparlar

#### nr 248

Design Method for Adaptive Daylight Systems for buildings covered by large (span) roofs Florian Heinzelmann

#### nr 249

Hardening, high-temperature resistance and acid resistance of one-part geopolymers Patrick Sturm

## nr 250

Effects of the built environment on dynamic repertoires of activity-travel behaviour Aida Pontes de Aquino

### nr 251

Modeling for auralization of urban environments: Incorporation of directivity in sound propagation and analysis of a framework for auralizing a car pass-by Fotis Georgiou

#### nr 252

Wind Loads on Heliostats and Photovoltaic Trackers Andreas Pfahl

#### nr 253

Approaches for computational performance optimization of innovative adaptive façade concepts Roel Loonen

### nr 254

Multi-scale FEM-DEM Model for Granular Materials: Micro-scale boundary conditions, Statics, and Dynamics Jiadun Liu

#### nr 255

Bending Moment - Shear Force Interaction of Rolled I-Shaped Steel Sections Rianne Willie Adriana Dekker

## nr 256

Paralympic tandem cycling and handcycling: Computational and wind tunnel analysis of aerodynamic performance Paul Fionn Mannion

#### nr 257

Experimental characterization and numerical modelling of 3D printed concrete: Controlling structural behaviour in the fresh and hardened state Robert Johannes Maria Wolfs

### nr 258

Requirement checking in the building industry: Enabling modularized and extensible requirement checking systems based on semantic web technologies Chi Zhang nr 259

A Sustainable Industrial Site Redevelopment Planning Support System Tong Wang

### nr 260

Efficient storage and retrieval of detailed building models: Multi-disciplinary and long-term use of geometric and semantic construction information Thomas Ferdinand Krijnen

#### nr 261

The users' value of business center concepts for knowledge sharing and networking behavior within and between organizations Minou Weijs-Perrée

## nr 262

Characterization and improvement of aerodynamic performance of vertical axis wind turbines using computational fluid dynamics (CFD) Abdolrahim Rezaeiha

#### nr 263

In-situ characterization of the acoustic impedance of vegetated roofs Chang Liu

#### nr 264

Occupancy-based lighting control: Developing an energy saving strategy that ensures office workers' comfort Christel de Bakker

#### nr 265

**Stakeholders-Oriented Spatial Decision Support System** Cahyono Susetyo

#### nr 266

**Climate-induced damage in oak museum objects** Rianne Aleida Luimes

## nr 267

Towards individual thermal comfort: Model predictive personalized control of heating systems Katarina Katic nr 268 Modelling and Measuring Quality of Urban Life: Housing, Neighborhood, Transport and Job Lida Aminian

## nr 269

**Optimization of an aquifer thermal energy storage system through integrated modelling of aquifer, HVAC systems and building** Basar Bozkaya

### nr 270

Numerical modeling for urban sound propagation: developments in wave-based and energy-based methods Raúl Pagán Muñoz

nr 271

Lighting in multi-user office environments: improving employee wellbeing through personal control Sanae van der Vleuten-Chraibi

nr 272

A strategy for fit-for-purpose occupant behavior modelling in building energy and comfort performance simulation Isabella I. Gaetani dell'Aquila d'Aragona

nr 273

Een architectuurhistorische waardestelling van naoorlogse woonwijken in Nederland: Het voorbeeld van de Westelijke Tuinsteden in Amsterdam.

Eleonore Henriette Marie Mens